Central States, Southeast and Southwest Areas Health and Welfare Fund Active Plan Document (Non-Grandfathered)



As Amended and Restated as of January 1, 2022

CENTRAL STATES, SOUTHEAST AND SOUTHWEST AREAS HEALTH AND WELFARE FUND ACTIVE PLAN (NON-GRANDFATHERED), a jointly administered, defined benefit active employee benefit plan

ADDRESS OF ADMINISTRATIVE OFFICE

8647 West Higgins Road Chicago, Illinois 60631

TELEPHONE NUMBER

(847) 518-9800 800-832-6227 (Toll-Free)

EMPLOYER IDENTIFICATION NUMBER

36-2154936

BOARD OF TRUSTEES

EMPLOYEE TRUSTEES

Charles A. Whobrey George J. Westley Gary Dunham Trevor Lawrence

EMPLOYER TRUSTEES

Gary F. Caldwell Christopher J. Langan Robert Whitaker Mark F. Angerame

EXECUTIVE DIRECTOR
(also Agent for Service of Legal Process)
Thomas C. Nyhan

TABLE OF CONTENTS

ARTICLE I.	DEFINITIONS	1
ARTICLE II.	. EFFECTIVE DATE OF GENERAL PROVISIONS	17
ARTICLE III	I. PARTICIPATION AND COVERAGE	18
3.01	QUALIFICATIONS AS A COVERED PARTICIPANT	18
3.02	QUALIFICATIONS AS A COVERED DEPENDENT	
3.03	COMMENCEMENT OF COVERAGE OF MEMBERS OF A NEW GROUP PREVIOUSLY INSURED	18
3.04	COMMENCEMENT OF COVERAGE OF MEMBERS OF A NEW GROUP NOT PREVIOUSLY INSURED	18
3.05	COMMENCEMENT OF COVERAGE OF NEW PARTICIPANTS	
3.06	COMMENCEMENT OF COVERAGE OF COVERED PARTICIPANTS FOR	
0.00	WHOM UNINTERRUPTED EMPLOYER CONTRIBUTIONS AND/OR	
	SELF-PAYMENTS HAVE BEEN RECEIVED	19
3.07	COMMENCEMENT OF COVERAGE OF EMPLOYEES COVERED UNDER	
0.07	ANOTHER PLAN OFFERED BY THE FUND	19
3.08	COMMENCEMENT OF COVERAGE OF NEW PARTICIPANTS ELIGIBLE	
0.00	UNDER A PLAN OFFERED BY A LOCAL UNION	20
3.09	EFFECT OF TEMPORARY WORK STOPPAGE ON COVERAGE	_
3.10	EFFECT ON COVERAGE OF ABSENCE FROM CONTINUOUS FUNDED	
	EMPLOYMENT DURING, AND AS A RESULT OF, SERVICE IN THE	
	UNIFORMED SERVICES	21
3.11	TERMINATION OF COVERAGE OF PARTICIPANTS HAVING TOTAL AND	
	PERMANENT DISABILITY OR ON SICK LEAVE	21
3.12	TERMINATION OF COVERAGE OF PARTICIPANTS CHANGING EMPLOYEE	
	STATUS	21
3.13	TERMINATION OF COVERAGE DUE TO EMPLOYER WITHDRAWAL	22
3.14	RETURN TO COVERED EMPLOYMENT	22
3.15	ELIGIBILITY TO ELECT CONTINUATION COVERAGE	22
3.16	DEFINITION OF QUALIFYING EVENT	22
3.17	NOTICE OF QUALIFYING EVENT	23
3.18	ELIGIBILITY FOR CONTINUATION COVERAGE	_
3.19	PROCEDURES TO ELECT CONTINUATION COVERAGE	23
3.20	SELF-PAYMENTS TO MAINTAIN CONTINUATION COVERAGE	24
3.21	TERMINATION OF CONTINUATION COVERAGE	24
3.22	STATUS OF A COVERED PARTICIPANT WHEN EMPLOYER FAILS TO MAKE CONTRIBUTIONS	25
3.23	STATUS AND COVERAGE OF A COVERED INDIVIDUAL WHEN EMPLOYER	
	FAILS TO REMIT EMPLOYER CONTRIBUTIONS: SUSPENSION OF BENEFITS	25
3.24	CERTAIN PARTICIPANTS NOT TO BE CLASSIFIED AS COVERED DEPENDENT	25
3.25	COMMENCEMENT OF COVERED DEPENDENT STATUS FOR SPOUSE OF A	
	COVERED PARTICIPANT	26

3.26	COMMENCEMENT OF COVERED DEPENDENT STATUS FOR CHILD OF A	
	COVERED PARTICIPANT	26
3.27	STATUS OF MENTALLY OR PERMANENTLY PHYSICALLY DISABLED CHILD	
	OF A COVERED PARTICIPANT	26
3.28	STATUS OF NEWBORN CHILD OF A COVERED PARTICIPANT	26
3.29	STATUS OF CHILD OF A COVERED PARTICIPANT	26
3.30	LOSS OF DEPENDENT'S COVERAGE	27
3.31	RESIDUAL COVERAGE OF FORMER COVERED PARTICIPANTS	27
3.32	ELIGIBILITY FOR FAMILY PROTECTION PLAN BENEFIT	27
3.33	COVERAGE UNDER AN HOURLY PLAN (on and after November 1, 2020)	28
3.34	RESERVED FOR FUTURE USE	
3.35	QUALIFYING EVENT - HOURLY PLAN (on and after November 1, 2020)	31
3.36	COMMENCEMENT OF COVERAGE UNDER A MULTI-TIER PLAN	
3.37	OPEN ENROLLMENT UNDER A MULTI-TIER PLAN	33
3.38	SPECIAL ENROLLMENT	33
3.39	DECLINING COVERAGE	35
ARTICLE	IV. GENERAL EXCLUSIONS, LIMITATIONS AND CONDITIONS FOR PAYMENT	
	OF CLAIMS	36
4.04	DAYMENT ONLY FOR OOMERED OF AIMO OF OOMERED INDIVIDUAL O	
4.01	PAYMENT ONLY FOR COVERED CLAIMS OF COVERED INDIVIDUALS	36
4.02	EXCLUSION OF PAYMENT FOR TREATMENT NOT CONSIDERED STANDARD	
4.00	MEDICAL CARE OR MEDICALLY NECESSARY	36
4.03	LIMITATION ON PAYMENT OF CLAIMS ARISING FROM WORK-RELATED	
4.0.4	INJURY OR COVERED BY WORKERS' COMPENSATION	36
4.04	EXCLUSION OF PAYMENT FOR TREATMENT OF INJURIES SUSTAINED	
4.0=	WHILE IN ANY UNIFORMED SERVICE	36
4.05	EXCLUSION OF PAYMENT FOR TREATMENT DUE TO ILLNESS OR INJURY	
	ARISING OUT OF ANY ACT OF WAR OR CIVIL DISTURBANCE	37
4.06	EXCLUSION OF PAYMENT FOR TREATMENT OF INJURIES ARISING AS A	
	RESULT OF PARTICIPATION IN CRIMINAL CONDUCT	37
4.07	LIMITATION ON PAYMENT FOR TREATMENT RECEIVED OUTSIDE THE	
4.00	UNITED STATES	37
4.08	EXCLUSION FOR PAYMENT FOR TREATMENT CONNECTED WITH SURGERY	
4.00	FOR COSMETIC PURPOSES	37
4.09	EXCLUSION OF PAYMENT FOR TREATMENT OTHERWISE COVERED UNDER	
4.46	THE SOCIAL SECURITY ACT	38
4.10	EXCLUSION OF PAYMENT FOR TREATMENT NOT RELATED TO ILLNESS,	0.0
	INJURY, CONGENITAL ANOMALY OR PREGNANCY	
4.11	PAYMENT FOR CERTAIN DENTAL TREATMENTS	39
4.12	EXCLUSION OF PAYMENT FOR TREATMENT OF NON-COMPENSABLE	
	PROCEDURES	
4.13	RESERVED FOR FUTURE USE	
4.14	EXCLUSION OF PAYMENT FOR CERTAIN ITEMS	
4.15	EXCLUSION OF PAYMENT FOR MAINTENANCE CARE	
4.16	EXCLUSION OF PAYMENT OVER PRESCRIBED MAXIMUMS	39

	4.17	LIMITATION ON ELIGIBILITY FOR COVERAGE OF ORGAN OR TISSUE	40
	4.40	TRANSPLANTS	40
	4.18	EXCLUSION OF PAYMENT FOR INFERTILITY TREATMENT AND SERVING AS	40
	4.40	A SURROGATE MOTHER	40
	4.19	EXCLUSION OF PAYMENT FOR CHARGES FOR WHICH THE COVERED	40
	4.00	INDIVIDUAL IS NOT RESPONSIBLE TO PAY	40
	4.20	LIMITATION ON PAYMENT OF CLAIMS FOR SERVICES BY PROVIDERS NOT	40
		IN PREFERRED PROVIDER ORGANIZATION NETWORK	40
AR	TICLE V.	COORDINATION OF BENEFITS	41
	5.01	PRIORITY OF COVERAGE WHERE COVERED INDIVIDUAL IS COVERED BY	
		AN OTHER PLAN	
	5.02	EFFECT OF PRIORITY RULES ON AMOUNT OF PAYMENTS UNDER THE PLAN	42
	5.03	RECOVERY OF PAYMENTS WHEN AN OTHER PLAN IS INVOLVED	
	5.04	PAYMENTS TO OTHER PLANS	43
	5.05	NO FAULT, PERSONAL INJURY PROTECTION OR FINANCIAL	
		RESPONSIBILITY MOTOR VEHICLE INSURANCE COVERAGE	43
	5.06	COORDINATION OF BENEFITS WITH ANOTHER PLAN	44
۸ ۵	TIOL	CONTRIBUTIONS AND FUNDING OF DENEFITS	4.5
AK	TICLE VI	. CONTRIBUTIONS AND FUNDING OF BENEFITS	45
	6.01	EMPLOYER'S OBLIGATION TO CONTRIBUTE TO THE PLAN	45
	6.02	CREDITS FOR ERRONEOUS EMPLOYER CONTRIBUTIONS	
	6.03	IRREVOCABLE NATURE OF CONTRIBUTIONS	
	6.04	SELF-PAYMENTS	
	0.01		
AR	TICLE VI	I. AMENDMENTS AND PLAN TERMINATION	46
	7.01	PROCEDURE FOR AMENDING THE PLAN	
	7.02	TERMINATION OF THE PLAN	46
۸ ۵		II. DI ANI ADMINISTRATIONI	47
AK	TICLE VI	II. PLAN ADMINISTRATION	47
	8.01	TRUSTEE STATUS AS "NAMED FIDUCIARY"	47
	8.02	POWERS OF THE TRUSTEES	
	8.03	DECISIONS OF TRUSTEES	
	8.04	EFFECT OF ANY MISREPRESENTATION WITH RESPECT TO CLAIMS	
	8.05	PAYMENTS TO PERSONS WHO HAVE FAILED TO INFORM THE TRUSTEES	17
	0.00	OF A CHANGE OF ADDRESS	47
	8.06	INFORMATION CONCERNING COVERED INDIVIDUALS	
	8.07	USE AND DISCLOSURE OF PROTECTED HEALTH INFORMATION	
	8.08	SAFEGUARDS FOR PROTECTED HEALTH INFORMATION	_
	0.00	S. A. E. C. C. A. C.	40
AR	TICLE IX	BENEFIT CLAIMS	49
	9.01	CLAIMS TO BE SUBMITTED IN WRITING ON AUTHORIZED FORMS	49

	9.02	PROCESSING CLAIMS INVOLVING URGENT CARE	49
	9.03	PROCESSING CLAIMS FOR BENEFITS (OTHER THAN URGENT CARE CLAIMS)	50
	9.04	NOTICE OF ADVERSE BENEFIT DETERMINATIONS	50
	9.05	CONCURRENT CARE DECISIONS	
	9.06	MISCELLANEOUS BENEFIT CLAIMS PROVISIONS	52
AR	TICLE X	. APPELLATE REVIEW PROCEDURES AND DETERMINATIONS	54
	10.01	PROCEDURES DURING APPELLATE REVIEW OF ADVERSE BENEFIT DETERMINATIONS	E 1
	10.02	DEFINITION OF TRUSTEE-REVIEWABLE DETERMINATIONS	
	10.02	TIME LIMITATIONS FOR APPELLATE REVIEW OF ADVERSE BENEFIT	50
	10.03	DETERMINATIONS	57
	10.04	NOTICE OF BENEFIT DETERMINATIONS AFTER APPELLATE REVIEW	
	10.04	MISCELLANEOUS APPELLATE REVIEW PROVISIONS	
	10.06	EXTERNAL REVIEW	
AR	TICLE X	I. MISCELLANEOUS PROVISIONS	63
	11.01	VALIDITY OF CHANGES IN THE PLAN	63
	11.02	CLAIM FORMS	
	11.03	TIME WITHIN WHICH CERTAIN CLAIMS ARE TO BE FILED	63
	11.04	RECOVERY OF EXCESS PAYMENTS	63
	11.05	FUND MAY ORDER PHYSICAL EXAMINATION AND/OR AUTOPSY	63
	11.06	TO WHOM BENEFITS ARE PAYABLE	63
	11.07	CERTAIN ACTS OF THE FUND DO NOT CONSTITUTE A WAIVER OF RIGHTS	
	11.08	NO MEDICAL EXAMINATION REQUIRED AS PREREQUISITE TO COVERAGE	64
	11.09	ALL MEDICAL BENEFIT PAYMENTS BASED ON ALLOWED AMOUNT FOR	
		THE SERVICE	
	11.10	PERIOD DURING WHICH BENEFIT PAYMENTS MUST BE CLAIMED	
	11.11	APPLICABLE LAW	
	11.12	SEVERABILITY OF PLAN PROVISIONS	
		RIGHT TO REVIEW ALL CLAIMS	
	11.14	SUBROGATION	
	11.15	WORKERS' COMPENSATION SUBROGATION	
	11.16	RIGHT TO PROVIDE ALTERNATIVE CAREPAYMENT BASED ON PROPRIETY OF PROCEDURES	
	11.17		
	11.18 11.19	SPECIAL ARRANGEMENTS FOR BENEFITS FORUM SELECTION CLAUSE	
AR		II. BASIC BENEFITS	
	12.04	OUTLINE OF BASIC BENEFITS	71
	12.01 12.02	SHORT-TERM DISABILITY BENEFIT—PARTICIPANT ONLY	
	12.02	HOSPITAL EXPENSE BENEFIT	
	12.03	SURGICAL AND MATERNITY EXPENSE BENEFIT	
	12.04	OUTPATIENT DIAGNOSTIC X-RAY AND LABORATORY EXPENSE BENEFIT	
	12.00	OUT ATIENT DIAGNOSTIC ATIAT AND LABORATORT EXPENSE DENERTI	14

12.06 OUTPATIENT ACCIDENTAL BODILY INJURY EXPENSE BENEFIT	74
12.07 PRESCRIPTION DRUG BENEFIT	75
12.08 BEHAVIORAL HEALTH BENEFIT - INPATIENT	76
12.09 BEHAVIORAL HEALTH BENEFIT - OUTPATIENT	77
12.10 ORGAN TRANSPLANT DONOR BENEFIT	78
12.11 HEARING AID BENEFIT	78
12.12 OUTPATIENT CANCER TREATMENT BENEFIT	79
12.13 AMBULANCE SERVICE BENEFIT	
12.14 CHIROPRACTIC EXPENSE BENEFIT	
12.15 WOMEN'S HEALTH BENEFIT	
12.16 MAYO CLINIC TREATMENT	
12.17 WELLNESS BENEFIT	
12.18 WAIVER OF DEDUCTIBLE/CO-PAYMENT REQUIREMENT (OFFICE VISITS)	
12.19 PAYMENT OF CERTAIN BASIC BENEFITS FOR TREATMENT CONTINUING	
AFTER TERMINATION OF COVERAGE UNDER THE PLAN	83
12.20 TERMINATION OF THE BASIC BENEFIT EXTENSION	84
12.21 RETIREE CONTRIBUTION BENEFIT	
12.21 RETIRES CONTRIBOTION BENEFIT	
ARTICLE XIII. MAJOR MEDICAL EXPENSE BENEFITS	85
13.01 OUTLINE OF MAJOR MEDICAL BENEFITS	85
13.02 MAXIMUM MAJOR MEDICAL EXPENSE BENEFITS	85
13.03 EXTENSION OF THE MAJOR MEDICAL EXPENSE BENEFIT	85
13.04 TERMINATION OF THE MAJOR MEDICAL EXTENSION	86
ARTICLE XIV. LIFE INSURANCE BENEFITS	87
14.01 BENEFITS	97
14.02 LIFE INSURANCE BENEFIT	
14.03 DEPENDENT LIFE INSURANCE BENEFIT	
14.04 TOTAL AND PERMANENT DISABILITY INSTALLMENT/WAIVER OF PREMIUM	_
DISABILITY BENEFITS	
14.05 TOTAL AND PERMANENT DISABILITY INSTALLMENT BENEFIT	
14.06 WAIVER OF PREMIUM DISABILITY BENEFIT	
14.07 ELIGIBILITY AND ADMINISTRATION	
14.08 ACCIDENTAL DEATH AND DISMEMBERMENT BENEFIT	
14.09 BENEFICIARY AND MODE OF SETTLEMENT	
14.10 GRACE PERIOD	
14.10 GRACE PERIOD	92
ARTICLE XV. DENTAL BENEFITS	93
15.01 PAYMENT FOR CERTAIN DENTAL TREATMENT AND SERVICES	93
15.02 COVERED DENTAL PROCEDURES AND MAXIMUM AMOUNT PAYABLE	93
15.03 LIMITATIONS ON PAYMENT OF DENTAL BENEFITS	93
15.04 ORTHODONTIA BENEFIT—DEPENDENT CHILDREN ONLY	96
· · · · · · · · · · · · · · · · · · ·	
15.05 EXTENSION OF DENTAL BENEFITS	96

	15.07	COVERAGE OF PEDIATRIC ORAL CARE	97
AR	ΓICLE X	VI. VISION BENEFITS	98
	16.01 16.02 16.03 16.04	COVERED VISION EXPENSES COVERED VISION PROCEDURES AND MAXIMUM AMOUNT PAYABLE LIMITATION ON PAYMENT FOR VISION BENEFITS COVERAGE OF PEDIATRIC VISION CARE	98 98
AR	TICLE X	VII. OUT-OF-POCKET EXPENSE LIMIT	. 100
	17.01	OUT-OF-POCKET EXPENSE IS THAT PORTION OF ELIGIBLE EXPENSES INCURRED THAT IS THE COVERED INDIVIDUAL'S RESPONSIBILITY AFTER THE FUND HAS PAID ITS REQUIRED BENEFITS. THE FUND PROVIDES COVERED INDIVIDUALS WITH OUT-OF-POCKET EXPENSE LIMITS. AMOUNTS OF THE OUT-OF-POCKET EXPENSE LIMIT (TO THE EXTENT THOSE LIMITS ARE APPLICABLE IN SPECIFIC CIRCUMSTANCES) ARE SPECIFIED IN SECTION 20.05.	100
	17.02	PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE IS THAT PORTION OF ELIGIBLE PRESCRIPTIONS DRUG EXPENSES INCURRED THAT IS THE COVERED INDIVIDUAL'S RESPONSIBILITY AFTER THE FUND HAS PAID ITS REQUIRED BENEFITS. THE FUND PROVIDES COVERED INDIVIDUALS WITH PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE LIMITS. AMOUNTS OF THE PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE LIMIT (TO THE EXTENT THOSE LIMITS ARE APPLICABLE IN SPECIFIC CIRCUMSTANCES) ARE SPECIFIED IN SECTION 20.08.	. 101
	17.03	MAXIMUM OUT-OF-POCKET LIMIT ON COST SHARING	. 102
AR	TICLE X	VIII. PLAN BENEFIT LIMIT	. 103
	18.01	THE TERM PLAN BENEFIT LIMIT IS DEFINED AS THE MAXIMUM PAYABLE BY THE PLAN IN A GIVEN CALENDAR YEAR UNDER ANY OR ALL APPLICABLE PLAN BENEFITS NOT TAKING INTO ACCOUNT ANY RETIREE CONTRIBUTION BENEFIT. THE PLAN MAY INCLUDE SUCH PLAN BENEFIT LIMIT AS DESCRIBED. THE AMOUNT OF THE PLAN BENEFIT LIMIT, IF ANY, IS DETERMINED BY REFERENCING SECTION 20.06.	. 103
AR	ΓICLE X	IX. EFFECTIVE DATE OF PLAN SPECIFIC PROVISIONS	. 104
	19.01	THE TERMS AND PROVISIONS OF THIS PUBLISHED EDITION OF THIS PLAN DOCUMENT (COVERING ALL PLANS) ARE APPLICABLE TO THE COVERAGE OF EVERY COVERED INDIVIDUAL WHO HAS BEEN, IS OR HEREAFTER BECOMES ENTITLED TO COVERAGE BY THE PLAN AS OF ANY DATE ON OR AFTER JANUARY 1, 2022, PROVIDED THAT THE TERMS, PROVISIONS, LIMITATIONS AND EXCLUSIONS OF COVERAGE AS OF ANY DATE PRIOR TO JANUARY 1, 2022, ARE GOVERNED BY THE EARLIER EDITION OF THIS PLAN	

	DOCUMENT THAT WAS IN EFFECT ON THAT DATE (INCLUDING ALL PLAN	
	AMENDMENTS OF THAT EDITION ADOPTED AND IN EFFECT ON THAT DATE),	
	AND PROVIDED FURTHER THAT THE TERMS AND PROVISIONS OF THIS	
	EDITION ALSO INCLUDE ALL PLAN AMENDMENTS ADOPTED AND IN EFFECT	
	AFTER PUBLICATION OF THIS EDITION (EVEN IF THEY ARE YET TO BE	
	INCORPORATED IN THIS EDITION)	104
ARTICLE X	(X. SCHEDULE OF BENEFITS	105
20.01	BASIC BENEFITS AND MAJOR MEDICAL EXPENSE BENEFITS	
20.02	DENTAL BENEFITS	112
20.03	VISION BENEFITS	114
20.04	LIFE INSURANCE BENEFIT AND ACCIDENTAL DEATH AND DISMEMBERMENT	
	BENEFIT	116
20.05	MEDICAL OUT-OF-POCKET-EXPENSE LIMIT	118
20.06	PLAN BENEFIT LIMIT	118
20.07	PLAN DEDUCTIBLE	119
20.08	TEAMCARE PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE LIMITS	120

ARTICLE I. DEFINITIONS

The terms used in this Plan shall have the following meanings.

1.01 Accidental Bodily Injury:

Physical damage to the body resulting from a sudden and unexpected event, injury or external force occurring without forewarning.

1.02 Accidental Death:

Any death directly and solely resulting from external means or an external cause, as opposed to a death caused or contributed to by a disease or infirmity. The Accidental Death and Dismemberment Benefit is not payable if a death or other loss that is otherwise within the scope of that benefit is caused or contributed to by any act, omission, condition or event that is described as a limitation or exclusion in Section 14.08.

1.03 Accidental Death and Dismemberment Benefit:

- (a) The benefit payable to the beneficiary of a Covered Participant upon the death of the Covered Participant resulting from Accidental Bodily Injury; or
- (b) The benefit payable to a Covered Participant upon the accidental loss of a limb or an eye as set forth in Sections 14.08 and 20.04.

The Accidental Death and Dismemberment Benefit is not payable if a death or other loss that is otherwise within the scope of that benefit is caused or contributed to by any act, omission, condition or event that is described as a limitation or exclusion in Section 14.08.

1.04 Active Employee:

A person who is actively at work as an Employee, after the commencement date of coverage, except that, with respect to periods during which an Employee's Employer is obligated to make Employer Contributions on his behalf pursuant to a Collective Bargaining Agreement or applicable law, or with respect to periods during which Coverage is otherwise available for the Employee under this Plan, an Employee on vacation, involved in a Temporary Work Stoppage, on Sick Leave, or confined to a hospital shall be considered an Active Employee. An Employee on Leave of Absence is not an Active Employee, regardless of whether the Employee has Coverage under the Plan, unless such Leave of Absence is also Family Medical Leave under federal law. An Employee on Lay-off, or otherwise unable to actively work, is not an Active Employee.

1.05 Alcoholism or Substance Abuse Treatment Facility:

A treatment facility or clinic which provides a program of effective medical and therapeutic treatment for either alcoholism and/or substance abuse approved by the attending Physician and the Fund, and which:

- (a) Is licensed, certified or approved as an Alcoholism and/or Substance Abuse Treatment Facility by the state or jurisdiction in which it is located, and does not have a license as a "Hospital";
- (b) Has or maintains a specific and detailed program requiring full residence or full participation by the patient; and
- (c) Provides at least the following basic services:
 - (1) Room and board (inpatient);
 - (2) Evaluation and diagnosis;
 - (3) Counseling; and
 - (4) Referral and orientation to specialized community resources.
- 1.06 Average Weekly Base Pay:

Average weekly base pay is calculated by averaging the paid hours (maximum of 40 per week) each week during the last quarter in which you worked the complete quarter and multiplying that average by the hourly rate of your base pay.

If you did not work the complete quarter, the Plan will look back a maximum of four quarters for the most recently worked complete quarter. If there is no complete quarter in the last four, the Plan uses the quarter with the most hours worked. The number of hours (maximum of 40 per week) and the number of weeks you worked in that quarter (not the total number of weeks in the quarter) are used to calculate your average weekly base rate.

1.07 Basic Benefits:

Basic Benefits are described in detail in Article XII and Section 20.01 (a) through (q).

1.08 Bone Marrow Transplant:

Any care, treatment services or supplies relating to the harvesting and/or transfer of stem cells and/or bone marrow into, or back into, the patient.

1.09 Child:

A Participant's natural child, adopted child, step-child; a child placed with a Participant for adoption; or a child for whom the Participant is obligated to provide support pursuant to a Qualified Medical Child Support Order. A child whose legal quardian or custodian is a Participant shall only be considered a "Child" under this definition if the Participant establishes that the guardianship or custodianship is permanent and established pursuant to court order and the Participant (or the Participant and spouse) is the sole support of the child unless that child is a beneficiary under a Qualified Medical Child Support Order under federal law. Temporary designation of guardianship entered into primarily for the purpose of obtaining coverage for a person under this Plan shall not qualify that person as a "Child" eligible for coverage. The term "Child" shall not include the child of a Participant in cases where the Participant or the Participant's spouse is serving as a Surrogate Mother (even if the Participant or Spouse provides the ovum).

1.10 Chiropractor:

A legally qualified and licensed Chiropractor.

- 1.11 Clinical Psychologist:
- (a) A person who is licensed or certified as a Psychologist by the appropriate governmental authority having jurisdiction over such licensure or certification, as the case may be, in the jurisdiction where such person renders service to the Covered Individual; or
- (b) A person who is a Member or Fellow of the American Psychological Association, if there is no licensure or certification in the jurisdiction where such person renders service to the Covered Individual.
- 1.12 Collective Bargaining Agreement:

An agreement reached by bargaining as to wages and conditions of work and to which the Local Union is a party.

1.13 Continuation Coverage:

A continuation of the same terms, conditions, limitations and exclusions of Coverage as are provided by the Fund to a Covered Individual on the day before a Qualifying Event upon completion of election procedures pursuant to Section 3.19.

1.14 Continuous Funded Employment:

Uninterrupted employment of a Participant during which continuous Employer Contributions have been made on his behalf, by an Employer, to the Fund or to any Other Plan having a Reciprocity Agreement in effect with the Fund.

1.15	Cosmetic:	Care, treatment, services or supplies the primary effect of which is to improve the physical appearance of a Covered Individual. The fact that there may be an incidental medical benefit does not prevent a determination that the care, treatment, services or supplies are cosmetic.
1.16	Coverage:	Full entitlement to all benefits of this Plan by a Participant or Dependent, unless limited or excluded by any provision of this Plan.
1.17	Coverage Month	A period of 4 or 5 weeks starting on the first Sunday on or before the first day of the calendar month and finishing on the final Saturday of the calendar month.
1.18	Covered Dependent:	A Dependent who qualifies for Coverage under this Plan in accordance with the provisions of Article III of this Plan.
1.19	Covered Individual:	A Covered Participant or a Covered Dependent. See also Section 3.31.
1.20	Covered Participant:	A Participant who qualifies for Coverage under this Plan in accordance with the provisions of Article III of this Plan.
1.21	Dental Benefits:	The benefits set forth in Article XV and Section 20.02.
1.22	Dentist:	A legally qualified and licensed Dentist.
1.23	Dependent:	A Participant's Spouse or Child.
1.24	Disability:	An illness, injury or pregnancy, except as provided in Section 12.10.
1.25	Discharge:	A permanent termination of employment initiated by the Employer.
1.26	Eligible Major Medical Expenses:	(a) The Allowed Amount incurred by a Covered Individual for medical services, supplies and treatments performed or prescribed by a Physician, including:

- (1) Charges by a Hospital for room and board and other services required for purposes of treatment. The allowance for a private room is limited to the Hospital's average semi-private room rate;
- (2) Charges by a Physician or surgeon for professional services, except those related to a routine physical examination;
- (3) Charges for services of legally licensed physiotherapists, graduate registered nurses and other allied licensed health professionals, provided such services are not rendered by a member of the person's family;
- (4) Charges for drugs and medicines purchased with a Physician's prescription, dispensed by a pharmacist and bearing the Federal or State Legend;
- (5) Charges for durable medical equipment as may be approved by the Fund. Rental charges shall in no event exceed the purchase price of such items. Purchase of such items will be approved only if deemed by the Fund to be more economical than rental;
- (6) Charges for prosthetics or prosthetic devices;
- (7) Charges for X-ray and laboratory procedures, except those related to a routine physical examination;
- (8) Charges by a Hospital or by a professional licensed ambulance service for necessary transportation by ambulance to and from the Hospital;
- (9) Charges for repair of the jaws and for repair or replacement of natural teeth damaged through accidental bodily injury;
- (10) Charges for chemotherapy or radiation treatments;
- (11) Charges for contact lenses and/or glasses prescribed to treat glaucoma, keratoconus or resulting from cataract surgery, once in a lifetime;
- (12) Charges for hearing aids and one (1) set of batteries, once every three (3) years;

- (13) Charges for surgical assistance;
- (14) Charges for renal dialysis; and
- (15) Charges for outpatient cardiac rehabilitation programs after onset of cardiac event or invasive cardiac procedure performed at a qualified facility.
- (b) The following are not considered Eligible Major Medical Expenses:
 - Any charge excluded under Article IV, General Exclusions, Limitations And Conditions For Payment Of Claims, or under Article V, Coordination Of Benefits;
 - (2) Any charge for eye examinations for the correction of vision, fitting of glasses or contact lenses, except as otherwise provided in Section 1.26(a)(11);
 - (3) Medicines or drugs which can be purchased over the counter, dietary supplements and vitamins, or any other drugs not specifically authorized by the Board of Trustees;
 - (4) Any charge, or any portion of any charge, covered by any benefit of this Plan, other than the Major Medical Expense Benefit under Article XIII;
 - (5) Any portion of a provider charge that exceeds the Allowed Amount;
 - (6) Cost of transportation and lodging in connection with medical treatment, transportation equipment, construction modifications, clothing (including undergarments) and capital asset items;
 - (7) Specialized furniture and equipment unless approved by the Fund;
 - (8) Any dental service or appliance, even if performed in conjunction with medical treatment, except as otherwise provided in Section 1.26(a)(9);
 - (9) Any charge for educational programs or materials; and

1.27 Employee:

- (10) Food and/or specialized nutritional products unless approved by the Fund.
- (a) All persons who are accepted by the Trustees for participation in the Fund, under the terms and conditions stated by the Trustees for participation, and who are Active Employees of an Employer under the terms and conditions of a Collective Bargaining Agreement which requires Employer Contributions be made to the Fund, and such other employees of the Employer as are proposed and accepted by the Trustees for participation, on whose behalf payments are required by the agreement of the Employer or applicable law to be made to the Fund;
- (b) All persons employed by the Union upon being proposed by the Union and accepted by the Trustees; as to such Union personnel, the Union shall be considered an Employer solely for the purpose of contributions within the meaning of the Trust Agreement, as herein defined, and shall, on behalf of such personnel, make payments to the Trust Fund at the same times and at the rate of payment equal to that made by any other Employer who is a party to the Trust Agreement;
- (c) All persons employed by Central States, Southeast and Southwest Areas Health and Welfare Fund or Central States, Southeast and Southwest Areas Pension Fund upon acceptance by the Trustees; as to such personnel, the Trustees shall be deemed an Employer solely for the purpose of contributions within the meaning of the Trust Agreement, on behalf of such personnel, make payments to the Trust Fund at the times and at the rate of payment equal to that made by any other Employer who is a party to the Trust Agreement; or
- (d) All persons who are Trustees of Central States, Southeast and Southwest Areas Health and Welfare Fund or Central States, Southeast and Southwest Areas Pension Fund upon acceptance by the Trustees; on behalf of such persons who are Trustees, their Employers shall make contributions to the Trust Fund at the times and at the rate of payment equal to that required by any other Employer who is a party to the Trust Agreement.

In all instances the common-law test for, or the applicable statutory definition of, master-servant relationship shall control Employee status. The continuation of Employee status shall be subject to such rules as the Trustees may adopt.

1.28 Employer:

Any Employer (including an Association of Employers) who is or becomes a party to a Collective Bargaining Agreement and who, with the acquiescence of the Trustees, agrees to be bound by the Trust Agreement and this Plan and is accepted for participation in the Plan by the Trustees, subject to such rules as the Trustees may in their discretion adopt. The Local Union, the Health and Welfare Fund and the Pension Fund shall be deemed to be Employers of those persons employed and proposed by such organizations and accepted by the Trustees.

1.29 Employer Contributions:

Contributions made by Employers to the Fund; contributions made by the Local Union or the Fund on behalf of their Employees; and, amounts set aside by the Fund on behalf of its Employees.

1.30 Employer Obligation Date:

The date on which an Employer first becomes obligated, by reason of contract or agreement with the Fund, to make contributions to the Fund on behalf of a group of Employees.

1.31 Family Medical Leave:

A voluntary absence from work taken by an employee pursuant to the provisions of the federal Family Medical Leave Act.

1.32 Former Covered Participant:

A person who was a Covered Participant but presently has no Coverage.

1.33 Fund:

The Central States, Southeast and Southwest Areas Health and Welfare Fund Active Plan (Non-Grandfathered) as set forth herein and as hereafter amended.

1.34 Health Maintenance Organization (HMO):

An organization operating as a Health Maintenance Organization.

1.35 Hospital:

A facility licensed by the State or jurisdiction in which it is located with inpatient beds and medical services, including physician services and continuous nursing services under the supervision of registered nurses, to provide diagnosis and medical or surgical treatment primarily for but not limited to acutely ill patients and accident victims, to provide

treatment for patients with mental illness or to provide treatment in special inpatient care facilities.

1.36 Hospital Confinement:

A hospital stay of at least overnight duration. An emergency room visit is not part of a Hospital Confinement unless it leads directly to a stay in a hospital room.

1.37 Hourly Bank

An amount (in dollars) used to determine eligibility under an Hourly Plan based on the total value of contributions received on behalf of a Covered Participant in excess of the amount necessary to maintain eligibility.

1.38 Hourly Plan

Any Plan covering a group under a Collective Bargaining Agreement accepted by the Fund for participation which provides for contributions based on the hours worked by Employees and under which eligibility is determined on a monthly basis.

1.39 Immediate Coverage:

Full entitlement to all benefits of this Plan without fulfilling the Initial Contribution Period.

1.40 Initial Contribution Period:

A period during which an Employer is obligated to make Employer Contributions for a Participant and ending on the Sunday of the eighth (8th) week thereafter, provided that the eight (8) weeks of Employer Contributions used to determine the Initial Contribution Period need not be consecutive but must all be made during a period of 52 consecutive weeks. This rule is retroactive to January 1, 2016 provided, however, that Participants who have Employer Contributions during the week ending December 26, 2015 may establish Coverage either: (1) under the prior version of Section 1.40 in effect as of January 1, 2015 with an initial Contribution Period of 8 weeks of consecutive Employer Contributions; or (2) under the current provision with an Initial Contribution Period beginning with the first week of Employer Contributions on or after December 27, 2015. The Initial Contribution Period for any Participant who did not have an Employer Contribution during the week ending December 26, 2015 shall begin with the first Employer Contribution on or after December 27, 2015.

1.41 Lay-Off:

An involuntary separation from employment caused by an Employer suspending Employees. Individuals shall not be deemed on Lay-off if they engage in gainful employment for any other employer, nor shall Lay-Off status continue when

an individual retires or otherwise terminates the employment relationship.

1.42 Leave of Absence:

An Employee's voluntary temporary absence from employment, approved by the Employer. Individuals on Leave of Absence shall not engage in gainful employment for any other employer, nor shall Leave of Absence status continue when an individual retires.

1.43 Life Insurance Benefit:

- (a) The benefit payable at a Covered Participant's death as set forth in Article XIV and Section 20.04 of the applicable plan.
- (b) The benefit payable to a Covered Participant upon the death of a Dependent Spouse or Child as set forth in Article XIV and Section 20.04 of the applicable plan.

1.44 Local Union:

Those Local Unions affiliated with the International Brotherhood of Teamsters who have executed Collective Bargaining Agreements which require contributions to be made to the Fund on behalf of the covered employees, and such other unions as the Trustees may agree upon.

1.45 Short-Term Disability Benefit:

The benefits set forth in Sections 12.02 and 20.01(a).

1.46 Short-Term Disability Coverage:

Coverage for a Covered Participant and his eligible Covered Dependents which may be available during periods when the Covered Participant is eligible for the Short-Term Disability Benefit. See Section 20.01(a) for application, if any.

1.47 Maintenance Care:

Maintenance Care is care provided to a person who needs assistance or support for activities of daily living but who is not under a course of treatment which will improve his condition to the extent necessary to enable him to function without such assistance or support, except for care which is necessary to treat a curable illness. A Maintenance Care determination is not precluded by the fact that a patient is under the care of a Physician and that the services are provided at the Physician's request.

1.48 Major Medical Expense Benefit:

The benefits set forth in Article XIII and Section 20.01(p).

1.49 Medical Out-of-Pocket Expense Limit:

A maximum liability per Covered Individual or per family per calendar year as set forth in Article XVII and Section 20.05.

1.50 Multi-tier Plan

Any Plan that permits the Covered Participant to elect which Dependent(s), (if any) of the Covered Participant shall be Covered Dependents.

1.51 New Group:

A group of Employees who become Participants when their Employer first becomes obligated to make Employer Contributions on their behalf.

1.52 New Participant:

An Employee who becomes a Participant by joining a group after the Employer of the group became obligated to make Employer Contributions to the Fund on behalf of the group.

1.53 Other Plan:

Any group plan, insurance policy or contract which provides benefits for hospital, surgical, dental, psychiatric, chiropractic or other medical treatment, and any plan or insurance coverage hereafter described in subparagraph (f). Other Plan includes a plan providing benefits through:

- (a) Group blanket or franchise insurance coverage;
- (b) Group health provider, group practice or other prepayment coverage;
- (c) Any coverage under labor-management trusteed plans, union welfare plans, employer organizations or employee benefits organization plans;
- (d) Any coverage under government programs or any coverage required or provided by statute;
- (e) Any other arrangement providing hospital, surgical, dental, psychiatric, chiropractic or other medical treatment for members of a group; and
- f) No fault, personal injury protection or financial responsibility motor vehicle insurance coverage which provides benefits to or for a Covered Individual for bodily or psychological injury, including but not limited to, benefits for hospital, surgical, dental, psychiatric, chiropractic and other medical treatment.

The term "Other Plan" shall be construed separately with respect to each policy or other provision thereof, sub-plan,

contract or other arrangement for benefits or services and separately with respect to that portion of any such policy or other provision thereof, sub-plan, contract or other arrangement (whether a separate plan or not) which reserves the right to take the benefits or services of Other Plans into consideration in determining its benefits and that portion which does not.

1.54	Participant:	An Employee for whom an Employer is obligated to make Employer Contributions or an Employee who is entitled to and who does make Self-Payments to the Fund.
1.55	Pension Fund:	The Central States, Southeast and Southwest Areas Pension Fund.
1.56	Physician:	A legally qualified and licensed Physician. A Physician does not include a Chiropractor.
1.57	Plan:	The Central States, Southeast and Southwest Areas Health and Welfare Active Plan (Non-Grandfathered) as set forth herein and as hereafter amended.
1.58	Plan Benefit Limit:	A maximum payout per calendar year for combined benefits, not taking into account the Retiree Contribution Benefit, as set forth in Article XVIII and Section 20.06.
1.59	Prescription Drug:	A drug or medicine prescribed by a Physician or Dentist, dispensed by a pharmacist, not available over the counter (except for insulin and insulin syringes) and bearing the Federal or State Legend.
1.60	Prior Carrier:	Any insurance company, welfare plan or other entity which provided life, hospital, surgical, dental or medical coverage for a group of Employees and their Dependents immediately before the group of Employees became Participants under

Part 164, Subparts A and E.

The Standards for Privacy of Individually Identifiable Health Information published at 45 C.F.R. Part 160 and 45 C.F.R.

this Plan.

1.61 Privacy Rule:

1.62 Protected Health Information:

1.63 Psychiatric Treatment Facility:

Shall have the same meaning as the term "protected health information" as defined at 45 C.F.R § 164.501.

A facility that is:

- (a) Engaged in providing, under the supervision of a Physician, psychiatric services for the diagnosis and treatment of mentally ill persons; and
- (b) Licensed, certified or approved by the state or jurisdiction in which it is located to provide psychiatric services.

- 1.64 (Reserved)
- 1.65 Qualified Medical Child Support Order:

Any order entered by a court of competent jurisdiction that complies with requirements of the federal Qualified Medical Child Support Act and which requires coverage for one or more dependent children.

1.66 Qualifying Event:

Any of the events described in Section 3.16 which would result in a loss of coverage of the Covered Individual unless continuation coverage is elected.

1.67 Quit:

A permanent termination of employment initiated by the Employee.

1.68 Reasonable and Customary:

The usual, Reasonable and Customary charge for the treatment, supply, or service, determined by comparison with the charges customarily made for similar treatments, supplies or services to individuals with similar medical conditions within a given geographical area.

1.69 Retiree Contribution Benefit:

The benefit set forth in section 12.21.

1.70 Retiree Plan:

As applicable, (a) the Central States, Southeast and Southwest Areas Health and Welfare Retiree Plan as amended from time to time, (b) the Central States, Southeast and Southwest Areas Health and Welfare UPS Retiree Plan as amended from time to time, (c) the Central States, Southeast and Southwest Areas Health and Welfare UPS Retiree RU Plan as amended from time to time or (d) the Central States, Southeast and Southwest Areas Health and Welfare UPS Retiree RV Plan as amended from time to time.

- 1.71 Self-Payments:
- 1.72 Service in the Uniformed Services:

Contributions to the Fund under this Plan by a Participant on his own behalf.

Service by a Covered Participant in the Uniformed Services, which means and includes the Armed Forces, the Army National Guard and the Air National Guard when engaged in active duty for training, inactive duty training, or full-time National Guard duty, the commissioned corps of the Public Health Service, and any other category of persons designated by the President in time of war or emergency, provided that such services includes the performance of duty by the Covered Participant on a voluntary or involuntary basis in a Uniformed Service under competent authority and also includes any period during which a Covered Participant is absent from employment by an Employer for the purpose of an examination to determine the Covered Participant's fitness to perform any such duty, and provided further that such service results in the absence of the Covered Participant from Continuous Funded Employment by an Employer.

1.73 Sick Leave:

A temporary absence from work caused by an Employee's illness, injury or pregnancy.

1.74 Spouse:

An individual who is married to a Participant in a legally recognized civil or religious ceremony. A Participant's common-law Spouse shall be considered a Spouse for purposes of the Plan, if:

- (a) The Participant's state of domicile recognizes commonlaw marriage; and
- (b) The Participant furnishes the Fund with appropriate documentation that the couple has fulfilled all conditions which his state of domicile requires for such a marriage.
- 1.75 Standard Medical Care, Treatment, Services or Supplies:

Care, treatment, services or supplies which are generally accepted by the medical community as Standard Medical Care, Treatment, Services or Supplies.

1.76 Surrogate Mother:

A Surrogate Mother is a woman who, before she becomes pregnant, has agreed by contract or other understanding (with or without compensation) to bear a child/children (including when the woman provides the ovum) that will be given up to another person or persons to raise following birth.

1.77 TeamCare:

A program of preferred providers who agree to negotiated rates for medical services and supplies for the Fund, in exchange for which the Fund provides financial incentives for Participants to use the services of these providers. The Fund publishes a list of TeamCare providers periodically, as well as the benefit modifications which apply and the areas covered by a TeamCare network.

1.78 Teamsters:

The International Brotherhood of Teamsters, and its affiliated Local Unions.

1.79 Temporary Work Stoppage:

A strike by Participants which is sanctioned by the Teamsters.

1.80 Terminated Employee:

An individual who is separated from his employment by reason of Quit or Discharge.

1.81 Total and Permanent Disability:

A disease or bodily injury which will permanently, continuously and wholly prevent a person from engaging in any occupation or employment for wage or profit for the duration of his life. Additionally, the complete and irrecoverable loss of the sight of both eyes, or the use of both hands, or of both feet, or of one hand and one foot.

1.82 Total and Permanent Disability Installment Benefit:

The benefit set forth in Sections 14.05 and 20.04 provided to those Covered Participants under the age of fifty (50), who become totally and permanently disabled, as defined in Section 1.81 of this Document.

1.83 Trust Agreement:

The Agreement and Declaration of Trust made and entered into on the fourteenth (14th) day of March, 1950, by and between Central Conference of Teamsters, Southern Conference of Teamsters and their affiliated Local Unions, and the Southeastern Area Motor Carriers Labor Relations Association; Southwest Operators Association; and Motor Carriers Employers Conference-Central States, and as amended from time to time thereafter by the Trustees.

1.84 Trustees:

The Trustees designated and appointed in accordance with the terms of the Trust Agreement.

1.85 Trust Fund:

All assets, including principal and interest, of the Active Subaccount of Trust created by the Trust Agreement.

1.86 Vision Benefits:

The benefits set forth in Article XVI and Section 20.03.

1.87 Waiver of Premium Disability Benefit:

The benefit set forth in Sections 14.06 and 20.04 provided to those Covered Participants, between the ages of fifty (50) and fifty-nine (59) inclusive, who become totally and permanently disabled by bodily injury or disease, as defined in Section 1.81 of this document.

ARTICLE II. EFFECTIVE DATE OF GENERAL PROVISIONS

2.01 THE TERMS AND PROVISIONS OF THIS PUBLISHED EDITION OF THIS PLAN DOCUMENT ARE APPLICABLE TO THE COVERAGE OF EVERY COVERED INDIVIDUAL WHO HAS BEEN, IS OR HEREAFTER BECOMES ENTITLED TO COVERAGE BY THE PLAN AS OF ANY DATE ON OR AFTER JANUARY 1, 2022, PROVIDED THAT THE TERMS, PROVISIONS, LIMITATIONS AND EXCLUSIONS OF COVERAGE AS OF ANY DATE PRIOR TO JANUARY 1, 2022, ARE GOVERNED BY THE EARLIER EDITION OF THIS PLAN DOCUMENT THAT WAS IN EFFECT ON THAT DATE (INCLUDING ALL PLAN AMENDMENTS OF THAT EDITION ADOPTED AND IN EFFECT ON THAT DATE), AND PROVIDED FURTHER THAT THE TERMS AND PROVISIONS OF THIS EDITION ALSO INCLUDE ALL PLAN AMENDMENTS ADOPTED AND IN EFFECT AFTER PUBLICATION OF THIS EDITION (EVEN IF THEY ARE YET TO BE INCORPORATED IN THIS EDITION).

ARTICLE III. PARTICIPATION AND COVERAGE

3.01 QUALIFICATIONS AS A COVERED PARTICIPANT

Unless otherwise provided herein, a Participant shall become a Covered Participant when Employer Contributions have been made on his behalf by an Employer, to the Fund or to any Other Plan having a Reciprocity Agreement in effect with the Fund, for a period that equals or exceeds his Initial Contribution Period and, unless otherwise provided herein, a Participant shall be a Covered Participant only in the period for which his Employer is required to make Employer Contributions on his behalf pursuant to the Collective Bargaining Agreement or applicable law.

3.02 QUALIFICATIONS AS A COVERED DEPENDENT

Unless otherwise provided herein, a Dependent shall be considered a Covered Dependent when the person upon whom he is dependent is considered a Covered Participant provided the Covered Participant has elected the dependent coverage option under his plan.

3.03 COMMENCEMENT OF COVERAGE OF MEMBERS OF A NEW GROUP PREVIOUSLY INSURED

If an Employer of a New Group had provided group health coverage for members of the New Group and their Dependents immediately prior to the Employer Obligation Date, an Active Employee on the Employer Obligation Date shall receive Immediate Coverage retroactive to the Employer Obligation Date.

A member of the New Group who is not an Active Employee on the Employer Obligation Date shall receive Immediate Coverage on the date he becomes an Active Employee provided the Employer informed the Fund that the Employee was in a valid inactive status on the Employer Obligation Date.

3.04 COMMENCEMENT OF COVERAGE OF MEMBERS OF A NEW GROUP NOT PREVIOUSLY INSURED

If an Employer of a New Group had not provided group health coverage for members of the New Group and their Dependents immediately prior to the Employer Obligation Date, members of the New Group shall have Coverage under the Plan as follows:

- (a) A member of the New Group who is an Active Employee on the Employer Obligation Date shall have Coverage under the Plan on the first (1st) day following the Initial Contribution Period, provided the Fund receives from the Employer an initial list of Active Employees (as of the Employer Obligation Date) on the "509-New Group Report" form which includes the name of such Active Employee.
- (b) A member of the New Group who is not an Active Employee shall become a Participant only upon becoming an Active Employee, at which time his Initial Contribution Period shall commence and his Coverage shall be determined in accordance with the rules in Section 3.05.

3.05 COMMENCEMENT OF COVERAGE OF NEW PARTICIPANTS

A New Participant shall have Coverage under the Plan on the first (1st) day following the Initial Contribution Period, provided the Participant is an Active Employee on this day. The Fund shall accept Self-Payments from the Participant during the Initial Contribution Period only after the first contribution has been made by the Employer.

A New Participant who previously had Coverage and then terminated employment with an Employer in order to enter Service in the Uniformed Services shall receive Immediate Coverage upon his return to covered employment provided it is within twenty-six (26) weeks of his discharge from Service in the Uniformed Services.

A New Participant whose Coverage is terminated and returns to Coverage within fifty-two (52) weeks of the date that his previous Coverage (including Short-Term Disability Coverage) terminated shall have Immediate Coverage. This rule is retroactive to January 1, 2016 and applies to any Participant whose Coverage was terminated on or after June 28, 2015. Any Participant whose Coverage was terminated on or before June 27, 2015 shall have Coverage following an Initial Contribution Period as required by Section 3.01.

A Former Covered Dependent who goes to work for an Employer who is obligated to make contributions to the Fund shall receive Immediate Coverage upon the date the Employer becomes required to make contributions on behalf of this former Covered Dependent, provided such contributions are required to begin within sixty (60) days after the termination of Dependent Coverage.

3.06 COMMENCEMENT OF COVERAGE OF COVERED PARTICIPANTS FOR WHOM UNINTERRUPTED EMPLOYER CONTRIBUTIONS AND/OR SELF-PAYMENTS HAVE BEEN RECEIVED

If the Fund has received uninterrupted Employer Contributions and/or Self-Payments on behalf of a Covered Participant, such Covered Participant shall have Immediate Coverage when he commences employment for any Employer required to make contributions under the Plan on his behalf.

3.07 COMMENCEMENT OF COVERAGE OF EMPLOYEES COVERED UNDER ANOTHER PLAN OFFERED BY THE FUND

An Employee and his Dependents covered under one plan offered by the Fund shall be considered a Covered Participant and Covered Dependents under a successor plan offered by the Fund on the date the Employer of the Covered Participant becomes obligated to make the amount of contributions required for the successor plan, with the exception of the Total and Permanent Disability Installment Benefit.

- (a) Payment of the Short-Term Disability Benefit will be governed by the terms and conditions of the successor plan if the Disability began while the first plan was in effect.
- (b) A Participant under one plan offered by the Fund who is making Self-Payments in accordance with that Plan may not elect to gain Coverage under a different plan by increasing or decreasing the level of his Self-Payments.

(c) Payment of the Total and Permanent Disability Installment Benefit will be governed by the terms and conditions of the first plan if the Disability began while the first plan was in effect.

3.08 COMMENCEMENT OF COVERAGE OF NEW PARTICIPANTS ELIGIBLE UNDER A PLAN OFFERED BY A LOCAL UNION

A New Participant who has, through his previous Employer, established eligibility for benefits under a health and welfare plan offered by a Local Union affiliated with the Teamsters and not offered by the Fund, shall receive Immediate Coverage on the date he begins to work for an Employer, subject to the following conditions:

- (a) His coverage must be continuous, and
- (b) On the date he begins to work for an Employer, there must be a reciprocity agreement in effect between the Fund and the Local Union.

3.09 EFFECT OF TEMPORARY WORK STOPPAGE ON COVERAGE

Unless the Trustees in their sole discretion decide at any time to disallow or terminate Plan Coverage before or during a specific Temporary Work Stoppage, a Covered Participant who is absent from employment because of a Temporary Work Stoppage (but not Covered Participants who elect to return to work during the Temporary Work Stoppage) shall receive Plan Coverage regardless of whether contributions to the Fund are actually made, subject to the following conditions:

- (a) The Temporary Work Stoppage must be sanctioned by the Teamsters (or by the union representing the bargaining unit if not a union affiliated with the Teamsters);
- (b) Except for a Temporary Work Stoppage by Employees subject to the National Master Freight Agreement, the Local Union involved must provide the Fund with confirmation of the sanctioning, the inception and termination dates thereof, and a list of Employers and Employees involved in said Temporary Work Stoppage; and
- (c) The Covered Participant must be an Active Employee when the Temporary Work Stoppage begins.

A Participant who has not established Coverage and who is absent from employment because of a Temporary Work Stoppage shall, unless the Trustees in their sole discretion decide at any time to disallow or terminate Plan Coverage before or during a Temporary Work Stoppage, receive Coverage on the first (1st) day after the Initial Contribution Period would have ended had the Employer remained obligated to make Employer Contributions throughout the Temporary Work Stoppage, regardless of whether contributions to the Fund are actually made subject to the conditions set forth in (a) and (b) above. If a Temporary Work Stoppage is in progress before the Trustees decide to disallow or terminate Plan Coverage during that Temporary Work Stoppage, that decision by the Trustees will be applied prospectively and will not result in a loss of Coverage during periods prior to the decision.

3.10 EFFECT ON COVERAGE OF ABSENCE FROM CONTINUOUS FUNDED EMPLOYMENT DURING, AND AS A RESULT OF, SERVICE IN THE UNIFORMED SERVICES

A Covered Participant who is absent from Continuous Funded Employment by an Employer for a period of less than thirty-one (31) days during, and as a result of, Service in the Uniformed Services, shall receive Coverage for himself and all of his Covered Dependents throughout such period, without making Self-Payments, regardless of whether Employer Contributions are actually made to the Fund during such period. A Covered Participant who is absent from Continuous Funded Employment for a period of more than thirty (30) days during, and as a result of, Service in the Uniformed Service, shall be eligible (as will all of his Covered Dependents) to elect Continuation Coverage in accordance with the provisions of Sections 3.15 through 3.21. If Coverage of a Covered Participant (and his Covered Dependents) is terminated at any time during his absence from Continuous Funded Employment by an Employer as a result of Service in the Uniformed Services and he makes a timely application for reemployment by the same Employer at the conclusion of such Service in the Uniformed Services, any such reinstatement of Coverage of the Covered Participant (and his Covered Dependents) shall be Immediate Coverage, provided that there shall be no such Immediate Coverage relative to any illness or injury of the Covered Participant which is determined by the Secretary of Veterans Affairs to have been incurred or aggravated during performance of Service in the Uniformed Services. As used in the preceding sentence, "a timely application for reemployment by the same Employer" means such application within the following time limitations (except as those limitations are required by law to be extended):

- (a) within ninety (90) days after completion of a period of Service in the Uniformed Services that was more than one hundred eighty (180) days; and
- (b) within thirty (30) days after completion of a period of Service in the Uniformed Services that was more than thirty (30) days and less than one hundred eighty-one (181) days.

3.11 TERMINATION OF COVERAGE OF PARTICIPANTS HAVING TOTAL AND PERMANENT DISABILITY OR ON SICK LEAVE

A Covered Participant who has a Total and Permanent Disability or who goes on Sick Leave shall lose his Coverage after the Saturday of the last week for which Employer Contributions are required to be made on his behalf under the applicable law or Collective Bargaining Agreement, unless Short-Term Disability Coverage is in effect (see Section 20.01(a) for application, if any). Upon exhaustion of his Coverage, such Former Covered Participant may maintain Coverage if he elects to make Self-Payments pursuant to Section 3.20.

3.12 TERMINATION OF COVERAGE OF PARTICIPANTS CHANGING EMPLOYEE STATUS

A Covered Participant who goes on Lay-Off or a Leave of Absence, whose employment relationship is terminated, or whose employer is the subject of a voluntary or involuntary petition in bankruptcy shall lose his Coverage after the Saturday of the last week for which Employer Contributions are required to be made on his behalf or (in the case of bankruptcy) after the Saturday of the week of filing. A Former Covered Participant may maintain Coverage by making Self-Payments pursuant to the Plan's Continuation Coverage provisions.

3.13 TERMINATION OF COVERAGE DUE TO EMPLOYER WITHDRAWAL

A Covered Participant shall lose his Coverage after the Saturday of the last week for which Employer Contributions are required to be made on behalf of any Employees of the Employer of the Covered Participant, including a cessation of Employer Contributions that results from a rejection by the Trustees, acting pursuant to the Trust Agreement, of a Collective Bargaining Agreement of the Employer. Such a Former Covered Participant may not maintain Coverage by making Self-Payments, and the Fund will not accept Self-Payments from such a Former Covered Participant, except as authorized by any other applicable section of this Plan.

3.14 RETURN TO COVERED EMPLOYMENT

A Former Covered Participant who returns to covered employment that requires Employer Contributions on his behalf may re-establish eligibility for Coverage by making Self-Payments for periods in which his status permits him to do so. All Employer Contributions and/or Self-Payments must be submitted as set forth in Section 3.05 of this Plan Document. Any Self-Payments made pursuant to this section must be received within forty-five (45) days after the week to which they are to be applied.

3.15 ELIGIBILITY TO ELECT CONTINUATION COVERAGE

Each Covered Individual who would lose Coverage under this Plan as a result of a Qualifying Event, as defined in Section 3.16, is eligible to elect Continuation Coverage, as defined in Section 3.18, upon compliance with notice requirements and election procedures described in Sections 3.17 and 3.19.

3.16 **DEFINITION OF QUALIFYING EVENT**

For purposes of Sections 3.15 through 3.21, the term "Qualifying Event" means, with respect to any Covered Individual, any of the following events which, unless there is an election of Continuation Coverage pursuant to Section 3.19, would result in a loss of Coverage of the Covered Individual under this Plan:

- (a) the death of a Covered Participant;
- (b) the termination, or reduction of hours, of a Covered Participant's employment (including an absence and/or separation from employment caused by Sick Leave, Disability, Leave of Absence, Lay-Off, Quit, Discharge and the pendency of any bankruptcy proceeding filed by or against the Covered Participant's Employer; and including an absence and/or separation from employment, for a period of more than thirty (30) days, during and as a result of Service in the Uniformed Services);
- (c) the divorce or legal separation of a Covered Participant from the Covered Participant's Spouse;
- (d) the commencement of entitlement of a Covered Participant to Medicare benefits under Title XVIII of the Social Security Act (42 U.S.C. § 1395 et seq.); and
- (e) the termination of Coverage of a Child as a Covered Dependent relative to a Covered Participant, under generally applicable terms of this Plan, including Sections 3.29 and 3.30.

3.17 **NOTICE OF QUALIFYING EVENT**

With respect to each Qualifying Event defined in Section 3.16, notice shall be provided in accordance with the following requirements:

- (a) General notice by the Fund -- at the time of commencement of Coverage of a Covered Participant under this Plan, the Fund shall provide written notice to the Covered Participant and his or her Spouse of their rights pursuant to Sections 3.15 through 3.21.
- (b) Specific notice by Employers -- within sixty (60) days after a Qualifying Event defined in any of Sections 3.16(a), 3.16(b) and 3.16(d), the Employer (relative to any Covered Participant affected by the event) shall provide written notice of the Qualifying Event to the Fund.
- (c) Specific notice by Covered Individuals -- within sixty (60) days after a Qualifying Event defined in either of Sections 3.16(c) and 3.16(e), either the Covered Participant or another Covered Individual affected by the event (or both) shall provide written notice of the Qualifying Event to the Fund.
- (d) Specific notice by the Fund -- within sixty (60) days after notice of a Qualifying Event has been provided to the Fund in compliance with subsection (b) or (c) of this section, the Fund shall provide written notice of the Qualifying Event to the Covered Participant affected by the event (if living), to his or her Spouse and to any other Covered Individual who is both affected by the event and not residing with either the Covered Participant or the Covered Participant's Spouse.

3.18 ELIGIBILITY FOR CONTINUATION COVERAGE

After the occurrence of a Qualifying Event and upon timely election pursuant to Section 3.19, a Covered Individual is eligible to receive Continuation Coverage until termination of eligibility pursuant to Section 3.21, subject to the same modifications of coverage that apply to Covered Individuals to whom no Qualifying Event has occurred. As an alternative, the affected Covered Individual may elect to receive a reduced plan of coverage.

3.19 PROCEDURES TO ELECT CONTINUATION COVERAGE

After occurrence of a Qualifying Event and timely notice to the Fund is received in compliance with Section 3.17(b) or Section 3.17(c), a Covered Individual may elect Continuation Coverage by providing written election of Continuation Coverage to the Fund, in such form as the Fund prescribes, before expiration of the "election period." The "election period" begins on the date of the Qualifying Event and ends on the sixtieth (60th) day after the date of the Qualifying Event or, if later and if the Fund received timely notice of the Qualifying Event pursuant to Section 3.17(b) or Section 3.17(c), on the sixtieth (60th) day after notice of Qualifying Event is provided by the Fund to the Covered Individual pursuant to Section 3.17(d). A timely election shall be deemed to include Continuation Coverage for each Covered Individual who would otherwise lose coverage as a result of the Qualifying Event.

3.20 SELF-PAYMENTS TO MAINTAIN CONTINUATION COVERAGE

If a written election of Continuation Coverage is provided to the Fund in compliance with Section 3.19, Self-Payments must be remitted to the Fund on behalf of the Covered Individuals on whose behalf the election is made, in order to maintain their eligibility for Continuation Coverage. Self-Payments must be remitted for all periods of Continuation Coverage in such amounts, form and manner as the Fund prescribes, except that the Fund will not require any Self-Payments to be remitted prior to expiration of forty-five (45) days after the initial written election of Continuation Coverage is provided to the Fund in compliance with Section 3.19. A Covered Participant who is absent from employment by a Contributing Employer for a period of less than thirty-one (31) days during, and as a result of, Service in the Uniformed Services shall receive coverage for himself and all of his Covered Dependents throughout such period without making Self-Payments.

3.21 TERMINATION OF CONTINUATION COVERAGE

Continuation Coverage provided to a Covered Individual pursuant to an election in compliance with Section 3.19 shall begin on the date of the Qualifying Event and shall terminate on the earliest of:

- (a) For a Qualifying Event defined in Section 3.16(b), the date twenty-four (24) months after the Qualifying Event; except, that if during the initial period, another Qualifying Event, as defined in Section 3.16, occurs, the termination date is the date thirty-six (36) months after the initial Qualifying Event;
- (b) For a Qualifying Event defined in Section 3.16 other than Section 3.16(b), the date thirty-six (36) months after the date of the Qualifying Event;
- (c) The date which is thirty-one (31) days after the date on which Self-Payments pursuant to Section 3.20 (and related procedures) are owed to the Fund but remain unpaid (unless such Self-Payments are remitted to the Fund within such thirty-one (31)-day period);
- (d) The date on which the Covered Individual first becomes, after the date of the election, covered under any other employee welfare benefit plan providing coverage for medical care (as defined in Section 213[d] of Title 26 of the United States Code) to participants and beneficiaries directly or through insurance, reimbursement or otherwise, if such Other Plan does not contain any applicable exclusion or limitation with respect to any preexisting condition of the Covered Individual, except that, on such date of coverage by another plan, the Continuation Coverage of only Covered Individuals covered under such other employee welfare benefit plan will be terminated, subject to the provisions of Section 3.30;
- (e) The date on which the Covered Individual first becomes, after the date of the election, entitled to Medicare benefits under Title XVIII of the Social Security Act (42 U.S.C. § 1395 et seq.), except that on such date the Continuation Coverage of only the Covered Individual thus entitled to Medicare benefits will be terminated; or
- (f) the date on which this Plan is terminated.

Continuation Coverage for a Qualifying Event described in Section 3.16(b) may be extended to twentynine (29) months if a Covered Individual is determined under Title II or XVI of the Social Security Act to have been disabled during the first sixty (60) days of Continuation Coverage and if a Covered Individual notifies the plan administrator of the determination within sixty (60) days of the determination and before the end of the initial twenty-four (24)-month period of Continuation Coverage. This extension will terminate if there is a final determination under Title II or XVI of the Social Security Act that the disability has ended, but the termination will apply only to the formerly disabled Covered Individual. The termination of the extension in such a case will be effective on the first (1st) day of the first (1st) month which begins more than thirty (30) days after such final determination, so long as the final determination is after the twenty-four (24)-month initial Continuation Coverage period.

3.22 STATUS OF A COVERED PARTICIPANT WHEN EMPLOYER FAILS TO MAKE CONTRIBUTIONS

Default by an Employer in making Employer Contributions on behalf of a Participant shall not prevent such person from being a Covered Participant, except in the case of a Suspension of Benefits as described in Section 3.23.

3.23 STATUS AND COVERAGE OF A COVERED INDIVIDUAL WHEN EMPLOYER FAILS TO REMIT EMPLOYER CONTRIBUTIONS: SUSPENSION OF BENEFITS

Except as otherwise provided herein, any default by an Employer in remitting Employer Contributions on behalf of a Participant who is otherwise entitled to Coverage shall not disqualify such Participant as a Covered Participant. However, all benefits payable by the Plan to or on behalf of a Participant or his Dependent shall be suspended during certain periods, established by the Trustees, in which the Employer of such Participant remains continuously delinquent in his obligations to remit Employer Contributions. During such benefits suspension periods the Participant shall have the right to remit Self-Payments subject to rules established by the Trustees. No benefits suspension period shall commence until the expiration of a reasonable and adequate period (as determined by the Trustees) and after advance written notice of such suspension to all affected Participants. After any benefits suspension period is concluded by remittance of all delinquent Employer Contributions or by alternative commitments by the Employer approved by the Trustees, all proper benefit claims accrued but unpaid during the suspension period will be paid by the Plan to or on behalf of Covered Participants and their Covered Dependents and all Self-Payments will be reimbursed.

3.24 CERTAIN PARTICIPANTS NOT TO BE CLASSIFIED AS COVERED DEPENDENT

A Participant's Child shall not be a Covered Dependent if that Child is also a Participant. A Participant's Spouse shall not be a Covered Dependent if that Spouse is also a Participant, except as provided in Article V

3.25 COMMENCEMENT OF COVERED DEPENDENT STATUS FOR SPOUSE OF A COVERED PARTICIPANT

The Spouse of a Covered Participant shall be a Covered Dependent (a) on the date of marriage or (b) on the date the Participant becomes a Covered Participant, whichever ((a) or (b)) comes later, provided the Covered Participant has elected the Spouse option under his or her plan.

3.26 COMMENCEMENT OF COVERED DEPENDENT STATUS FOR CHILD OF A COVERED PARTICIPANT

A Child of a Covered Participant shall be a Covered Dependent and shall participate in and have Coverage under this Plan on the date such status as Child begins or on the date the Participant becomes a Covered Participant, whichever occurs later, provided the Covered Participant has elected the dependent coverage option under his plan.

3.27 STATUS OF MENTALLY OR PERMANENTLY PHYSICALLY DISABLED CHILD OF A COVERED PARTICIPANT

A Covered Participant's unmarried Child who is incapable of independent financial self-support because of mental or permanent physical disability and who is dependent on the Covered Participant for support and maintenance, shall remain a Covered Dependent or be eligible to receive benefits (except the Life Insurance Benefit) beyond attaining twenty-six (26) years of age subject to the following conditions:

- (a) The Child must be a Covered Dependent prior to reaching twenty-six (26) years of age and may not be covered under any Other Plan (other than, for purposes of this Section only, Medicare and Medicaid) when the child reaches twenty-six (26) years of age.
- (b) After the Child reaches twenty-six (26) years of age, the Fund may, at its discretion and at any time, require that a certification, signed by a licensed Physician and stating that the Child is mentally or permanently physically disabled, accompany each claim.
- (c) In no case will the Child be a Covered Dependent or be eligible to receive benefits unless the mental or permanent physical disability is sustained prior to the Child's twenty-sixth (26th) birthday and the Child's Disability has been continuous.
- (d) A Child otherwise meeting the requirements of this Section shall no longer be a Covered Dependent or be eligible for Coverage under this Section if he becomes covered under any Other Plan, provided for purposes of this Section only, that Medicare and Medicaid shall not be considered an Other Plan. Once coverage under this Section is terminated, a Child shall not be or become eligible for further Coverage under this Section.

3.28 STATUS OF NEWBORN CHILD OF A COVERED PARTICIPANT

A Covered Participant's newborn Child shall become a Covered Dependent or eligible to receive benefits at the time of birth.

3.29 STATUS OF CHILD OF A COVERED PARTICIPANT

A Covered Participant's Child will maintain Coverage under the Plan, until the date he reaches twenty-six (26) years of age.

3.30 LOSS OF DEPENDENT'S COVERAGE

A Covered Dependent shall lose his Coverage on the earliest of the dates below:

- (a) On the date the Covered Participant loses Coverage. The Spouse of the deceased Covered Participant may elect to make Self-Payments in accordance with the terms and procedures specified in Section 3.20;
- (b) On the contribution due date, if the Covered Participant is making Self-Payments and fails to do so;
- (c) In the case of a Spouse, on the day which said Spouse ceases to be legally married to a Covered Participant;
- (d) Subject to the provisions of Sections 3.27, on the day a Child reaches twenty-six (26) years of age.

3.31 RESIDUAL COVERAGE OF FORMER COVERED PARTICIPANTS

In the remaining provisions of the Plan, the term "Covered Participant" shall be extended to include a Former Covered Participant, and the term "Covered Individual" shall be extended to include a Former Covered Participant and/or his Dependents during the period that such Former Covered Participant and/or his Dependents were eligible to receive benefits provided according to Plan provisions.

3.32 ELIGIBILITY FOR FAMILY PROTECTION PLAN BENEFIT

A Covered Dependent shall be eligible, subject to the conditions described below, for a family protection plan benefit, consisting of the extension of coverage of Covered Dependents of a Participant who dies while residing in an area covered by a TeamCare network or while recorded by the Fund as an enrolled participant of such a network (regardless of residence), for a maximum of five (5) years (the "Five-Year Period") after the date of death of the Participant. The Five-Year Period shall begin on the date of death of the Participant and end five years after the date of death of the Participant. A Covered Dependent's eligibility for the family protection plan benefit shall be determined on the date of death of the Participant. Notwithstanding eligibility for coverage under the Plan, the family protection plan benefit shall terminate upon the occurrence of any one or more of the following events: coverage under any Other Plan, remarriage, Medicare eligibility, or loss of dependent status by a Dependent Child as set forth in Section 3.30(d) (collectively referred to as a "Terminating Event"). If an eligible Covered Dependent loses the family protection plan benefit due to the occurrence of a Terminating Event, then that Covered Dependent may gain or regain eligibility within the original Five-Year Period provided that the condition(s) that caused the Terminating Event no longer exists. This benefit is subject to the following conditions:

(a) If a different family protection plan benefit has been established and published to Participants in a specific TeamCare area, it shall apply in lieu of the benefit set forth in this provision.

- (b) The level of benefits provided under family protection will follow the level of the active plan immediately preceding the family protection plan. The active plan benefits are outlined in Article XX, Schedule of Benefits.
- (c) The family protection plan benefit will be lost and cannot be reestablished if, within twenty-four (24) months prior to the Participant's death, charges payable by the Fund are incurred by the Participant or a Covered Dependent through non-emergency use of a Hospital or Physician outside the Participant's TeamCare network.
- (d) If eligibility for the family protection plan benefit has been granted, services from TeamCare providers for non-emergency care will be required for benefits to be payable.
- (e) family protection plan benefits do not include Life Insurance or Short-term Disability benefits.

3.33 COVERAGE UNDER AN HOURLY PLAN (on and after November 1, 2020)

- (a) Only groups associated with the Building Material and Construction craft, or as otherwise approved by the Trustees, would be eligible for participation in an Hourly Plan. Coverage under an Hourly Plan shall be based on Employer Contributions, deductions from the Employee's Hourly Bank and Self-Payments. Coverage for Employees in an Hourly Plan shall be calculated on a calendar monthly basis. For Employees covered under an Hourly Plan, the "Hourly Coverage Month" is the calendar month which is two months following the month for which an Employer is obligated to submit Employer Contributions. For example, if Employer Contributions are submitted on behalf of an Employee who worked during the month of November 2020, those contributions would be applied to provide Coverage for the month of January 2021.
- (b) Employees in a New Group in an Hourly Plan will be eligible for Immediate Coverage for the first two months following the New Group initial participation in the Fund, subject to verification and subject to the terms of Section 3.03 and 3.07 of the Active Plan (governing eligibility for Immediate Coverage), provided that Employer Contributions during the first and second full month in which the Employer was obligated to contribute on behalf of the New Group equal or exceed the minimum number of hours necessary to provide Coverage as determined by the bargaining parties.
- (c) An Employee who is not in a New Group will have Coverage under an Hourly Plan in an Hourly Coverage Month provided that the total of Employer Contributions required to be made on an Employee's behalf, any deductions from the Hourly Bank, and any Self-Payments for Continuation Coverage equal the requirement for monthly Coverage under that plan for the Hourly Coverage Month.
- (d) Employer Contributions submitted during any calendar month on or after November 1, 2020 (the "Hourly Contribution Month") shall be applied to provide Coverage during the related Hourly Coverage Month.
- (e) If Employer Contributions are not used to provide Coverage (either those amounts insufficient to provide Coverage or those amounts in excess of the amount necessary to provide Coverage), that amount will be added to the Employee's Hourly Bank.

- (f) If the Employer Contributions required to be made on an Employee's behalf in any Hourly Contribution Month are less than the minimum necessary to provide Coverage under the Plan during the related Hourly Coverage Month, an amount in dollars shall be deducted from the Hourly Bank on the first day of that Hourly Coverage Month, calculated at the then current contribution rate, which amount shall be sufficient such that the total of Employer Contributions required to be made on an Employee's behalf and the deduction from the Hourly Bank shall equal the amount of contributions necessary to provide Coverage under the Plan during the Hourly Coverage Month. A Covered Participant must use hours from the Hourly Bank to provide for continuous Coverage until the Hourly Bank is reduced to an amount less than the minimum necessary to provide Coverage during the Hourly Coverage Month or until the Employer no longer has an obligation to submit contributions on behalf of the group of Employees in the Hourly Plan.
- (g) If Employer Contributions required to be made on an Employee's behalf in any Hourly Contribution Month and any deductions from the Hourly Bank are insufficient to provide Coverage under the Plan in an Hourly Coverage Month, the Employee shall have a Qualifying Event within the meaning of Section 3.16 on the first day of the Hourly Coverage Month in which Employer Contributions and deductions from the Hourly Bank are insufficient to provide Coverage during that Hourly Coverage Month. In that event, a Covered Participant shall be permitted to elect Continuation Coverage and make Self-Payments for Continuation Coverage pursuant to Section 3.20.
- An increase or decrease in contribution rates will cause an increase or decrease in the hourly equivalent of the dollar value of the Hourly Bank which is calculated by dividing the dollar amount of the Hourly Bank by the currently related contribution rate. If an Employee has Employer Contributions submitted on his or her behalf under an Hourly Plan by two or more Employers during the same Hourly Contribution Month, the amount of the Hourly Bank will be computed using the rate of the Plan under which the higher number of hours of Employer Contributions were submitted on behalf of that Employee. All Employer Contributions required to be made on an Employee's behalf in an Hourly Contribution Month are applied first to provide Coverage for the related Hourly Coverage Month. If Employer Contributions required to be made on an Employee's behalf in any Hourly Contribution Month exceed the minimum amount necessary to provide Coverage under that Plan during the related Hourly Coverage Month, the value of the Employer Contributions required to be made which are in excess of that minimum amount will be added to the Hourly Bank. The value of any Employee's Hourly Bank may never exceed an amount sufficient to provide the maximum number of months of consecutive Coverage as set by the terms of the Plan related to the Employee.
- (i) Coverage under an Hourly Plan shall continue, and the Hourly Bank will not be reduced, if the Covered Participant is receiving Coverage during a Temporary Work Stoppage pursuant to Section 3.09 or if the Employee is receiving Short-Term Disability Coverage under a Plan which provides Short-Term Disability Coverage. The full number of hours necessary to provide Coverage under a Plan without Short-Term Disability Coverage will be deducted from the Hourly Bank to provide Coverage under their current Plan.
- (j) The Hourly Bank will not be reduced for a period of up to 12 months if the Employee establishes Coverage under any Plan, other than an Hourly Plan, which requires contributions at the same or a higher benefit rate. However, if the Employee continues to be covered under such Plan following the end of such 12-month grace period, the amount

in the Hourly Bank shall be reduced each month thereafter by the amount that would have been necessary to provide Coverage under the previous Hourly Plan, regardless of the fact that the Employee may have Coverage during that same month based on Employer Contributions. All amounts in an Employee's Hourly Bank shall be forfeited on the date of withdrawal if the Employee is a member of a group that voluntarily withdraws from an Hourly Plan.

- (k) If the Employee establishes Coverage under any Plan, other than an Hourly Plan, which requires weekly or monthly contributions at a lower benefit rate, the Hourly Bank will be reduced each month by the difference between the Employer Contributions required to be made on an Employee's behalf and the amount that would have been necessary to provide the prior benefit level for that Hourly Coverage Month and the Covered Participant and any Covered Dependent shall be entitled to Coverage at the higher benefit level until the Hourly Bank is reduced to zero, at which time Coverage shall be reduced to the level of Coverage based on the Employer Contributions.
- (I) If a group of Employees that had Coverage under a Plan which requires weekly or monthly contributions becomes Covered under an Hourly Plan, the group of Employees shall, subject to approval by the Board of Trustees, have Immediate Coverage during the two months which include the first month during which the Employer had an obligation to contribute on that group and the immediately following month.
- (m) If a Covered Participant has an amount in the Hourly Bank on his date of retirement, that amount may be used to continue Coverage under an Active Plan until the Hourly Bank has been exhausted. A Covered Participant who has retired and is using the amount in the Hourly Bank to continue Coverage under an Active Plan will forfeit any remaining amount in the Hourly Bank if the Employer for whom the Covered Individual was working on the date of retirement is no longer obligated to contribute to the Fund on behalf of that Employee group due to a voluntary withdrawal.

If a Covered Participant has an amount in the Hourly Bank on his date of death, the Hourly Bank shall be reduced to zero effective on the Covered Participant's date of death provided that any dependents of the employee qualify for and elect continuing coverage through Family Protection Plan.

(n) The Hourly Bank may not be voluntarily saved for use at a future date. In no event shall the amount in a Covered Participant's Hourly Bank be payable to the Employee or his Dependents.

Transitional provisions related to November and December 2020

(o) If an Employee has an amount in the Employee's Hourly Bank on November 1, 2020, at least equal to the amount of contributions necessary to provide Coverage under the Plan, the Employee shall have Coverage for the month of November 2020 and an amount in dollars shall be deducted from the Hourly Bank on the first day of November 2020 calculated at the then current contribution rate, which amount shall be sufficient such that the deduction from the Hourly Bank shall equal the amount necessary to provide Coverage under that Plan.

If an Employee does not have a sufficient amount in the Employee's Hourly Bank on November 1, 2020, at least equal to the amount of contributions necessary to provide

Coverage under the Plan but who had Coverage in September 2020 based entirely on Employer Contributions for work performed during that month shall have Coverage for the month of November 2020 without the need to make any Self-Payments. An amount up to and including the amount equal to the requirement for monthly Coverage under that Plan shall be deducted from the Hourly Bank.

If an Employee has an amount in the Employee's Hourly Bank on December 1, 2020, at least equal to the amount of contributions necessary to provide Coverage under the Plan, the Employee shall have Coverage for the month of December 2020 and an amount in dollars shall be deducted from the Hourly Bank on the first day of December 2020 calculated at the then current contribution rate, which amount shall be sufficient such that the deduction from the Hourly Bank shall equal the amount necessary to provide Coverage under that Plan.

If an Employee does not have a sufficient amount in the Employee's Hourly Bank on December 1, 2020, at least equal to the amount of contributions necessary to provide Coverage under the Plan but who had Coverage in October 2020 based entirely on Employer Contributions for work performed during that month shall have Coverage for the month of December 2020 without the need to make any Self-Payments. An amount up to and including the amount equal to the requirement for monthly Coverage under that Plan shall be deducted from the Hourly Bank.

(p) On or after November 1, 2020 if an Employee is a New Participant or an Employee that did not have Coverage in an Hourly Plan at any time between November 1, 2019 and November 1, 2020, then the Employee shall only be permitted to elect COBRA Continuation Coverage and make Self-Payments for COBRA Continuation Coverage pursuant to paragraph 3.20 of this Section.

3.34 RESERVED FOR FUTURE USE

3.35 QUALIFYING EVENT - HOURLY PLAN (on and after November 1, 2020)

- (a) A Covered Participant under an Hourly Plan shall have a Qualifying Event on the first day of an Hourly Coverage Month if a Covered Participant loses Coverage during that month because the Employer was not obligated to submit Employer Contributions in an amount equal to or greater than the amount necessary to provide Coverage under the plan during the related Hourly Contribution Month and the Covered Participant's Hourly Bank is insufficient to provide Coverage during that Hourly Coverage Month. In that event, a Covered Participant shall be permitted to elect Continuation Coverage and make Self-Payments for Continuation Coverage pursuant to Section 3.20.
- (b) If an Employee elects to submit Self-Payments pursuant to a valid election of Continuation Coverage, the cost of the first month of Continuation Coverage shall be reduced by the dollar value of any hours remaining in the Hourly Bank. Once an Employee elects Continuation Coverage, any subsequent Employer Contributions submitted on behalf of an Employee during an Hourly Contribution Month while that Employee is receiving Continuation Coverage pursuant to a Qualifying Event which occurred while that Employee was a Covered Participant in an Hourly Plan that are insufficient to provide Coverage shall be directly applied to reduce the amount that the Employee is required to pay for Continuation Coverage during that Hourly Contribution Month.
- (c) Such Continuation Coverage will terminate if the Employee thereafter becomes eligible for Coverage under an Hourly Plan and the Employer Contributions required to be made on an Employee's behalf during any Hourly Contribution Month are sufficient to provide Coverage during at least one Hourly Coverage Month. Continuation Coverage under this Section shall otherwise terminate pursuant to the terms of the Plan.

3.36 COMMENCEMENT OF COVERAGE UNDER A MULTI-TIER PLAN

- (a) The status of an Employee as a Covered Participant and the status of a Dependent of an Employee as a Covered Dependent under a Multi-tier Plan shall be governed by the terms of this Plan Document, any applicable collective bargaining agreement, participation agreement and the election of the Covered Participant.
- (b) Employees in a New Group shall elect Coverage within 60 days of participation of the New Group in the Plan. Such election shall be made in writing unless the Plan offers the Employees of a New Group the option to elect to continue any written election made under the prior plan and that election is timely disclosed to the Plan. New Employees shall elect Coverage in writing within 60 days of employment. The Employee may select Coverage for Dependents from among those options available to the group as provided in the collective bargaining agreement, however the tier of Coverage selected may not require a contribution rate higher than the rate necessary to provide Coverage for the Covered Individual and all of his or her Dependents.
- (c) The Employee is responsible for providing the Plan with a properly completed election of Coverage form for Covered Dependents as designated by the Plan. If the Employee fails to provide such an election of Coverage form within 60 days as set forth in this section, the Employee will be deemed to have elected Member Only Coverage.
- (d) Any Covered Participant that loses Coverage under a Multi-tier Plan due to a loss of employment or reduction in hours, and who returns to Covered Employment with the

- same Employer will be automatically be placed in the same Coverage tier as previously elected unless the Covered Individual submits an election of Coverage form within 60 days of Coverage as set forth in this section.
- (e) Once an Employee has elected or has been deemed to have elected a Coverage tier, that Coverage tier may only be changed if the Employee has a Qualifying Event as defined in Section 3.16, as provided in Section 3.39, during an Open Enrollment period, or during a Special Enrollment Period.

3.37 OPEN ENROLLMENT UNDER A MULTI-TIER PLAN

(a) The Open Enrollment Period for each Multi-tier Plan shall be established by the Plan each year. During an Open Enrollment Period, an Employee may voluntarily change his or her Coverage tier by adding or dropping Covered Dependents. The Open Enrollment Period is the only time during which an Employee may voluntarily change Coverage tier (unless the change is made in accordance with subsection (b) of this section). The Coverage options available to an Employee are governed by the applicable tier schedule as indicated in the Collective Bargaining Agreement applicable to the Employee. If the first day of the next calendar year is a Sunday, any election of a Coverage tier during an Open Enrollment period shall be effective on the first day of the next year and, in all other cases, any election of a Coverage tier shall be effective on the Sunday immediately preceding of the first day of the next year. If the Plan determines that one or more of a Participant's Covered Dependents are no longer eligible for Coverage, the Plan reserves the right to immediately reduce the tier of Coverage, provided that such reduced tier of Coverage shall include the Covered Participant and all other Dependents who were Covered Dependents at the time of the reduction.

3.38 SPECIAL ENROLLMENT

- (a) A Covered Individual shall be entitled to a Special Enrollment period if: (1) the Covered Participant or Covered Dependent loses eligibility for coverage under an Other Plan that is not COBRA continuation coverage (but shall not include a loss of coverage due to the failure of the Employee or Dependent to pay premiums on a timely basis and shall not include termination of coverage for cause); or (2) employer contributions under an Other Plan for Coverage that is not COBRA Continuation Coverage are terminated (including termination of contributions by any current or former Employer that was contributing to Coverage for the Employee or Dependent); or (3) the Covered Individual has exhausted all COBRA Continuation Coverage under an Other Plan. Any individual that qualifies for a Special Enrollment period under this paragraph must request enrollment within 60 days of the event that caused that individual to qualify for Special Enrollment. Coverage will become effective on the first day of the Coverage Month following the date on which a completed request for Special Enrollment is received by the Fund.
- (b) If an Employee marries, the Employee, Spouse and any individual who becomes a Dependent by the marriage shall each be entitled to enroll during a Special Enrollment period provided that they request enrollment within 60 days of the marriage. In the case of a Special Enrollment period caused by marriage, Coverage will become effective on the first day of the Coverage Month following the date on which a completed request for Special Enrollment is received by the Fund.

(c) If an individual becomes a Dependent of an Employee due to birth, adoption, or placement for adoption, the Employee, Spouse and Dependents shall each be entitled to a Special Enrollment period provided they request enrollment within 60 days of the birth, adoption, or placement for adoption. In the case of birth, adoption, or placement for adoption, Coverage will become effective as of the date of the birth, adoption, or placement for adoption.

3.39 **DECLINING COVERAGE**

- (a) An Employee who is a member of a New Group on the first date of Coverage for the New Group will be permitted to decline all Coverage, including Member Only Coverage under a Multi-tier Plan, provided that the Employee can establish that he or she had declined Coverage under the plan offered by the Employer immediately prior to acceptance of the New Group. Coverage is declined by completing a Waiver of Coverage form approved by the Fund and by providing adequate proof of other insurance within 30 days of the date on which the Employer was first obligated to contribute on behalf of the New Group. This is the only method to decline Coverage for an Employee. New Employees in an existing group may not decline Coverage and must, at a minimum, elect Member Only Coverage. Once an Employee who has declined Coverage elects any form of Coverage under a Plan, that Employee may not again decline Coverage.
- (b) An Employee who has declined Coverage under the terms of this paragraph may elect Coverage if the Employee has a Qualifying Event as defined in Section 3.16, may elect Coverage during any applicable Special Enrollment Period, and may elect Coverage during any Open Enrollment Period if the Covered Participant is covered by a Multi-tier Plan.
- (c) A Covered Participant shall be permitted to drop the Coverage of one or more Covered Dependents at any time, for good cause, provided that the Covered Participant: (1) provides the Plan with a written and notarized request to drop the Coverage of Covered Dependents; (2) provides adequate proof of other insurance for the Covered Dependents; and (3) provides written and notarized proof of Consent to Waiver of Coverage from the Covered Dependents in a format approved by the Plan. If the Covered Dependents are adults, the Consent shall be signed by the Covered Dependents. If the Covered Dependents are minors and the Participant is neither the sole surviving natural or adoptive parent nor a sole Guardian, the Consent shall be signed by the other natural or adoptive parent or all co-guardians. If the other natural or adoptive parent is deceased, the Covered Participant shall provide a death certificate for the other natural or adoptive parent. Coverage shall terminate as soon as the request is processed by the Plan in the ordinary course of business.

ARTICLE IV. GENERAL EXCLUSIONS, LIMITATIONS AND CONDITIONS FOR PAYMENT OF CLAIMS

4.01 PAYMENT ONLY FOR COVERED CLAIMS OF COVERED INDIVIDUALS

A Covered Individual shall not be entitled to any payment on a claim for benefits unless the benefits are provided by the Plan, the claimant is a Covered Individual and the claim for benefits is submitted in proper form as determined by the Fund.

4.02 EXCLUSION OF PAYMENT FOR TREATMENT NOT CONSIDERED STANDARD MEDICAL CARE OR MEDICALLY NECESSARY

A Covered Individual shall not be entitled to payment of any charges for care, treatment, services or supplies which are not medically necessary or are not generally accepted by the medical community as Standard Medical Care, Treatment, Services or Supplies.

4.03 LIMITATION ON PAYMENT OF CLAIMS ARISING FROM WORK-RELATED INJURY OR COVERED BY WORKERS' COMPENSATION

A Covered Individual shall not be entitled to payment on a claim for any charge incurred for any treatment or service for any illness or injury which is sustained as a result of any enterprise or occupation for wage or profit or is an illness or injury of the type covered by any applicable Workers' Compensation Act or similar law providing benefits to employees for on-the-job injuries.

In the event that a Covered Individual's claim for Workers' Compensation benefits is denied by the Workers' Compensation Carrier, the Covered Individual may be eligible to receive some benefits if the Covered Individual and his/her attorney enter into an agreement with the Fund to provide benefits during the appeal of the denial as set forth in Article 11.15.

After a five (5) year period from the date of Disability, any complication arising from the illness or injury shall be deemed payable in accordance with the Plan provisions, unless it is still compensable under Workers' Compensation.

4.04 EXCLUSION OF PAYMENT FOR TREATMENT OF INJURIES SUSTAINED WHILE IN ANY UNIFORMED SERVICE

A Covered Individual shall not be entitled to payment for any charge incurred for treatment or service due to illness or injury sustained while in any Uniformed Service, except short-term service provided for in Section 3.20, or for treatment of any complication of such illness or injury. After a five (5) year period from the date of Disability, any complication arising from the illness or injury shall be deemed payable in accordance with the Plan provisions.

4.05 **EXCLUSION OF PAYMENT FOR TREATMENT DUE TO ILLNESS OR INJURY ARISING OUT OF ANY ACT OF WAR OR CIVIL DISTURBANCE**

A Covered Individual shall not be entitled to payment for any charge incurred for treatment or services due to illness or injury arising out of declared or undeclared war or any act of war or civil disturbance, including riots, demonstrations and marches or for treatment of any complication of such illness or injury. After a five (5) year period from the date of Disability, any complication arising from the illness or injury shall be deemed payable in accordance with the Plan provisions, unless it is compensable by an Other Plan or government agency.

4.06 EXCLUSION OF PAYMENT FOR TREATMENT OF INJURIES ARISING AS A RESULT OF PARTICIPATION IN CRIMINAL CONDUCT

- (a) A Covered Individual shall not be entitled to payment for any charge incurred for treatment or services due to injury, and any complication thereof, sustained as a result of participation in conduct which results in a conviction for violating any federal, state or local law. The Fund shall have the right to recover the amount of any payment upon discovery that the injury, and any complication thereof, for which payment was made resulted from participation in conduct which results in a conviction for violating any federal or state criminal law.
- (b) Notwithstanding the provisions of subsection 4.06(a), a Covered Individual shall not be entitled to payment for any charge incurred for treatment or services due to injury, and any complication thereof, sustained as a result of participation in an illegal act if the Covered Individual dies within thirty (30) days of the date of the illegal act regardless of whether such illegal act results in a conviction for violating any federal, state or local law. The Fund shall have the right to recover the amount of any payment upon discovery that the injury, and any complication thereof, for which payment was made resulted from participation in an illegal act if the Covered Individual dies within thirty (30) days of the date of the illegal act regardless of whether such participation in an illegal act results in a conviction for violating any federal, state or local law.

4.07 LIMITATION ON PAYMENT FOR TREATMENT RECEIVED OUTSIDE THE UNITED STATES

A Covered Individual shall not be entitled to payment for treatment outside the United States if it is not care, treatment, services or supplies that is medically necessary and is generally accepted by the medical community in the United States as Standard Medical Care, Treatment, Services or Supplies. All exclusions and limitations of the Plan shall be fully applicable to all such care, treatment, services and supplies to the same extent as if it were provided within the United States. Interpretations relative to these exclusions and limitations will be resolved by the Fund at its discretion, assisted by the Fund's medical consultants. Benefits will be paid in United States currency.

4.08 EXCLUSION FOR PAYMENT FOR TREATMENT CONNECTED WITH SURGERY FOR COSMETIC PURPOSES

A Covered Individual shall not be entitled to payment on a claim for benefits for any charge incurred for treatment or service connected with a cosmetic procedure, even if performed for psychological reasons,

unless the treatment or service is medically required as a result of an illness, congenital anomaly or an Accidental Bodily Injury.

This exclusion includes, but is not limited to:

- (a) Any surgery primarily for obesity, including lipectomy, suction lipectomy, abdominoplasty and any other surgical procedure, a purpose and result of which is primarily to remove adipose tissue (except to the extent permitted by Section 12.04 Surgical and Obstetrical Expense Benefit);
- (b) Augmentation mammoplasty, unless part of reconstructive surgery for the treatment of malignancy of the breast necessitating removal of a portion or all of the breast tissue;
- (c) Rhinoplasty, unless the patient has sustained a traumatic fracture of the nasal septum, or unless the patient has chronic nasal obstruction and the procedure is undertaken to relieve this obstruction:
- (d) Otoplasty for irregular deformity or macrotia. This is sometimes referred to as plastic surgery for lop ears or cauliflower ears;
- (e) Blepharoplasty, or repair of drooping eyelids, unless the droop of the eyelids is such as to restrict the field of vision and the visual field restriction is documented by the ophthalmological consultant;
- (f) Radical Keratectomy or Keratotomy, unless the patient has myopia of such a severe degree that it cannot be corrected by lenses;
- (g) Rhytidectomy (face lift);
- (h) Dyschromia (tattoo removal); and
- (i) Genioplasty (chin augmentation).

4.09 EXCLUSION OF PAYMENT FOR TREATMENT OTHERWISE COVERED UNDER THE SOCIAL SECURITY ACT

A Covered Individual shall not be entitled to payment for any charge incurred for treatment or service to the extent that such charge is covered or provided by the Social Security Act, as amended, except as provided in Article V.

4.10 EXCLUSION OF PAYMENT FOR TREATMENT NOT RELATED TO ILLNESS, INJURY, CONGENITAL ANOMALY OR PREGNANCY

Except as otherwise provided herein, a Covered Individual shall not be entitled to payment of a claim unless the Covered Individual is ill, injured, has a congenital anomaly, is pregnant or is an organ transplant donor, and receives treatment, compensable under this Plan, related to the illness, injury, congenital anomaly, pregnancy or organ donation.

4.11 PAYMENT FOR CERTAIN DENTAL TREATMENTS

A Covered Individual shall only be entitled to payment of Basic Benefits, as outlined in Article XII, and Major Medical Expense Benefits, as outlined in Article XIII, for charges incurred in connection with dental operations, if prescribed by a Physician as a result of a medical condition and as approved by the Fund, or for Hospital Confinements where the patient or condition is any one of the following:

- (a) Dependent Child five (5) years or younger;
- (b) Physically disabled Child or adult;
- (c) Intellectually disabled Child or adult (includes those with Cerebral Palsy);
- (d) Hemophilia;

The Plan shall not pay Basic Benefits or Major Medical Expense Benefits for dental materials or dental procedures, except as outlined in Section 15.06.

4.12 EXCLUSION OF PAYMENT FOR TREATMENT OF NON-COMPENSABLE PROCEDURES

A Covered Individual shall not be entitled to payment for any charge incurred for treatment of complications arising from the performance of any procedure not compensable under this Plan.

4.13 **RESERVED FOR FUTURE USE**

4.14 EXCLUSION OF PAYMENT FOR CERTAIN ITEMS

A Covered Individual shall not be entitled to payment for any charge incurred for sales taxes, surcharges, interest, late charges, completion of any claim form or missed appointments.

4.15 EXCLUSION OF PAYMENT FOR MAINTENANCE CARE

A Covered Individual shall not be entitled to payment for any charge incurred for Maintenance Care, as defined in Section 1.47.

4.16 **EXCLUSION OF PAYMENT OVER PRESCRIBED MAXIMUMS**

A Covered Individual shall not be entitled to payment for any charge which would exceed the stated maximum benefit, maximum allowable fee or stated percentage of covered charges payable as set forth in Article XX of the Plan unless "balance under Major Medical" immediately follows such reference.

4.17 LIMITATION ON ELIGIBILITY FOR COVERAGE OF ORGAN OR TISSUE TRANSPLANTS

Benefits for organ or tissue transplants, including all related services, are payable only if approved on a pre-admission basis. The criteria for patient selection is in agreement with generally accepted medical standards.

4.18 EXCLUSION OF PAYMENT FOR INFERTILITY TREATMENT AND SERVING AS A SURROGATE MOTHER

A Covered Individual shall not be entitled to payment for any charge incurred in connection with the treatment of infertility. This limitation includes but is not limited to:

- (a) Charges incurred in connection with in vitro fertilization;
- (b) Charges incurred in connection with artificial insemination;
- (c) Charges for Prescription Drugs designed to enhance the ability to conceive, used in connection with (a) and (b) above, including but not limited to Clomid, Milophene, Metrodin, Lutrepulse, Pergonal and human chorionic gonadotropin in any form; and
- (d) Charges incurred in connection with reversal of prior sterilization procedures.

A Covered Individual shall not be entitled to payment for any charge incurred in connection with serving as a Surrogate Mother (even if the Covered Individual has provided the ovum).

4.19 EXCLUSION OF PAYMENT FOR CHARGES FOR WHICH THE COVERED INDIVIDUAL IS NOT RESPONSIBLE TO PAY

A Covered Individual shall not be entitled to payment for any charge incurred for any Care, Treatment, Services or Supplies if the medical service provider has waived any co-payments due from the Covered Individual or has otherwise relieved the Covered Individual from an obligation to pay for the Care, Treatment, Services or Supplies or agrees to accept as full payment whatever amount is payable under this Plan.

In no event will the Fund pay for any Care, Treatment, Services or Supplies which have been represented by the person supplying those examinations, services or supplies to be free to the Covered Individual.

4.20 LIMITATION ON PAYMENT OF CLAIMS FOR SERVICES BY PROVIDERS NOT IN PREFERRED PROVIDER ORGANIZATION NETWORK

- (a) Benefits otherwise payable shall be reduced by ten percent (10%) if there exists a TeamCare network covering the Covered Participant; and
- (b) The charges were incurred by the Covered Participant or Covered Dependent through non-emergency use of Hospitals, Physicians or ancillary providers outside of the TeamCare network.

ARTICLE V. COORDINATION OF BENEFITS

5.01 PRIORITY OF COVERAGE WHERE COVERED INDIVIDUAL IS COVERED BY AN OTHER PLAN

An Other Plan providing no fault, personal injury protection or financial responsibility motor vehicle insurance coverage or benefits shall always have primary responsibility. If the benefits of this Plan duplicate or overlap with benefits for hospital, surgical, dental, psychiatric, chiropractic or other medical treatment provided by an Other Plan, such duplication or overlapping shall be avoided. In this regard, primary responsibility for providing benefits shall be determined in the following order:

- (a) A plan that does not contain a coordination of benefits provision is always primary.
- (b) A plan that covers the person other than as a dependent is the primary plan and the plan that covers the person as a dependent is the secondary plan. However, if the person is a Medicare beneficiary and, as a result of federal law, Medicare is secondary to the plan covering the person as a dependent; and primary to the plan covering the person as other than a dependent then the order of benefits between the two plans is reversed so that the plan covering the person other than as a dependent is the secondary plan and the other plan is the primary plan.
- (c) Unless there is a court decree stating otherwise, when a dependent child is covered by more than one plan the order of benefits is determined as follows:
 - (1) For a dependent child whose parents are married or are living together, whether or not they have ever been married:
 - (i) The Plan of the parent whose birthday falls earlier in the calendar year is the primary Plan; or
 - (ii) If both parents have the same birthday, the Plan that has covered the parent the longest is the primary Plan.
 - (2) For a dependent child whose parents are divorced or separated or not living together, whether or not they have ever been married:
 - (i) If a court decree states that one of the parents is responsible for the dependent child's health care expenses or health care coverage and the Plan of that parent has actual knowledge of those terms, that Plan is primary. This rule applies to Plan years commencing after the Plan is given notice of the court decree; or
 - (ii) If a court decree states that both parents are responsible for the dependent child's health care expenses or health care coverage, the provisions of Subparagraph (c)(1) above shall determine the order of benefits; or
 - (iii) If a court decree states that the parents have joint custody without specifying that one parent has responsibility for the health care expenses or health care coverage of the dependent child, the provisions of Subparagraph (c)(1) above shall determine the order of benefits; or

- (iv) If there is no court decree allocating responsibility for the dependent child's health care expenses or health care coverage, the order of benefits for the child are as follows:
 - The Plan covering the Custodial parent;
 - The Plan covering the spouse of the Custodial parent;
 - The Plan covering the non-custodial parent; and then
 - The Plan covering the spouse of the non-custodial parent.

In the absence of an applicable court decree, the Custodial parent is the parent with whom the dependent child resides more than one half of the calendar year excluding any temporary visitation.

- (v) If the dependent child does not reside with either parent, the provisions of Subparagraph (c)(1) above shall determine the order of benefits.
- (3) For a dependent child covered under more than one Plan of individuals who are not the parents of the child, the provisions of Subparagraph (c)(1) or (c)(2) above shall determine the order of benefits as if those individuals were the parents of the child.
- (d) The Plan that covers a person as an active employee is the primary Plan. The Plan covering that same person as a retired or laid-off employee is the secondary Plan. The same would hold true if a person is a dependent of an active employee and that same person is a dependent of a retired or laid-off employee. If the other Plan does not have this rule, and as a result, the Plans do not agree on the order of benefits, this rule is ignored. This rule does not apply if the rule labeled (b) can determine the order of benefits.
- (e) If a person whose coverage is provided pursuant to COBRA or under a right of continuation provided by state or other federal law is covered under another Plan, the Plan covering the person as an employee, member, subscriber or retiree or covering the person as a dependent of an employee, member, subscriber or retiree is the primary Plan and the COBRA or state or other federal continuation coverage is the secondary Plan. If the other Plan does not have this rule, and as a result, the Plans do not agree on the order of benefits, this rule is ignored. This rule does not apply if the rule labeled (b) can determine the order of benefits.
- (f) The Plan that covered the person as an employee, member, policyholder, subscriber or retiree longer is the primary Plan and the Plan that covered the person the shorter period of time is the secondary Plan.
- (g) If there is coverage provided by a governmental program, that coverage shall have primary responsibility unless prohibited by federal law.
- (h) If both the person and spouse are Covered Participants of this Plan or another Central States Plan, this Plan will be considered an Other Plan.
- (i) If the preceding rules do not determine the order of benefits, the Allowable expenses shall be shared equally between the Plans meeting the definition of Plan.
- 5.02 **EFFECT OF PRIORITY RULES ON AMOUNT OF PAYMENTS UNDER THE PLAN**

Whenever this Plan is determined to have primary responsibility, the Covered Individual shall receive benefits without regard to coverage under the Other Plan. Whenever this Plan is determined not to have primary responsibility, this Plan shall pay, after the Other Plan has paid its maximum allowable benefits, any remaining covered charges up to the amount this Plan would have paid if this Plan had primary responsibility and without the payments from the Other Plan being taken into account in applying the specific benefit maximums indicated in this Plan. As used in this section, the term "Covered Charges" means charges by a provider that a Covered Individual is responsible to pay the provider and that entitle a Covered Individual to benefits under the Plan. Covered Charges shall not include payment for charges for which the Covered Individual is not responsible to pay as enumerated in Section 4.19, regardless of whether this Plan is the primary or secondary provider. Maximum allowable benefits will always be construed under this section to mean the amount payable by the Other Plan without regard to coverage under this Plan; that is, the Other Plan's maximum allowable benefits will be computed as if there were no coverage under this Plan.

5.03 RECOVERY OF PAYMENTS WHEN AN OTHER PLAN IS INVOLVED

Whenever this Plan has made benefit payments which exceed the amount of benefits payable under the terms of this Plan or which an Other Plan was required to make under Section 5.01, the Fund shall have the right to recover the amount of such payments from any persons receiving such payments or from any Other Plans having primary responsibility for the payment of benefits. The Trustees are authorized to file suit on behalf of the Fund to recover any such payments or to seek a judicial declaration that an Other Plan has primary responsibility for the payment of benefits.

Further, where the Plan has made benefit payments which exceed the amount of benefits payable under the terms of this Plan, and those payments were required to be made by an Other Plan under Section 5.01 (so that the Plan's payment of the Other Plan's obligation has unjustly enriched the Other Plan), the Other Plan shall be liable to make equitable restitution to the Plan, and/or for other appropriate equitable relief to the Plan, in order to enforce the terms of the Plan Document in the amount of benefit payments that were paid by the Plan, but should have been paid by the Other Plan.

5.04 PAYMENTS TO OTHER PLANS

Whenever payments should have been made by this Plan pursuant to the provisions of Article V, but are made by an Other Plan, the Fund shall have the right to pay over to such Other Plan any amounts this Plan should have paid under the provisions of Article V.

5.05 NO FAULT, PERSONAL INJURY PROTECTION OR FINANCIAL RESPONSIBILITY MOTOR VEHICLE INSURANCE COVERAGE

Benefits under no fault, personal injury protection or financial responsibility motor vehicle insurance coverage, as described in Section 1.53(f), shall be primary to benefits under this Plan notwithstanding state or local law or regulation to the contrary. In the event an Other Plan, as described in Section 1.53(f), fails or refuses to assume primary responsibility for the payment of benefits, the Fund may provide the Covered Individual with benefits under this Plan. Such benefits shall be provided under a reservation of rights and without prejudice to the Fund's right to recover the amount of such benefits from the Other Plan.

A Covered Individual shall cooperate with this Plan in any attempt to recover the amount of benefits paid under a reservation of rights pursuant to this Section. Upon request by the Fund, the Covered Individual shall execute an assignment of rights and any other documents the Fund deems necessary to effect a

recovery of the amount of benefits paid under such reservation of rights. A Covered Individual shall do nothing to prejudice the Fund's rights under Article V and is not authorized to subordinate no fault, personal injury protection, financial responsibility or medical reimbursement benefits to benefits under this Plan.

5.06 COORDINATION OF BENEFITS WITH ANOTHER PLAN

When a Covered Individual has primary coverage through an Other Plan, this Plan will coordinate benefits as described in Section 5.01 provided the Covered Individual is utilizing the Other Plan's network of Hospitals, Physicians and ancillary providers. If the Covered Individual is denied benefits by the Other Plan for using out of network providers or not obtaining proper referrals or authorization, this Plan will deny benefits.

ARTICLE VI. CONTRIBUTIONS AND FUNDING OF BENEFITS

6.01 EMPLOYER'S OBLIGATION TO CONTRIBUTE TO THE PLAN

Each Employer shall make continuing and prompt Employer Contributions to the Fund, as required by either a Collective Bargaining Agreement or an applicable law (including but not limited to the Uniformed Services Employment and Reemployment Rights Act of 1994), at rates established by the terms of a Collective Bargaining Agreement. Any Employer which, based upon the Uniformed Services Employment and Reemployment Rights Act of 1994, is required to make Employer Contributions to the Fund, shall make those Employer Contributions at the rates and in the amounts of Employer Contributions which that Employer would have been obligated to pay to the Fund, relative to the Covered Participant, if his employment by that Employer had continued throughout (and had not been interrupted by) such Service in the Uniformed Services (plus interest, to the extent such Employer Contributions are not paid at the time of such absence from employment as a result of Service in the Uniformed Services, in accordance with the trust agreement of the Fund). Each Employer shall make Employer Contributions to the Fund on behalf of each Employee whose classification of work is covered by its Collective Bargaining Agreement. All Employer Contributions must be made according to rules established by the Board of Trustees. The obligation to make Employer Contributions shall continue during periods when a Collective Bargaining Agreement is being negotiated, but such Employer Contributions shall not be required in case of strike or after termination of the Collective Bargaining Agreement, unless the parties mutually agree otherwise or unless required by an applicable law.

6.02 CREDITS FOR ERRONEOUS EMPLOYER CONTRIBUTIONS

The Fund shall credit the Employer's account for contributions that have been billed to an Employer only if permitted by Article XI, Section 1 of the Fund's Trust Agreement.

6.03 IRREVOCABLE NATURE OF CONTRIBUTIONS

Except as provided in Section 6.02, any and all contributions made by or on behalf of an individual shall be irrevocable and shall be transferred to the Trustees and held as provided in this Plan and in the Trust Agreement, to be used in accordance with the provisions of this Plan in providing benefits and paying the expenses of the Fund.

6.04 **SELF-PAYMENTS**

All Participants making Self-Payments shall comply with Fund procedures, including the submission of prescribed forms to confirm the Participant's status.

ARTICLE VII. AMENDMENTS AND PLAN TERMINATION

7.01 PROCEDURE FOR AMENDING THE PLAN

This Plan may be amended, from time to time, by majority vote of the Trustees. A copy of each amendment of this Plan shall be adopted and filed by the Trustees as part of the records and minutes of the Trustees.

7.02 TERMINATION OF THE PLAN

This Plan shall be maintained and operated in full force and effect until the occurrence of any of the following events, in which case the Plan shall be terminated:

- (a) The Trust Fund, in the opinion of the Trustees, shall be inadequate to effectuate the intent and purposes of the Trust Agreement.
- (b) The Trust Fund, in the opinion of the Trustees, shall be inadequate to meet payments due, or to become due, to persons already drawing benefits.
- (c) There are no individuals living who can qualify as Employees, as defined in Article I of this Plan.
- (d) All contracts between the Fund and Employers expire, terminate or are canceled.

In the event of termination, the Trust Fund shall be distributed by the Trustees in accordance with any plan which conforms to the intent and purposes of the Trust Agreement and the "Employee Retirement Income Security Act of 1974".

ARTICLE VIII. PLAN ADMINISTRATION

8.01 TRUSTEE STATUS AS "NAMED FIDUCIARY"

Each Trustee by reason of his position is a "named fiduciary" of this Plan within the meaning of the "Employee Retirement Income Security Act of 1974".

8.02 **POWERS OF THE TRUSTEES**

The Trustees shall have authority to jointly control and manage the operation and administration of the Fund and of this Plan, in accordance with the terms of the Trust Agreement and of this Plan, including the authority to allocate fiduciary responsibilities among the Trustees and the authority to designate persons other than Trustees to carry out fiduciary responsibilities in the administration of the Fund and this Plan, except that the Trustees may allocate only a committee of Trustees or one (1) or more "investment managers" (as defined by the "Employee Retirement Income Security Act of 1974") to administer investment and other responsibilities relating to assets of the Fund. All questions or controversies, of whatsoever character, arising in any manner or between any parties or persons in connection with any claim for any benefits preferred by any Participant, beneficiary, or any other person, or whether as to the construction of the language or meaning of the rules and regulations contained in this Plan, shall be submitted to the Trustees, or to a Committee of the Trustees, and the decision of the Trustees or of such Committee thereof shall be binding upon all persons dealing with the Fund or claiming any benefit under the terms of this Plan.

8.03 **DECISIONS OF TRUSTEES**

All decisions by the Trustees, including all rules and regulations adopted by the Trustees, all amendments of the Trust Agreement and this Plan by the Trustees and all interpretations by the Trustees of any of said documents shall be binding upon all parties to the Trust Agreement, the Local Union, each Employer, all individuals claiming benefits pursuant to this Plan or any amendment thereof and all other individuals engaging in any transaction with the Fund. The Trustees are vested with discretionary and final authority in making all such decisions, including Trustee decisions upon claims for benefits by Covered Participants, Covered Dependents and other claimants, and including Trustee decisions interpreting plan documents of the Fund.

8.04 EFFECT OF ANY MISREPRESENTATION WITH RESPECT TO CLAIMS

Any misrepresentation in any claim or document submitted by a claimant to the Fund shall constitute grounds for rejection of the claim, for the denial of any requested benefits, and for the recovery by the Fund of all benefit payments made in reliance upon said misrepresentation.

8.05 PAYMENTS TO PERSONS WHO HAVE FAILED TO INFORM THE TRUSTEES OF A CHANGE OF ADDRESS

If any person who is entitled to receive payment of benefits, in accordance with this Plan, fails to inform the Trustees in writing and by registered mail of a change of address, and if the Trustees are unable to communicate with such person at the address last recorded by the Fund, and if a letter sent by registered mail from the Fund to such person is returned because it is not deliverable, all payments due such person shall be held without interest until a claim has been received and approved by the Trustees.

8.06 INFORMATION CONCERNING COVERED INDIVIDUALS

For purposes of implementing the terms of the Plan, the Fund may, without notice to or consent of any Covered Individual, obtain from any person or entity such information concerning the Covered Individual as the Fund deems necessary. Any person claiming benefits under the Plan shall furnish the Fund with such information as the Fund deems necessary to implement the Plan. When any claim for benefits under the terms of this Plan is submitted by a Covered Individual or any medical service provider that provided Care, Treatment Services or Supplies to a Covered Individual then, the furnishing of such claim shall act as a release by the Covered Individual to any medical service provider to allow the Fund, without further notice or consent of any Covered Individual, to obtain any medical records of the Covered Individual from any medical service provider whose claims for treatment of the Covered Individual are submitted for payment.

Failure on the part of any Covered Individual or medical service provider that provided Care, Services, Treatment or Supplies to any Covered Individual to supply or furnish any information requested by the Fund or its agent may result in the rejection of a claim for benefits and/or the recoupment of previously paid benefits.

8.07 USE AND DISCLOSURE OF PROTECTED HEALTH INFORMATION

The Plan may make any use and disclosure of Protected Health Information to the extent of and in accordance with the uses and disclosures permitted or required by the Health Insurance Portability and Accountability Act of 1996 (HIPAA) as amended.

8.08 SAFEGUARDS FOR PROTECTED HEALTH INFORMATION

The Trustees will implement administrative, physical and technological safeguards to reasonably and appropriately protect the confidentiality, integrity and availability of the electronic Protected Health Information that they create, receive, maintain or transmit on behalf of the Plan.

The Trustees will ensure that the "adequate separation" required by the Privacy Rule and by Section 8.07 of the Plan is supported by reasonable and appropriate security measures.

The Trustees will ensure that any agent, including a subcontractor, to whom they provide Protected Health Information agrees to implement reasonable and appropriate security measures to protect the Protected Health Information.

The Trustees will report to the Plan any security incident (within the meaning of 45 C.F.R. § 164.304) of which it becomes aware.

ARTICLE IX. BENEFIT CLAIMS

9.01 CLAIMS TO BE SUBMITTED IN WRITING ON AUTHORIZED FORMS

Claims for benefits other than the Retiree Contribution Benefit shall be submitted electronically or in writing, within the time limits specified in Section 11.03, in a method or form authorized by the Fund.

9.02 PROCESSING CLAIMS INVOLVING URGENT CARE

- (a) The Fund, (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09) upon receipt of a claim involving urgent care (as defined in Section 9.02[c]), shall notify the claimant of the benefit determination (whether adverse or not) as soon as possible, taking into account the medical exigencies, but not later than seventy-two (72) hours after receipt of the claim unless the claimant fails to provide sufficient information to determine whether and/or to what extent benefits are covered or payable in accordance with this Plan. If the claimant fails to provide such sufficient information, the Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09) shall notify the claimant as soon as possible, but not later than twenty-four (24) hours after the Fund receives the claim, of the specific information necessary to complete the claim. The claimant shall then be afforded a reasonable amount of time, taking into account the circumstances, but not less than fortyeight (48) hours, to provide the specified information, and the Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09) shall thereafter notify the claimant of the benefit determination as soon as possible, but in no case later than forty-eight (48) hours after the earlier of:
 - (1) the receipt of the specified information;
 - (2) the end of the period afforded the claimant to provide the specified additional information.
- (b) Notice of any adverse benefit determination pursuant to this Section 9.02 shall be provided in accordance with Section 9.04.
- (c) A "claim involving urgent care" is any claim for medical care or treatment with respect to which the application of the time periods for making non-urgent care determinations (as those periods are specified in Section 9.03):
 - (1) could seriously jeopardize the life or health of the claimant or the ability of the claimant to regain maximum function; or
 - (2) would, in the opinion of a physician with knowledge of the claimant's medical condition, subject the claimant to severe pain that cannot be adequately managed without the care or treatment that is the subject of the claim.

Any claim that a physician with knowledge of the claimant's medical condition determines is a "claim involving urgent care" shall be treated as a "claim involving urgent care." In the absence of such a physician's determination, any question whether or not a claim is a "claim involving urgent care" is to be

determined by an individual acting on behalf of the Fund and applying the judgment of a prudent layperson who has average knowledge of health and medicine.

9.03 PROCESSING CLAIMS FOR BENEFITS (OTHER THAN URGENT CARE CLAIMS)

- (a) The Fund, upon its receipt of a claim that is neither a claim involving urgent care nor a claim involving a benefit described in Section 9.03(b) (which relates to benefits requiring the Fund's pre-approval), shall notify the claimant of the Fund's benefit determination (if it is an adverse benefit determination) within a reasonable period of time and not later than thirty (30) days after the Fund receives the claim, provided that this period may be extended for an additional fifteen (15) days if the Fund both determines that such an extension is necessary due to matters beyond the control of the Fund and notifies the claimant, prior to the expiration of the initial thirty (30)-day period, of the circumstances requiring the extension of time and the date by which the Fund expects to render a decision. If such an extension is necessary due to a failure of the claimant to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the claimant shall be afforded at least forty-five (45) days from receipt of the notice within which to provide the specified information.
- (b) The Fund, upon its receipt of a claim (not involving urgent care) for a benefit, the receipt of which is conditioned by this Plan upon required approval by the Fund in advance of obtaining the medical care, shall notify the claimant of the Fund's benefit determination (whether adverse or not) within a reasonable period of time appropriate to the medical circumstances but not later than fifteen (15) days after the Fund receives the claim, provided that this period may be extended for an additional fifteen (15) days if the Fund both determines that such an extension is necessary due to matters beyond the control of the Fund and notifies the claimant, prior to the expiration of the initial fifteen (15)-day period, of the circumstances requiring the extension of time and the date by which the Fund expects to render a decision. If such an extension is necessary due to a failure of the claimant to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the claimant shall be afforded at least forty-five (45) days from receipt of the notice within which to provide the specified information.
- (c) In the event that a time period for notice of any benefit determination by the Fund is extended pursuant to this Section 9.03 in order for the claimant to submit information necessary to decide the claim, the time period for making the benefit determination and providing related notice shall be tolled (i.e., not counted) from the date on which the notification of the extension is sent to the claimant until the date on which the claimant responds to the request for additional information.
- (d) Notice of any adverse benefit determination pursuant to this Section 9.03 shall be provided in accordance with Section 9.04.

9.04 NOTICE OF ADVERSE BENEFIT DETERMINATIONS

(a) Whenever an adverse benefit determination (as defined in Section 9.04(c)) is made by the Fund, except upon a claim involving urgent care (in which instance Section 9.04(b) governs the notice), the Fund shall provide the claimant with written (or electronic) notice

of the determination that shall include statements, in a manner calculated to be understood by the claimant, of the following:

- (1) the specific reason or reasons for each adverse benefit determination;
- (2) references to the specific provisions of this Plan on which each adverse benefit determination is based;
- (3) a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary;
- (4) a description of the Fund's appellate review procedures and the time limitations applicable to those procedures, including a statement of the claimant's right to bring a civil action pursuant to Section 502 of the Employee Retirement Income Security Act following an adverse benefit determination at the end of appellate review by the Fund;
- (5) in the case of any adverse benefit determination,
 - (i) if an internal rule, guideline, protocol or other similar criterion was relied upon in making the adverse benefit determination, either a statement of the contents of the specific rule, guideline, protocol or other criterion or a statement that the specific rule, guideline, protocol or other criterion will be provided free of charge to the claimant upon request; and
 - (ii) if the adverse benefit determination is based on a medical-necessity requirement or an experimental-treatment exclusion or a similar exclusion or limitation, either an explanation of the scientific or clinical judgment for the determination, applying the terms of this Plan to the claimant's medical circumstances, or a statement that such explanation will be provided free of charge to the claimant upon request; and
- (6) in the case of any adverse benefit determination of a claim involving urgent care, a description of the expedited review process that is applicable to such claims.
- (b) Whenever an adverse benefit determination (as defined in Section 9.04[c]) is made by the Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09) upon a claim involving urgent care, the Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09) may provide the information described in Section 9.04(a) to the claimant by oral notification within the time limitations prescribed in Section 9.02(a), provided that written (or electronic) notice of the determination that includes the information described in Section 9.04(a) is also to be furnished to the claimant not later than three (3) days after the oral notification.
- (c) An "adverse benefit determination" means any of the following: a denial, reduction or termination of, or a failure to provide or make payment (in whole or in part) or a recoupment for, a benefit, including any such denial, reduction, termination, recoupment or failure to provide or make payment that is based on a Plan exclusion of Coverage or a Plan limitation of Coverage as applied to a claim for benefits, or that is based on a

determination relative to the question of a Covered Individual's or any other individual's eligibility for Coverage.

9.05 **CONCURRENT CARE DECISIONS**

- (a) If the Fund has approved an ongoing course of treatment to be provided over a period of time and/or to include a number of treatments, any reduction or termination by the Fund of such course of treatment (other than by plan amendment or termination) before the end of such period of time or number of treatments shall constitute an adverse benefit determination for all purposes of this Plan. In such an event, the Fund shall notify the claimant, in accordance with Section 9.04, of the adverse benefit determination at a time sufficiently in advance of the reduction or termination to allow the claimant to appeal and to obtain a determination on review of that adverse benefit determination before the benefit is reduced or terminated.
- (b) Any request by a claimant to extend such an ongoing course of treatment, previously approved by the Fund, beyond the approved period of time or the approved number of treatments shall, if the course of treatment is a claim involving urgent care, be decided as soon as possible, taking into account the medical exigencies, and the Fund shall notify the claimant of the Fund's benefit determination (whether adverse or not) within twenty-four (24) hours after receipt of the request by the Fund, provided that any such request is received by the Fund at least twenty-four (24) hours prior to the expiration of the prescribed period of time or number of treatments.
- (c) Whenever an adverse benefit determination (as defined in Section 9.04[c]) is made by the Fund upon a request by a claimant to extend an ongoing course of treatment, previously approved by the Fund, beyond the approved period of time or the approved number of treatments, whether or not urgent care is involved, notice of the determination shall be provided in accordance with Section 9.04.

9.06 MISCELLANEOUS BENEFIT CLAIMS PROVISIONS

- (a) Any time limitation specified in this Article IX for a determination and/or a notice by the Fund may be waived and/or modified at any time on the basis of a request, agreement or consent by the claimant or by an authorized representative of the claimant, including a retroactive waiver and/or modification of an applicable time limitation after it has expired.
- (b) The burden of proof in demonstrating any fact essential to the approval of any claim for benefits, including eligibility for any claimed benefit and the extent to which a claimed benefit is covered or payable in accordance with this Plan, shall at all times be the responsibility of the claimant.
- (c) It is a condition precedent to any civil action by a Covered Individual or other individual to recover benefits covered or payable in accordance with this Plan and/or to clarify the individual's rights to past, present or future benefits covered or payable in accordance with this Plan, including any civil action pursuant to Section 502 of the Employee Retirement Income Security Act, that the claimant or other individual files a benefit claim and initiates and actively pursues appellate review of any adverse benefit determination

- upon any claim, and secures all related benefit determinations by the Fund, in accordance with Articles IX and X of this Plan, prior to the commencement of any civil action.
- (d) To the extent that a Hospital, Physician or other provider or person is assigned a claim of a Covered Individual for reimbursement by the Fund of the costs of medical or other services or benefits, any and all rights and authority of such assignee:
 - (1) are limited by the validity, enforceability and terms of the assignment;
 - (2) are limited by all exclusions, limitations, terms and provisions of this Plan;
 - (3) are subordinate to any claims and defenses of the Fund against the Covered Individual; and
 - (4) are conditioned upon complete compliance by the assignee with all conditions and requirements imposed upon claimants by Articles IX and X of this Plan, including the requirement that the assignee (as claimant) files a benefit claim and initiates and actively pursues appellate review of any adverse benefit determination upon any claim, and secures all related benefit determinations by the Fund, in accordance with Articles IX and X, prior to the commencement of any civil action.

ARTICLE X. APPELLATE REVIEW PROCEDURES AND DETERMINATIONS

10.01 PROCEDURES DURING APPELLATE REVIEW OF ADVERSE BENEFIT DETERMINATIONS

- (a) Whenever an adverse benefit determination (as defined in Section 9.04[c]) is made by the Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09), there are multiple available stages of appellate review of the determination, as follows:
 - (1) for any Trustee-Reviewable Determination, as defined in Section 10.02, there are two available stages of appellate review of that determination, the first of which is conducted by the Appeals Committee and the second of which is conducted by the Trustee Appellate Review Committee;
 - (2) for any adverse benefit determination upon a claim involving urgent care (as defined in Section 9.02[c]), there is a single stage of appellate review, which is conducted by the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09 and, in all other cases, is conducted by the Appeals Committee; and
 - (3) for any other adverse benefit determination ("Other Determination") that is neither a Trustee-Reviewable Determination nor an adverse benefit determination upon a claim involving urgent care, there are two available stages of appellate review of that determination, the first of which is conducted by the Staff Interim-Review Committee and the second of which is conducted by the Staff Final-Review Committee.

The Appeals Committee and the Staff Final-Review Committee each shall be composed of one or more employees of the Fund appointed as members of the committee by the Executive Director of the Fund, provided that the Executive Director retains the authority to terminate any such appointment at any time. The Staff Interim-Review Committee shall be composed of one or more employees of the Fund. If a claimant requests appellate review of multiple adverse benefit determinations, and they consist of both one or more Trustee-Reviewable Determinations and one or more Other Determinations, the Fund in its sole discretion may consolidate all of the determinations into the same appellate review.

- (b) All authority and responsibilities of the Board of Trustees with respect to appellate review of adverse benefit determinations is delegated to a committee of Trustees designated as the Trustee Appellate Review Committee.
- (c) The following procedures shall govern the operations of the Trustee Appellate Review Committee:
 - (1) a quorum of the Trustees at any meeting of the Trustee Appellate Review Committee, for the conduct of its business and for all benefit determinations on review by that committee, shall be at least one Employer Trustee and at least one Employee Trustee (all Trustee members of the Board of Trustees are and shall be de facto members of the Trustee Appellate Review Committee);

- (2) for each matter voted upon at any meeting of the Trustee Appellate Review Committee, the Employee Trustees and the Employer Trustees shall each have the same number of votes based upon the larger number (of Employee Trustees or Employer Trustees) in attendance, provided that each vote shall be cast as the vote of an individual Trustee and not as part of a block, and each determination by the Trustee Appellate Review Committee shall be based upon a majority vote of those present and voting;
- (3) the meetings of the Trustee Appellate Review Committee shall be monthly according to a schedule approved by the Trustees;
- (4) the Trustees who attend and participate in any meeting of the Trustee Appellate Review Committee shall be vested, relative to all appellate review of adverse benefit determinations, with all authority and responsibilities of the Board of Trustees established by the Fund's benefit plan documents, as heretofore and hereafter amended, including discretionary and final authority in making determinations during all such appellate review; and
- (5) the records of monthly meetings of the Trustee Appellate Review Committee, and of its determinations during appellate review, shall be regularly kept and maintained with records of meetings of the Board of Trustees.
- (d) At all stages of appellate review of any adverse benefit determination, the following procedures shall be enforced:
 - (1) the claimant shall be provided an opportunity to submit written comments, documents, records and other information relating to the claim for benefits;
 - (2) the claimant shall be provided, upon request and free of charge, reasonable access to, and copies of, all documents, records and other information possessed by the Fund and relevant to the claimant's claim for benefits;
 - (3) the appellate review shall take into account all comments, documents, records and other information submitted by the claimant relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination;
 - (4) the appellate review shall not afford deference to the initial adverse benefit determination by the Fund and shall be conducted by one or more individuals each of whom shall be an appropriate named fiduciary of the Fund who is neither an individual who made the adverse benefit determination that is the subject of the review nor a subordinate of any such individual;
 - (5) the appellate review shall require that, in deciding an appeal of any adverse benefit determination that is based in whole or in part on a medical judgment, including any determination whether a particular treatment, drug or other item is experimental, investigational or not medically necessary or appropriate, the appropriate named fiduciary shall consult with a health care professional who has appropriate training and experience in the field of medicine involved in the medical judgment;

- (6) the appellate review shall require the identification to the claimant of any medical or vocational expert whose advice was obtained on behalf of the Fund in connection with the claimant's adverse benefit determination, whether or not the advice was relied upon in making that determination;
- (7) the appellate review shall require that each health care professional engaged by the appropriate named fiduciary for purposes of a consultation during appellate review, pursuant to this Section 10.01, shall be an individual who is neither an individual who was consulted in connection with the adverse benefit determination that is the subject of the review nor a subordinate of any such individual; and
- (8) the appellate review in the case of a claim involving urgent care (as defined in Section 9.02[c]) shall require an expedited review process pursuant to which a request for an expedited appeal from an adverse benefit determination may be submitted orally or in writing by the claimant, and all necessary information, including the benefit determination on review, shall be transmitted to the claimant by telephone, facsimile or other available and similarly expeditious method.

10.02 **DEFINITION OF TRUSTEE-REVIEWABLE DETERMINATIONS**

"Trustee-Reviewable Determinations" are defined to include any adverse benefit determination (as defined in Section 9.04[c]) which is within any of the following classifications (other than a *de minimis* determination which means a single or series of adverse benefit determinations upon monetary claims which involve potential aggregate Fund liability no greater than \$5,000: *de minimis* determinations shall be reviewed as Other Determinations):

- (a) all adverse benefit determinations based upon Article III (PARTICIPATION AND COVERAGE);
- (b) all adverse benefit determinations based upon Article IV (GENERAL EXCLUSIONS, LIMITATIONS AND CONDITIONS FOR PAYMENT OF CLAIMS) except determinations based upon Section 4.16 (EXCLUSION OF PAYMENT OVER PRESCRIBED MAXIMUMS);
- (c) all adverse benefit determinations based upon Section 11.14 (SUBROGATION) or Section 11.15 (WORKERS' COMPENSATION SUBROGATION);
- (d) all adverse benefit determinations based upon Article XII (BASIC BENEFITS) except determinations upon claims for Prescription Drug Benefits (Section 12.07), Hearing Aid Benefits (Section 12.11) and Chiropractic Expense Benefits (Section 12.14);
- (e) all adverse benefit determinations based upon Article XIII (MAJOR MEDICAL EXPENSE BENEFITS);
- (f) all adverse benefit determinations based upon Article XIV (LIFE INSURANCE BENEFITS); and
- (g) all other types of adverse benefit determinations which the Fund expressly classifies as Trustee-Reviewable Determinations.

10.03 TIME LIMITATIONS FOR APPELLATE REVIEW OF ADVERSE BENEFIT DETERMINATIONS

- (a) Whenever an adverse benefit determination is made, including a determination affecting a payment amount claimed by a Covered Individual or any medical provider who is an assignee or beneficiary of a Covered Individual, the claimant may initiate appellate review of the determination by submission to the Fund, within one hundred eighty (180) days after the claimant's receipt of the notice of such adverse benefit determination, of a request for such appellate review. All requests for appellate review shall be submitted, electronically or in writing, in a method or form authorized by the Fund.
- (b) The Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09), upon receipt of a claimant's timely request for appellate review of an earlier adverse benefit determination upon a claim involving urgent care (as defined in Section 9.02[c]), shall notify the claimant of the benefit determination by the Appeals Committee as soon as possible, taking into account the medical exigencies, but not later than seventy-two (72) hours after the Fund receives the claimant's request for appellate review.
- (c) The Fund, upon its receipt of a claimant's timely request for appellate review of a Trustee-Reviewable Determination, shall perform and complete appellate review, and shall notify the claimant of the determinations upon completion of such review, in accordance with the following time limitations:
 - (1) all appellate review and benefit determinations by the Appeals Committee shall be completed, and the Fund shall provide written notice to the claimant of those determinations, no later than thirty (30) days after the Fund's receipt of the claimant's timely request for appellate review of a Trustee-Reviewable Determination;
 - (2) whenever an adverse benefit determination is made by the Appeals Committee at the end of its appellate review, the claimant may initiate appellate review by the Trustee Appellate Review Committee, by request to the Fund within one hundred eighty (180) days after the claimant's receipt of the Fund's notice of such determination;
 - (3) all appellate review and benefit determinations by the Trustee Appellate Review Committee shall be completed within a reasonable time and at the <u>first</u> monthly meeting that takes place on a date thirty (30) or more days after the Fund receives the claimant's timely request for appellate review by the Trustee Appellate Review Committee; and
 - (4) after appellate review and benefit determinations by the Trustee Appellate Review Committee, the Fund shall provide written notice to the claimant of those determinations by the Trustees no later than five (5) days after the determinations are made.
- (d) The Fund, upon its receipt of a claimant's timely request for appellate review of an Other Determination, shall perform and complete appellate review, and shall notify the claimant

of the determinations upon completion of such review, in accordance with the following time limitations:

- (1) all appellate review and benefit determinations by the Staff Interim-Review Committee shall be completed, and the Fund shall provide written notice to the claimant of those determinations, no later than thirty (30) days after the Fund's receipt of the claimant's timely request for appellate review of an Other Determination;
- (2) whenever an adverse benefit determination is made by the Staff Interim-Review Committee at the end of its appellate review, the claimant may initiate appellate review by the Staff Final-Review Committee, by written request to the Fund within one hundred eighty (180) days after the claimant's receipt of the Fund's notice of such determination; and
- (3) all appellate review and benefit determinations by the Staff Final-Review Committee shall be completed, and the Fund shall provide written notice to the claimant of those determinations, no later than thirty (30) days after the Fund's receipt of the claimant's timely request for appellate review by the Staff Final-Review Committee.
- (e) The Fund, upon its receipt of a claimant's timely request for appellate review of an adverse benefit determination upon a claim (not involving urgent care) for a benefit, the receipt of which is conditioned by this Plan upon required approval by the Fund in advance of obtaining the medical care, shall arrange a single stage of appellate review within a reasonable period of time appropriate to the medical circumstances, provided that the Appeals Committee shall complete (and send the claimant notice of) the Fund's benefit determination on review no later than thirty (30) days after the Fund receives the claimant's request for appellate review.
- (f) Notice of any adverse benefit determination pursuant to this Section 10.03 shall be provided in accordance with Section 10.04.

10.04 NOTICE OF BENEFIT DETERMINATIONS AFTER APPELLATE REVIEW

Whenever a benefit determination is made after appellate review, the Fund (or the applicable TeamCare network administrator with respect to claims under Sections 12.08 and 12.09) shall provide the claimant with written (or electronic) notice of the determination that shall include statements, in a manner calculated to be understood by the claimant, of the following:

- (a) the specific reason or reasons for each adverse benefit determination;
- (b) references to the specific provisions of this Plan on which each adverse benefit determination is based;
- (c) a statement that the claimant is entitled to receive, upon request and free of charge, reasonable access to, and copies of, all documents, records and other information relevant to the claimant's claim for benefits:
- (d) a description of the Fund's appellate review procedures and the time limits applicable to such procedures, including a statement of the claimant's right to bring a civil action

pursuant to Section 502 of the Employee Retirement Income Security Act following an adverse benefit determination at the end of appellate review by the Fund; and

- (e) in the case of any adverse benefit determination relating to disability benefits,
 - (1) if an internal rule, guideline, protocol or other similar criterion was relied upon in making the adverse benefit determination, either a statement of the contents of the specific rule, guideline, protocol or other criterion or a statement that the specific rule, guideline, protocol or other similar criterion was relied upon in making the adverse determination and that a copy will be provided free of charge to the claimant upon request; and
 - (2) if the adverse benefit determination is based on a medical-necessity requirement or an experimental-treatment exclusion or a similar exclusion or limitation, either an explanation of the scientific or clinical judgment for the determination, applying the terms of this Plan to the claimant's medical circumstances, or a statement that such explanation will be provided free of charge to the claimant upon request.
- (e) a statement describing the availability, upon request, of the diagnosis code and its corresponding meaning, and the treatment code and its corresponding meaning; and
- (f) a statement describing the availability of external review for an adverse benefit determination.

10.05 MISCELLANEOUS APPELLATE REVIEW PROVISIONS

- (a) Any time limitation specified in this Article X for a determination and/or a notice may be waived and/or modified at any time on the basis of a request, agreement or consent by the claimant or by an authorized representative of the claimant, including a retroactive waiver and/or modification of an applicable time limitation after it has expired.
- (b) In the event that any time period for any appellate review of an earlier adverse benefit determination, and of notice of the determinations upon completion of such review, is extended based upon a failure by the claimant to submit information necessary to decide the claim, each time period for the conduct and completion of such appellate review, and for making benefit determinations, and of providing notice of those determinations, relative to the claimant's claim, shall be tolled (i.e., not counted) from the date on which the notification of the extension is sent to the claimant until the date on which the claimant responds to the request for additional information.
- (c) Each individual who is authorized to conduct interim appellate review (as a member of the Appeals Committee or the Staff Interim-Review Committee) is vested with discretionary and final authority in making any determination within the scope of this Article X, except that, upon further appellate review by the Trustee Appellate Review Committee or the Staff Final-Review Committee, the prior discretionary and final authority of the interim appellate-review agency is displaced by the discretionary and final authority of the final appellate-review agency (the Trustee Appellate Review Committee or the Staff Final-Review Committee), which shall not afford any deference to any determination by the interim appellate-review agency.

- (d) The Trustees are vested with discretionary and final authority in making any determination within the scope of this Article X. The Trustees appoint the applicable TeamCare network administrators as the named fiduciaries authorized to administer the claim and appeal procedures set forth in the Plan with respect to any and all claims for urgent care under Sections 12.08 and 12.09 and any appeals arising out of such claims. Each applicable TeamCare network administrator shall be vested with discretionary and final authority in making any determination concerning such claim and appeal. The applicable TeamCare network administrators will strictly follow the terms of the Plan and any applicable federal regulations when resolving claim appeal determinations on behalf of the Plan.
- (e) The burden of proof in demonstrating any fact essential to the approval of any claim for benefits, including eligibility for any claimed benefit and the extent to which a claimed benefit is covered or payable in accordance with this Plan, shall at all times be the responsibility of the claimant, provided that the Fund will at all times during appellate review of an adverse benefit determination provide to the claimant, upon request and free of charge, reasonable access to, and copies of, all documents, records and other information possessed by the Fund and relevant to the claimant's claim for benefits.
- (g) It is a condition precedent to any civil action by a Covered Individual or other individual to recover benefits covered or payable in accordance with this Plan and/or to clarify the individual's rights to past, present or future benefits covered or payable in accordance with this Plan, including any civil action pursuant to Section 502 of the Employee Retirement Income Security Act, that the claimant or other individual files a benefit claim and initiates and actively pursues appellate review of any adverse benefit determination upon any claim, and secures all related benefit determinations by the Fund, in accordance with Articles IX and X of this Plan, prior to the commencement of any civil action.

10.06 **EXTERNAL REVIEW**

- (a) A claimant may file a request for an external review with the Fund if the request is filed within four months after the date of receipt of a notice of an adverse benefit determination or final adverse benefit determination that involves medical judgment (including, but not limited to, those based on the Plan's requirements for medical necessity, appropriateness, health care setting, level of care, or effectiveness of a covered benefit; or its determination that a treatment is experimental or investigational), as determined by the external reviewer; or a rescission of coverage (whether or not the rescission has any effect on any particular benefit at that time). If there is no corresponding date four months after the date of receipt of such a notice, then the request must be filed by the first day of the fifth month following the receipt of the notice. For example, if the date of receipt of the notice is October 30, because there is no February 30, the request must be filed by March 1. If the last filing date would fall on a Saturday, Sunday, or Federal holiday, the last filing date is extended to the next day that is not a Saturday, Sunday, or Federal holiday.
- (b) Within five business days following the date of receipt of the external review request, the Fund must complete a preliminary review of the request to determine whether:

- (1) The claimant is or was covered under the Plan at the time the health care item or service was requested or, in the case of a retrospective review, was covered under the Plan at the time the health care item or service was provided;
- (2) The adverse benefit determination or the final adverse benefit determination does not relate to the claimant's failure to meet the requirements for eligibility under the terms of the Plan (e.g. worker classification or similar determination);
- (3) The claimant has exhausted the Plan's internal appeal process unless the claimant is not required to exhaust the internal appeals process under the law;
- (4) The claimant has provided all the information and forms required to process an external review.

Within one business day after completion of the preliminary review, the plan must issue a notification in writing to the claimant. If the request is complete but not eligible for external review, such notification must include the reasons for its ineligibility and contact information for the Employee Benefits Security Administration (toll-free number 866-444-EBSA (3272)). If the request is not complete, such notification must describe the information or materials needed to make the request complete and the Plan must allow a claimant to perfect the request for external review within the four-month filing period or within the 48 hour period following the receipt of the notification, whichever is later.

- (c) The Plan must assign an independent review organization (IRO) that is accredited by URAC or by a similar nationally-recognized accrediting organization to conduct the external review.
- (d) The assigned IRO will timely notify the claimant in writing of the request's eligibility and acceptance for external review. This notice will include a statement that the claimant may submit in writing to the assigned IRO within ten business days following the date of receipt of the notice additional information that the IRO must consider when conducting the external review. The IRO is not required to, but may, accept and consider additional information submitted after ten business days.
- (e) Within five business days after the date of assignment of the IRO, the Plan must provide to the assigned IRO the documents and any information considered in making the adverse benefit determination or final internal adverse benefit determination. Within one business day after making the decision, the IRO must notify the claimant and the Plan.
- (f) Upon receipt of any information submitted by the claimant, the assigned IRO must within one business day forward the information to the plan. Upon receipt of any such information, the plan may reconsider its adverse benefit determination or final internal adverse benefit determination that is the subject of the external review. Reconsideration by the plan must not delay the external review. The external review may be terminated as a result of the reconsideration only if the plan decides, upon completion of its reconsideration, to reverse its adverse benefit determination or final adverse benefit

determination and provide coverage or payment. Within one business day after making such a decision, the Plan must provide written notice of its decision to the claimant and the assigned IRO. The assigned IRO must terminate the external review upon receipt of the notice from the Plan.

- (g) The assigned IRO must provide written notice of the final external review decision within 45 days after the IRO receives the request for the external review. The IRO must deliver the notice of final external review decision to the claimant and the Plan.
- (h) A claimant may make a request for an expedited external review at the time the claimant receives:
 - (1) An adverse benefit determination if the adverse benefit determination involves a medical condition of the claimant for which the time frame for completion of an expedited internal appeal under the law would seriously jeopardize the life or health of the claimant or would jeopardize the claimant's ability to regain maximum function and the claimant has filed a request for an expedited internal appeal; or
 - (2) A final internal adverse benefit determination, if the claimant has a medical condition where the time frame for completion of a standard external review would seriously jeopardize the life or health of the claimant or would jeopardize the claimant's ability to regain maximum function, or if the final internal adverse benefit determination concerns an admission, availability of care, continued stay, or health care item or service for which the claimant received emergency services, but has not been discharged from a facility.
 - (i) Immediately upon receipt of the request for expedited external review, the Plan must determine whether the request meets the reviewability requirements set forth in paragraphs (a) and (b) above for standard external review. The Plan must immediately send a notice to that meets the requirements of paragraph (b) above for standard external review to the claimant of its eligibility determination.
 - (ii) Upon a determination that a request is eligible for external review following the preliminary review, the Plan will assign an IRO pursuant to the requirements set forth in paragraph (c) above for standard review. The Plan must provide or transmit all necessary documents and information considered in making the adverse benefit determination or final adverse benefit determination to the assigned IRO electronically or by telephone or facsimile or any other available expeditious method.
- (i) The IRO must provide notice of the final external review decision as expeditiously as the claimant's medical condition or circumstances require, but in no event more than 72 hours after the IRO receives the request for an expedited external review. If the notice is not in writing, within 48 hours after the date of providing that notice, the assigned IRO must provide written confirmation of the decision to the claimant and the Plan.

ARTICLE XI. MISCELLANEOUS PROVISIONS

11.01 VALIDITY OF CHANGES IN THE PLAN

No agent has authority to change the terms of this Plan or to waive any of its provisions. No change in this Plan shall be valid unless adopted by the Trustees of the Fund.

11.02 CLAIM FORMS

The Fund, upon written request of the claimant, will furnish to the claimant such forms as are required for filing a claim.

11.03 TIME WITHIN WHICH CERTAIN CLAIMS ARE TO BE FILED

A claim for any loss (excluding Total and Permanent Disability Installment Benefit or Waiver of Premium) must be filed within one (1) year after the date of such loss. A claim for Life Insurance, Accidental Death and Dismemberment, Total and Permanent Disability Installment Benefit or Waiver of Premium must be filed within three (3) years after the date of the event.

11.04 RECOVERY OF EXCESS PAYMENTS

Whenever this Plan has made benefit payments which exceed the amount of benefits payable under the terms of this Plan, the Fund shall have the right to recover the excess payments from any responsible persons or entities, including the right to deduct the amount of excess payments from any subsequent payable benefits.

11.05 FUND MAY ORDER PHYSICAL EXAMINATION AND/OR AUTOPSY

The Fund shall have the right and opportunity to have a claimant examined by a medical service provider, of the Fund's choosing and at the Fund's expense, when, and so often as it may reasonably require during the pendency of a claim. In the case of death, the Fund shall also have the right to request an autopsy where it is not forbidden by law.

11.06 TO WHOM BENEFITS ARE PAYABLE

All benefits are payable to, or for the benefit of, a Covered Individual or his estate, except as otherwise provided in this Plan.

Any provider of medical or healthcare services or goods receiving, or seeking to receive, payment from the Fund will, in the absence of evidence to the contrary, be presumed to have claimed a right to do so pursuant to a valid assignment of benefits under the Plan from a Covered Individual. Any such provider will accordingly be presumed to have presented a claim as a beneficiary under this Plan and will be bound by all provisions of this Plan, including but not limited to, the provisions relating to Amendments and Plan Termination (Article VII), Plan Administration (Article VIII), Benefit Claims (Article IX), Appellate Review Procedures and Determinations (Article X), and Miscellaneous Provisions (Article XI).

11.07 CERTAIN ACTS OF THE FUND DO NOT CONSTITUTE A WAIVER OF RIGHTS

The furnishing of forms by the Fund for filing a claim for benefits, or the acceptance of such filings or the investigation of any claim hereunder, shall not operate as a waiver of any rights of the Fund.

11.08 NO MEDICAL EXAMINATION REQUIRED AS PREREQUISITE TO COVERAGE

In no case shall any individual be required to submit to a medical examination as a prerequisite to Coverage under this Plan.

11.09 ALL MEDICAL BENEFIT PAYMENTS BASED ON ALLOWED AMOUNT FOR THE SERVICE

In all instances, other than when a specific dollar amount is the stated allowance, medical benefits to be paid by the Fund will be based upon the Allowed Amount which is defined as follows:

- (a) For providers or facilities which have a written agreement with the Fund or with a network utilized by the Fund, the amount specified in such agreement.
- (b) For providers that do not have a written agreement with the Fund or with a network utilized by the Fund, the lesser of (1) the provider's billed charges; and (2) 200% of the base Medicare reimbursement rate, excluding any Medicare adjustment(s).
- (c) For facilities that do not have a written agreement with the Fund or with a network utilized by the Fund, the lesser of (1) the facility's billed charges; and (2) 175% of the base Medicare reimbursement rate, excluding any Medicare adjustment(s).
- (d) If there is no base Medicare reimbursement rate for a claim under (b) and (c), then the lesser of (1) the provider's or facility's billed charge; and (2) the median rate that would be paid for an in-network claim for such claim.
- (e) If there is no base Medicare reimbursement rate for a claim under (b) and (c), and no median rate that would be paid for an in-network claim for such claim under (d), then the Reasonable and Customary amount.
- (f) In determining the Allowed Amount, all other Plan rules, conditions, exclusions, and limitations apply, including without limitation the specified percentages of the Allowed Amount payable under the Plan.

11.10 PERIOD DURING WHICH BENEFIT PAYMENTS MUST BE CLAIMED

Any benefit amounts payable under this Plan must be claimed by the proper beneficiaries within a period of six (6) years from the date that such amounts become due and payable to such beneficiaries. Benefits unclaimed after this six (6) year period shall be considered the property of the Fund which shall have the immediate right to recover the amount of any unclaimed benefits from any persons in possession of said benefit amounts or from any Other Plans having primary responsibility for payment of such benefits.

11.11 APPLICABLE LAW

The Trust Agreement was created and accepted in the State of Illinois. All questions pertaining to the validity or construction of the Trust Agreement shall be determined in accordance with the laws of the State of Illinois.

The Fund is a self-funded employee benefit plan governed by the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), 29 U.S.C. § 1001 et seq. All questions concerning the validity of the terms of the Plan shall be determined under ERISA. Any state law that relates to the operation or administration of the Plan is preempted by ERISA and this Plan. The Fund shall be entitled to assert a lien against third parties, insurers and attorneys when necessary to protect the rights of the beneficiaries of the Plan or when necessary to protect the Fund's rights to recover any reimbursements provided for by the terms of this Plan.

11.12 **SEVERABILITY OF PLAN PROVISIONS**

Should any provision of this Plan be held null and void by a court of competent jurisdiction, such holding shall not adversely affect any other provision of this Plan.

11.13 RIGHT TO REVIEW ALL CLAIMS

The Fund reserves the right to question any charge or procedure and to have the same professionally reviewed to determine if it is covered under the Plan. The results of said professional review shall not be binding on the Trustees.

11.14 **SUBROGATION**

- (a) The Fund, whenever it makes any payment for any benefits on behalf of a Covered Individual or other person related to any illness, injury or disability (collectively and separately "Disability") of the person, is immediately subrogated and vested with subrogation rights ("Subrogation Rights") to all present and future rights of recovery ("Loss Recovery Rights") arising out of the Disability which that person and his/her parents, heirs, guardians, executors, attorneys, agents and other representatives (individually and collectively called the "Covered Individuals") may have. The Fund's Subrogation Rights extend to all Loss Recovery Rights of the Covered Individual. The Loss Recovery Rights of the Covered Individual include, without limitation, all rights based upon any one or more of the following:
 - (1) Any act or omission by any person or entity, including the Covered Individual; and/or
 - (2) Any policy, contract, plan or other document creating responsibility for any insurance, indemnity or reimbursement (collectively "insurance") including but not limited to every document within the definition of "Other Plan" in Section 1.53 and also including every other form of no-fault liability insurance, personal-injury-protection insurance, financial responsibility insurance, uninsured and/or underinsured motorist insurance and any casualty liability insurance or medical payments coverage including, but not limited to, homeowners or premises insurance, school insurance, workers' compensation insurance, athletic team insurance and any other specific risk insurance or coverage; and/or

- (3) Any medical reimbursement insurance; and/or
- (4) Any government-funded or government-sponsored financial entity which may be a source of payment or reimbursement of Loss Recovery Rights to a Covered Individual.
- (b) The Covered Individual shall promptly advise the Fund any time he/she is pursuing any third-party for any Loss Recovery Rights arising out of any Disability. In addition, the Covered Individual shall fully cooperate with the Fund in enforcement of the Fund's Subrogation Rights, shall make prompt, full, accurate and continuous disclosures to the Fund's representatives of all information about all circumstances of his/her Disability and about all other specifics of his/her Loss Recovery Rights (including prompt, full, accurate and continuous disclosures of the specifics of applicable insurance and all other potential sources of recovery), shall upon request by a Fund representative execute whatever documents are appropriate to enforce and preserve the Fund's Subrogation Rights, shall perform whatever acts are requested by a Fund representative to enable the Fund to effectively prosecute a civil action in the name of the Covered Individual and/or the Fund and one or more Trustees if the Fund deems such action necessary or appropriate and shall refrain from any act or omission that would to any extent prejudice or impair the Fund's Subrogation Rights.
- (c) The payment by the Fund for any benefits on behalf of a Covered Individual related to his/her Disability, and the simultaneous creation of the Fund's Subrogation Rights to the full extent of present and future payments, shall by itself (without any documentation from, or any act by, the Covered Individual) result in an immediate assignment to the Fund of all right, title and interest of the Covered Individual to and in any and all of his/her Loss Recovery Rights to the extent of such payments, and said payment by the Fund on behalf of a Covered Individual shall be deemed to constitute the Covered Individual's direction to his/her attorneys and other representatives to reimburse the full amount of the Fund's Subrogation Rights, from any settlement proceeds or other proceeds (collectively "Proceeds") which are paid to the attorneys or representatives for or on behalf of the Covered Individual, before the Covered Individual receives any Proceeds in full or partial satisfaction of his/her Loss Recovery Rights, and before any fees or expenses are paid, including attorneys' fees.
- (d) No Covered Individual (including his/her attorneys and other representatives) is authorized to act on behalf of the Fund with respect to the Fund's Subrogation Rights, or to receive any payment or reimbursement on behalf of the Fund or to release or impair the Fund's Subrogation Rights to any extent. The Fund is entitled to receive payment and reimbursement in the full amount of the Fund's Subrogation Rights before the Covered Individual receives any Proceeds in full or partial satisfaction of his/her Loss Recovery Rights and before any fees or expenses are paid, including attorneys' fees. If the Fund is vested with Subrogation Rights pursuant to this Section 11.14, then, before the Covered Individual receives any Proceeds, the Covered Individual, and every person and entity that provides any recovery of Proceeds to or on behalf of a Covered Individual, are obligated to cause all such Proceeds to be paid primarily and directly to the Fund until the Fund has received full payment and reimbursement of the Fund's Subrogation Rights.
- (e) If at any time, either before or after the Fund becomes vested with Subrogation Rights pursuant to this Section 11.14, a Covered Individual: 1) fails to inform the Fund that he/she is pursuing any third-party for any Loss Recovery Rights arising out of any Disability, 2)

fails to cooperate in any way with the Fund in connection with the enforcement of its Subrogation Rights, 3) takes any action which impairs or prejudices the Fund's Subrogation Rights, 4) takes any action which seeks to or has a foreseeable consequence of impairing or prejudicing the Fund's Subrogation Rights, or 5) directly or indirectly receives any Proceeds as full or partial satisfaction of his/her Loss Recovery Rights, including arrangements for an annuity or other similar installment benefit plan, and including any payment or reimbursement of expenses (including attorneys' fees) incurred by or on behalf of the Covered Individual, without prior written approval of an authorized Fund representative, the Fund shall be vested with each of the following mutually independent rights:

- (1) The right, at any time, to decline to make any payment for any benefits on behalf of the Covered Individual related to the Disability on which the Loss Recovery Rights were based:
- (2) The right, at any time after the Fund becomes vested with Subrogation Rights, to decline to make any payment for any benefits on behalf of the Covered Individual and his/her Covered Dependents, related to any circumstance or condition for which the Fund otherwise has a Coverage obligation, until the amount of such unpaid Coverage is equal to the unrecovered amount of the Fund's Subrogation Rights;
- (3) The right to recoup any payments made for any benefits on behalf of the Covered Individual related to the Disability on which the Loss Recovery Rights were based; and
- (4) The right, at any time after the Fund becomes vested with Subrogation Rights, to prosecute a civil action against the Covered Individual and/or against any person and/or any other entity (including any insurance company) which the Fund claims to be responsible, in whole or in part, to provide payment or reimbursement to the Fund of the unrecovered amount of the Fund's Subrogation Rights.
- (f) The Fund may assert a lien, for recovery of the Fund's Subrogation Rights against any person or entity. The fact that the Fund does not initially assert or invoke its Subrogation Rights until a time after a Covered Individual, acting without prior written approval of an authorized Fund representative, has made any settlement or other disposition of, or has received any Proceeds as full or partial satisfaction of, his/her Loss Recovery Rights, shall not relieve the Covered Individual of his/her obligation to reimburse the Fund in the full amount of the Fund's Subrogation Rights.
- (g) The Fund shall not be financially responsible for any expenses, including attorneys' fees, incurred by or on behalf of a Covered Individual in the enforcement of his/her Loss Recovery Rights, except to the extent such responsibility is formally accepted by written agreement of an authorized Fund representative.
- (h) The Fund is authorized but not required to bring civil actions in enforcement of the Fund's Subrogation Rights, including direct actions (as subrogee or otherwise) against any person, entity, or Responsible Person which the Fund claims to be responsible, in whole or in part, to provide payment or compensation or reimbursement to the Fund of the unrecovered amount of the Fund's Subrogation Rights, and including actions against any person, entity, or Responsible Person to enjoin any act or practice which violates any terms of the Plan, the Fund's Subrogation Rights and/or to obtain other appropriate

equitable relief to redress such violations and/or to enforce the Fund's Subrogation Rights. A Responsible Person is any person or entity, including attorneys or other representatives of a Covered Individual in any claim for damages for a Disability suffered by the Covered Individual, resulting from any act or omission of another person or entity, and who receives any Proceeds, by way of settlement or award from said claim for damages. If such a civil action is filed by the Fund and the Fund prevails in any amount on any of its claims, all persons, entities and Responsible Person(s) against whom such action is filed shall jointly and severally be responsible for all costs and expenses, including attorneys' fees, incurred by the Fund in connection with or related to such civil action.

- (i) The Trustees are vested with discretionary and final authority in making decisions that interpret plan documents of the Fund that relate to subrogation. Any one or more Trustees and the Executive Director, and any other person authorized by the Trustees or the Executive Director, may, in his/her sole discretion, compromise or settle any Fund claim of Subrogation Rights.
- The Fund's entitlement to full payment and reimbursement of its Subrogation Rights is (j) absolute and unqualified, and is not to be reduced or impaired by the relationship of the gross or net amount of the Proceeds to the aggregate monetary damages sustained, or claimed to be sustained, by the Covered Individual in connection with the Disability related to his/her Loss Recovery Rights. The Fund's Subrogation Rights are not in any way subordinate to or affected by any 'make whole' rule. The Fund's Subrogation Rights are not in any way subordinate to or affected by any 'common fund' principle or factor -sometimes described as the equitable concept of a 'common fund' which governs the allocation of attorney's fees in any case in which a lawyer hired by one party creates through his/her efforts a fund in which others are entitled to share as well. The acceptance of plan benefits from the Fund entirely subordinates the Loss Recovery Rights of the Covered Individual to the Subrogation Rights of the Fund (without any 'common fund' reduction or other reduction of those Subrogation Rights). Every payment and reimbursement to the Fund based upon its Subrogation Rights results in a monetary benefit to all of the Covered Individuals of the Fund.

11.15 WORKERS' COMPENSATION SUBROGATION

If any Covered Individual has a claim denied pursuant to Section 4.03 of this Plan and the Covered Individual's claim for Workers' Compensation benefits is denied by the Workers' Compensation Carrier, the Fund may, at its discretion, enter into an agreement with the Covered Individual to provide benefits during the appeal of the denial. Such an agreement would be entitled "Agreement to Reimburse Central States Health and Welfare Fund" and is subject to the following conditions:

- (a) The Covered Individual and his/her attorney provide proof that a claim is pending before the appropriate Compensation Commission or court;
- (b) The Covered Individual and his/her attorney agree to pursue the claim for Workers' Compensation benefits to a final disposition;
- (c) The Covered Individual and his/her attorney agree to notify the Fund of the disposition of his claim and to notify the Workers' Compensation Carrier of the Agreement;

- (d) The Covered Individual and his/her attorney establish sufficient need for the Fund to consider application of this Section; and
- (e) The Covered Individual and his/her attorney agree to fully reimburse the Fund for benefits paid from the proceeds of any recovery.

The Agreement described in this Section is a binding contract between the Covered Individual, his/her attorney and the Fund, and in the event the Covered Individual and/or his/her attorney does not honor this Agreement, the Fund reserves the right to take any necessary step to protect its interest, including a civil action in enforcement of the Fund's rights under this Section, against the Covered Individual, his/her attorney and/or any other person, entity, or Responsible Person which are obligated under this Section to reimburse the Fund.

If such a civil action is filed by the Fund, and the Fund prevails in any amount on any of its claims, all persons, entities and Responsible Person(s) against whom such action is filed shall jointly and severally be responsible for all costs and expenses, including attorneys' fees, incurred by the Fund in connection with or related to such civil action.

11.16 **RIGHT TO PROVIDE ALTERNATIVE CARE**

The Trustees reserve the right to provide benefits for medical care not addressed in this Plan where such alternative care is in the best interest of the Fund and its beneficiaries, and where the Covered Individual or his or her legal guardian and, in the case of a Child, the Covered Participant agree in writing to such alternative care. Alternative care options will be examined on a case by case basis and subject to the approval of the Trustees.

11.17 PAYMENT BASED ON PROPRIETY OF PROCEDURES

If multiple procedures and/or services are performed on the same day or in the same setting or are inclusive with other services performed, benefits are allowed based on the propriety of the procedures according to accepted medical/dental standards. Reimbursement is based on the Reasonable and Customary allowance for dental claims and the Allowed Amount for medical claims for the appropriate procedure(s).

11.18 SPECIAL ARRANGEMENTS FOR BENEFITS

The Fund enters into agreements with providers from time to time in order to enhance benefits. These providers agree to special arrangements for Fund Participants who patronize them. The Fund separately publishes the details of these agreements. These agreements are, however, subject to change without prior notice.

11.19 **FORUM SELECTION CLAUSE**

All claims (including without limitation claims under the terms of the Plan, the Trust Agreement, or the Employee Retirement Income Security Act of 1974, as amended) against the Plan (including without limitation claims against the Central States, Southeast and Southwest Areas Health and Welfare Fund,

 	•	n employees) shall be filed and strict of Illinois, Eastern Division.

ARTICLE XII. BASIC BENEFITS

12.01 OUTLINE OF BASIC BENEFITS

The Sections of Article XII describe policies and procedures applicable to all Plans offering the referenced Basic Benefit. However, levels of payment, if any, and/or program limitations specific to the Covered Individual's Plan are determined by referencing the appropriate sub-section of Section 20.01.

The specific Basic Benefits and the Sections of this Article pertaining to the same are as follows:

	Section
Short-Term Disability Benefit—Participant Only	12.02
Hospital Expense Benefit	12.03
Surgical and Obstetrical Expense Benefit	12.04
Outpatient Diagnostic X-ray and Laboratory Expense Benefit	12.05
Outpatient Accidental Bodily Injury Expense Benefit	12.06
Prescription Drug Benefit	12.07
Behavioral Health Benefit - Inpatient	12.08
Behavioral Health Benefit - Outpatient	12.09
Organ Transplant Donor Benefit	12.10
Hearing Aid Benefit	12.11
Outpatient Cancer Treatment Benefit	12.12
Ambulance Service Benefit	12.13
Chiropractic Expense Benefit	12.14
Women's Health Benefit	12.15
Mayo Clinic Treatment	
Wellness Benefit	

12.02 SHORT-TERM DISABILITY BENEFIT—PARTICIPANT ONLY

A Covered Participant may receive from the Plan a Short-Term Disability Benefit as follows:

Periods of Disability for the Short-Term Disability Benefit—Participant Only

- (a) A Covered Participant shall receive from the Plan a weekly Short-Term Disability Benefit in the amount and for the maximum benefit period referenced under Section 20.01(a) during a single period of Disability, as determined in Sub-Section (c) of this Section, for Short-Term Disability from employment as a result of being unable to work because of illness, injury or pregnancy. To qualify for the Short-Term Disability Benefit a Covered Participant must:
 - (1) Be absent from work because of a Disability, the treatment of which is compensable under this Plan;
 - (2) Be an Active Employee at the onset of the Disability for which the Short-Term Disability Benefit is claimed; and
 - (3) Following a treatment regimen prescribed by a Physician.
- (b) Benefits shall begin to accrue as follows:
 - (1) For Short-Term Disability due to illness or pregnancy, benefits shall accrue from the eighth (8th) day of lost work time specified by a Physician and verified by the Employer, if the Covered Participant first received medical attention by a Physician within one (1) day before or three (3) days after the date of disablement specified by the Physician as resulting from the illness or pregnancy. If the Covered Participant did not receive medical attention within these time frames, benefits shall accrue from the eighth (8th) day after the date medical attention is received from a Physician; and
 - (2) For Short-Term Disability due to Accidental Bodily Injury, benefits shall accrue from the first (1st) day of lost work time specified by the Physician and verified by the Employer, provided that the Covered Participant first received medical attention within one (1) day before or three (3) days after the date of disablement specified by the Physician as resulting from the injury. Benefits shall accrue from the date medical attention is received from the Physician, if this medical attention was received more than three (3) days after the occurrence of the injury. If a period of Short-Term Disability due to an injury begins more than two (2) weeks after the date of the occurrence of the injury, it shall be treated for purposes of this Sub-Section (1) as Short-Term Disability due to an illness.
- (c) All periods of Short-Term Disability for which a Short-Term Disability Benefit is payable shall be deemed to occur during a single period of Disability except:
 - (1) Short-Term Disability due to related illnesses, injuries or pregnancy shall be deemed to arise during separate periods of Disability, if, and only if, thirty (30) or more consecutive calendar days of active employment separate the dates on which the Covered Participant is absent from work; or
 - (2) Short-Term Disability due to unrelated illnesses, injuries or pregnancy shall be deemed to arise during separate periods of Disability, if, and only if, one (1) day of active employment separates any periods of absence from work.

12.03 HOSPITAL EXPENSE BENEFIT

A Covered Individual may receive from the Plan a Hospital Expense Benefit as follows:

(a) Covered Expenses—

For the number of days and at the level of benefits specified under Section 20.01(b), the Plan shall pay for all covered charges incurred in a Hospital by Covered Individuals for services which are required for purposes of treatment. Coverage shall include, but not be limited to: room and board, general nursing care, operating room, administration of anesthesia, X-ray examinations, laboratory analyses, drugs and medicines, unreplaced blood, and necessary disposable items. Standard hospital admission kits are also payable. A private room used for isolation purposes will only be covered if the Covered Individual is contagious to other patients.

(b) Non-Covered Expenses—

- (1) Telephone, television, radio, barber and beauty services, other personal items and replaced blood;
- (2) Taxes, surcharges or interest charges; and
- (3) Any amount in excess of the Hospital's average Semi-Private Room rate, except as provided in Paragraph (b) above.

12.04 SURGICAL AND MATERNITY EXPENSE BENEFIT

A Covered Individual may receive from the Plan a Surgical and Maternity Expense Benefit.

Bariatric procedures (including, but not limited to, gastric bypass, gastric stapling and intestinal bypass) if both the procedure and the facility are pre-approved by TeamCare.

(a) Covered Expenses—

The Plan shall pay the level of benefits specified under Section 20.01(c) for the covered Physician's charges. All payments shall be based upon the Allowed Amount as established by the Fund.

(b) Non-Covered Expenses—

- (1) Charges for stand-by surgeons and any portion of the surgical expense which exceeds the Allowed Amount; and
- (2) Charges for Bariatric procedures unless both the procedure and the facility are preapproved by TeamCare.

12.05 OUTPATIENT DIAGNOSTIC X-RAY AND LABORATORY EXPENSE BENEFIT

A Covered Individual may receive from the Plan an Outpatient Diagnostic X-ray and Laboratory Expense Benefit as follows:

(a) Conditions—

- (1) Except as otherwise provided in this Plan, the X-ray or laboratory examination must be medically necessary; and
- (2) X-ray and laboratory charges will be paid on the basis of the Allowed Amount and/or usage limitations.

(b) Covered Expenses—

The Plan shall pay the level of benefits for out-patient X-ray and laboratory charges as a Major Medical Expense Benefit and as specified under Sections 20.01(d) and 20.01(p).

(c) Non-Covered Expenses—

- (1) Diagnostic procedures rendered as part of a routine physical examination, except as otherwise provided in this Plan;
- (2) Any portion of the X-ray and laboratory charges that exceed the Allowed Amount and/or usage limitations; and
- (3) Dental X-rays or laboratory work.

12.06 OUTPATIENT ACCIDENTAL BODILY INJURY EXPENSE BENEFIT

A Covered Individual may receive from the Plan an Outpatient Accidental Bodily Injury Expense Benefit as follows:

(a) Conditions—

- (1) The emergency care must be the result of an Accidental Bodily Injury; and
- (2) The Covered Individual must not be confined to the Hospital as a resident patient.

(b) Covered Expenses—

The Plan shall pay the level of benefits specified under Section 20.01(e) for all covered expenses incurred due to any Accidental Bodily Injury on the first (1st) day of treatment only.

(c) Non-Covered Expenses—

No payment shall be made under this benefit for illness.

12.07 PRESCRIPTION DRUG BENEFIT

A Covered Individual may receive from the Plan a Prescription Drug Benefit as follows:

(a) Conditions—

The Plan will pay the level of benefits specified under Section 20.01(f) for covered charges for these Prescription Drugs.

(b) Covered Expenses—

The Plan will provide benefits only for those covered drugs prescribed by a Physician or Dentist, dispensed by a Pharmacist and not available over the counter (except insulin and insulin syringes), including:

- (1) Any medicinal substance which bears the legend: "Caution: Federal Law Prohibits Dispensing Without a Prescription";
- (2) Any medicinal substance which may be dispensed by prescription only according to state law;
- (3) Any medicinal substance which has at least one ingredient that is a federal or state restricted drug in a therapeutic amount; and
- (4) Insulin and syringes.
- (5) Nutritional supplements for Covered Individuals with Phenylketonuria (PKU)

(c) Non-Covered Expenses—

- (1) Therapeutic devices or appliances, hypodermic needles, support garments and other non-medicinal substances;
- (2) Medications supplied to Covered Individuals in a Hospital or other treatment facility (includes take home drugs);
- (3) Drugs or medicines supplied to the Covered Individual by a prescribing Physician or Dentist;
- (4) Cosmetic or beauty aids, dietary supplements, vitamins, medications prescribed for weight-loss;
- (5) Immunizing agents, blood and blood plasma or medication prescribed for parenteral administration;
- (6) Medication for which the cost is recoverable under any Workers' Compensation or Occupational Disease Law or any state or federal agency. Any medication furnished by any other drug or medical service for which no charge is made to the Covered Individual:

- (7) Any drug listed on DEA Schedule I, or labeled: "Caution: Limited by Federal Law to Investigational Use";
- (8) Any drug or medication available over the counter;
- (9) Any drug or medication for enhancing sexual function, including but not limited to Viagra;
- (10) Any drug or medication primarily intended for cosmetic or lifestyle enhancement rather than treatment of an illness or injury; and
- (11) Food and/or specialized nutritional products unless approved by the Fund.

12.08 BEHAVIORAL HEALTH BENEFIT - INPATIENT

A Covered Individual may receive from the Plan a Behavioral Health Benefit - Inpatient as follows:

(a) Covered Expenses—

The Plan will provide benefits for necessary care and treatment including, but not limited to: room at semi-private rate, meals, nursing care, medical supplies and other services as regularly rendered by a qualified Hospital or a licensed Psychiatric Treatment Facility or Alcoholism or Substance Abuse Treatment Facility, as approved by the Fund.

(b) Amount Paid—

The Plan will pay the level of benefits specified under Section 20.01(g), all covered expenses incurred by a Covered Individual in a qualified Hospital or in a licensed Psychiatric Treatment Facility or Alcoholism and Substance Abuse Treatment facility.

- (c) Non-Covered Expenses—
 - (1) Maintenance Care;
 - (2) Treatment not prescribed by a psychiatrist, Physician or Clinical Psychologist;
 - (3) Half-way house type facilities;
 - (4) Legal Services;
 - (5) Recreational, vocational, financial, educational, family or marital counseling;
 - (6) Services rendered by a federal, state or other facility for which the member is not legally required to pay;
 - (7) Detoxification or substance withdrawal programs not rendered by a Hospital, a licensed Psychiatric Treatment Facility or Alcoholism and Substance Abuse Treatment Facility;

- (8) Services rendered by an individual who is not licensed or not registered in the state where services are performed; and
- (9) Telephone, television, radio, barber, beauty services and other personal comfort items.

12.09 BEHAVIORAL HEALTH BENEFIT - OUTPATIENT

A Covered Individual may receive from the Plan a Behavioral Health Benefit - Outpatient as follows:

(a) Covered Expenses—

All services rendered on an outpatient basis under the direction of a psychiatrist, Physician, Clinical Psychologist, or licensed social worker performed for maintenance of a psychiatric condition or chemical abuse program in affiliation with a Hospital, Psychiatric Treatment Facility or Alcoholism or Substance Abuse Treatment Facility.

(b) Amount Paid—

The Plan will pay the level of benefits specified under Section 20.01(h) for all covered outpatient expenses for each Covered Individual.

(c) Non-Covered Expenses—

The Plan will not pay charges for:

- (1) Educational, vocational, financial, family or marital counseling;
- (2) Legal Services;
- (3) Recreational counseling;
- (4) Services rendered by a federal, state or other institution for which the member is not legally required to pay;
- (5) Dues or contributions to a supportive organization or facility; and
- (6) Services rendered by an individual who is not licensed or not registered in the state where services are performed.
- (7) Regressive therapy, megavitamin therapy, nutritionally based therapies for chemical dependency treatment, and non-abstinence based chemical dependency treatment;
- (8) Custodial care, including, but not limited to, treatment not expected to reduce the disability to the extent necessary to enable the patient to function outside a protected, monitored or controlled environment;

- (9) Services and treatment for intellectual disabilities, autism spectrum disorders (ASD) and/or developmental delays except for initial diagnosis. Applied Behavioral Analysis (ABA) may be covered for ASD;
- (10) Treatment for gender dysphoria;
- (11) Treatment for obesity and/or weight reduction;
- (12) Treatment for stammering or stuttering; and
- (13) Treatment for chronic pain except for psychotherapy, biofeedback or hypnotherapy provided in connection with a psychiatric disorder.

12.10 ORGAN TRANSPLANT DONOR BENEFIT

A Covered Individual may receive from the Plan an Organ Transplant Donor Benefit as follows:

- (a) This Plan will provide Coverage for the donor of an organ only in the absence of any other group or individual policy coverage for the donation of an organ, and only if such donation pertains to a procedure which has met the requirements for coverage as defined in Sections 4.02 and 4.17. The donor's medical expenses will be considered part of, and subject to, the provisions of the recipient's Plan. Donors shall be entitled to Basic Benefits, except for the Short-Term Disability Benefit. Major Medical Expense Benefits for donors are payable while the donor is confined in a Hospital for the actual donation of the organ and for ninety (90) days after the end of this Hospital confinement; and
- (b) If both the recipient and the donor are covered under any Fund Plan, each shall be entitled to the maximum benefits provided and outlined under their respective Plans.
- (c) Non-Covered Expenses—
 - (1) Surgical procedures considered not to be Standard Medical Care, Treatment, Services or Supplies; and
 - (2) Charges covered by any Other Plan.

12.11 HEARING AID BENEFIT

A Covered Individual may receive from the Plan a Hearing Aid Benefit as follows:

(a) Covered Expenses—

The Plan will provide benefits for all medically necessary services rendered by an audiologist or a certified hearing aid specialist, if recommended or prescribed by a Physician, including fitting, initial batteries and cost of approved hearing aid correction devices.

(b) Amount Paid—

The Plan will pay the level of benefits under the time frames specified in Section 20.01(j) for the Allowed Amount, up to a maximum of \$1,000 per hearing aid, per ear, incurred by each Covered Individual.

- (c) Non-Covered Expenses—
 - (1) Replacement of lost, missing or stolen appliances;
 - (2) Repair or replacement of broken appliances;
 - (3) Replacement of batteries;
 - (4) Hearing aids purchased without prescription or recommendation by a Physician or without a Waiver approved by the Food and Drug Administration; and
 - (5) Services and supplies for which the Covered Individual is not legally required to pay.

12.12 OUTPATIENT CANCER TREATMENT BENEFIT

A Covered Individual may receive from the Plan an Outpatient Cancer Treatment Benefit as follows:

(a) Covered Expenses—

The Plan will provide benefits for medical necessary outpatient nuclear therapy, radiation therapy, chemotherapy, X-ray and laboratory procedures and Physician visits for the treatment of cancer.

(b) Amount Paid—

The Plan will pay the level of benefits specified under Section 20.01(k) for the Allowed Amount incurred during the treatment period.

- (c) Non-Covered Expenses—
 - (1) Services and supplies for which the Covered Individual is not legally required to pay; and
 - (2) Services and supplies determined to be follow-up or screening services, occurring during a period when active treatment, such as chemotherapy or radiation therapy is not occurring.

12.13 AMBULANCE SERVICE BENEFIT

A Covered Individual may receive from the Plan an Ambulance Service Benefit as follows:

(a) Covered Expenses—

- (1) The Plan will provide benefits for professional licensed ambulance service charges incurred solely for medical treatment and or services required during transport, including licensed Air Ambulance; and
- (2) The Fund will provide benefits for transportation by a Commercial Carrier when it is more economical than a private ambulance service.

(b) Amount Paid—

The Plan will pay the level of benefits specified under Section 20.01(I) for the Allowed Amount for each Covered Individual.

(c) Non-Covered Expenses—

- (1) Transportation in any privately owned vehicle;
- (2) Services and supplies for which the Covered Individual is not legally required to pay;
- (3) Transportation for reason other than receiving required medical treatment;
- (4) Transportation to receive medical treatment which is available at point of origin; and
- (5) When an alternative means of transportation other than an ambulance could be utilized without endangering the Covered Individual's health, whether or not such other transportation is available or is a covered benefit.

12.14 CHIROPRACTIC EXPENSE BENEFIT

A Covered Individual may receive from the Plan a Chiropractic Expense Benefit as follows:

- (a) Conditions and Covered Expenses—
 - (1) The services must be necessitated by illness or injury;
 - (2) The Covered Individual must be age twelve (12) or older;
 - (3) The services can be performed on either an inpatient or an outpatient basis; and
 - (4) All services provided by a Chiropractor, or under his direction, including x-rays, laboratory, therapy, hospitalization and office visits, or any other covered services will be processed solely under this benefit.

(b) Amount Paid—

(1) The Plan shall pay the level of benefits specified under Section 20.01(m) for covered charges up to the calendar year maximum specified under Section 20.01(m). In no event shall payment exceed the scheduled benefit; and

- (2) Each individual charge will be reviewed and only the portion of the charge deemed the Allowed Amount will be considered for payment.
- (c) Non-Covered Expenses—
 - Any portion of the Chiropractic expense which exceeds the scheduled benefit;
 - (2) Any portion of the Chiropractic expense which exceeds the Allowed Amount as established by the Fund; and
 - (3) Any services not necessitated by illness or injury.

12.15 WOMEN'S HEALTH BENEFIT

A Covered Individual may receive from the Plan a Women's Health Benefit as follows:

- (a) Covered Expenses (services must be provided by a Physician participating in a TeamCare preferred provider organization network)—
 - (1) An annual office visit for women eighteen (18) years of age or older;
 - (2) An annual Pap test for women; and
 - (3) An annual mammogram for women forty (40) years of age or older.
- (b) Amount Paid—

The Plan will pay the level of benefits specified under Section 20.01(o) for the Allowed Amount for each Covered Individual.

(c) Non-Covered Expenses—

No payment shall be made for any routine laboratory procedures performed in conjunction with annual examination/pap test/mammogram, except as provided in Section 12.17(a)(2); and

12.16 MAYO CLINIC TREATMENT

The Plan will pay the level of benefits specified under Section 20.01(n). Covered Individuals shall not be entitled to payment for travel, lodging and other non-medical costs associated with obtaining medical services at Mayo Clinic.

12.17 WELLNESS BENEFIT

A Covered Individual may receive from the Plan a Wellness Benefit as follows:

- (a) Covered Expenses—
 - (1) The Plan shall pay 100% of the Allowed Amount for one routine physical examination per calendar year for a Covered Individual by a Physician participating in a TeamCare preferred provider organization network;
 - (2) The Plan shall pay 100% of the Allowed Amount for routine X-ray, laboratory and other diagnostic screening tests, examinations and procedures provided to a Covered Individual in a Hospital or other medical services facility (including a physician's office) to the extent such routine diagnostic services are performed by or under the direction of a Physician participating in a TeamCare preferred provider organization network (diagnostic services that are covered by this subsection include prostate-specific antigen [PSA] tests, bone density tests, colonoscopies, sigmoidoscopies, complete blood count [CBC] tests, basic metabolic profile or panel [BMP] tests and diabetes screening procedures);
 - (3) The Plan shall pay 100% of the Allowed Amount for routine immunizations provided by a Physician participating in a TeamCare preferred provider organization;
 - (4) Evidence-based items or services that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force:
 - (5) Immunizations that have in effect a recommendation from the Advisory Committee on Immunization Practices of the Centers for Disease Control and Prevention:
 - (6) With respect to infants, children and adolescents, evidence-informed preventive care and screenings provided for in the comprehensive guidelines supported by the Health Resources and Services Administration:
 - (7) With respect to women and Women's Preventive Services, such additional preventive care and screenings as provided for in comprehensive guidelines supported by the Health Resources and Services Administration. Preventive services include but not limited to well-woman exams, including pap smears, and prenatal care; and
 - (8) Preventive care Benefits defined under the Health Resources and Services Administration (HRSA) requirement include the cost of renting one breast pump per Pregnancy in conjunction with childbirth. Benefits for breast pumps also include the cost of purchasing one breast pump per Pregnancy in conjunction with childbirth. These Benefits are only available if breast pumps are obtained from an in-network provider or Physician.
- (b) Non-Covered Expenses— The Plan will not cover as a Wellness Benefit, or pay (as a Wellness Benefit), any charges for:

- Any test, examination, procedure, service or product which is not provided by or under the direction of a physician participating in a TeamCare preferred provider organization network;
- (2) Any test, examination, procedure, service or product which is not generally accepted by the medical community as Standard Medical Care, Treatment, Services or Supplies;
- (3) Any amount in excess of the Allowed Amount for the procedure or service provided, as determined by the Fund;
- (4) Any charge which is covered by any Other Plan and/or which is paid by any employer, governmental unit or other entity;
- (5) Any test, examination, procedure, service or product for which the Covered Individual is not required to pay;
- (6) Educational, vocational, recreational, exercise and weight-loss programs, routine hearing tests except as indicated under Section 12.17(a)(6), medications and vitamins that are not covered as Prescription Drugs under Section 12.07, and any test, examination, procedure, service or product that is provided exclusively for the personal and non-medical comfort of a Covered Individual;
- (7) Any test, examination, procedure, service or product that is within the scope (and subject to the exclusions and limitations) of the Women's Health Benefit (Sections 12.15 and 20.01), Dental Benefits (Article XV and Section 20.02) or Vision Benefits (Article XVI and Section 20.03); and
- (8) Any taxes, any surcharges and any charges for late payments, interest, document preparation or missed appointments.

12.18 WAIVER OF DEDUCTIBLE/CO-PAYMENT REQUIREMENT (OFFICE VISITS)

Deductibles shall not apply to the cost of covered Physician office visits by Covered Individuals if the Physician is participating in a TeamCare preferred provider organization network, except for a per visit copayment as specified in Section 20.01(q), which shall be required.

12.19 PAYMENT OF CERTAIN BASIC BENEFITS FOR TREATMENT CONTINUING AFTER TERMINATION OF COVERAGE UNDER THE PLAN

(a) Conditions—

A Covered Individual whose Coverage under this Plan terminates for any reason other than Employer withdrawal, shall continue to be eligible for all Basic Benefits for a period of thirteen (13) weeks after the date the Individual's Coverage terminates, if:

(1) The condition being treated exists on the date the Individual's Coverage under the Plan terminates; and

- (2) The Covered Individual incurs expenses compensable under the Plan on or prior to the date on which the Individual's Coverage under the Plan terminates; and
- (3) The Covered Individual is, on the date his Coverage under the Plan is terminated, so incapacitated by the condition being treated that he is severely restricted from engaging in normal activities.

(b) Covered Expenses—

Only those expenses directly related to the condition being treated will be covered by the extension of Basic Benefits.

(c) Non-Covered Expenses—

If, immediately upon termination of his Coverage under this Plan, a Covered Individual becomes eligible for benefits under the Retiree Plan, he shall not be eligible for benefits under this Section.

12.20 TERMINATION OF THE BASIC BENEFIT EXTENSION

On the date the Covered Individual becomes eligible for insurance coverage under an Other Plan, all benefits provided under Section 12.19 shall cease.

12.21 RETIREE CONTRIBUTION BENEFIT

The Fund shall pay a contribution to the Retiree Plan on behalf of a Covered Participant whose Employer has elected to provide for Retiree benefits. The contribution shall be in the amount determined by the trustees of the Retiree Plan.

ARTICLE XIII. MAJOR MEDICAL EXPENSE BENEFITS

13.01 OUTLINE OF MAJOR MEDICAL BENEFITS

A Covered Individual may receive Major Medical Expense Benefits from the Plan. The Sections of Article XIII describe policies and procedures applicable to all plans offering Major Medical Expense Benefits. However, levels of payment, if any, and/or program limitations specific to the Covered Individual's Plan are determined by referencing Section 20.01(p). Certain benefits, defined in Section 1.26 of this Plan as Eligible Major Medical Expenses, may be modified by the provisions of Article XX. Thus, in cases of conflict between provisions, Article XX shall control.

13.02 MAXIMUM MAJOR MEDICAL EXPENSE BENEFITS

The maximum Major Medical Expense Benefit payable for any Covered Individual shall be the calendar year maximum specified under Section 20.01(p).

13.03 EXTENSION OF THE MAJOR MEDICAL EXPENSE BENEFIT

The Major Medical Expense Benefit may be extended if, on the date his Coverage under the Plan terminates, a Covered Individual:

- (a) Is suffering from a Disability; the treatment of which is compensable under the Plan; and
- (b) Is so incapacitated by that Disability that he is severely restricted from engaging in normal activities; and
- (c) Becomes eligible for Major Medical Expense Benefits relating to that Disability.

The Covered Individual may establish further eligibility for Major Medical Expense Benefits related to that Disability for twenty-four (24) months from the last day of Basic Benefit Coverage (including Short-Term Disability Coverage and the Basic Benefit extension, if applicable), after deduction of the annual Plan Deductible amount specified in Section 20.07 to covered charges per calendar year incurred by the Covered Individual. The benefits are payable at the applicable percentage as outlined in Section 20.01(p) except that the Out-of-Pocket Expense Limit does not apply.

A Participant making Self-Payments subsequent to the end of his Active Coverage period, and who is applying for the Total and Permanent Disability Installment Benefit or the Waiver of Premium Disability Benefit and is subsequently approved for the Total and Permanent Disability Installment Benefit or the Waiver of Premium Disability Benefit, shall not be considered ineligible for the twenty-four (24) month extension.

Only those compensable expenses directly related to the Disability necessitating the Major Medical extension will be covered.

13.04 TERMINATION OF THE MAJOR MEDICAL EXTENSION

The eligibility of a Covered Individual for the extended Major Medical Benefits provided in Section 13.03 shall terminate if:

- (a) The Covered Individual's condition improves so that he is no longer incapacitated by his disability; or
- (b) The Covered Individual becomes eligible for coverage under an Other Plan.

Termination of the Major Medical Extension shall not be considered a COBRA event for purposes of Self-Payments (as described in Article III) to extend coverage.

ARTICLE XIV. LIFE INSURANCE BENEFITS

14.01 **BENEFITS**

The Fund may provide Covered Participants with a Life Insurance Benefit, Dependent Life Insurance Benefit, Total and Permanent Disability Installment Benefit, Waiver of Premium Disability Benefit and Accidental Death and Dismemberment Benefit. The following sections describe policies and procedures applicable to all plans offering the referenced benefits. However, the levels of payment, if any, and/or program limitations specific to the Covered Individual's Plan are determined by referencing Section 20.04.

14.02 LIFE INSURANCE BENEFIT

The amount of the Life Insurance Benefit is specified in Section 20.04.

The Life Insurance Benefit shall not be provided to a beneficiary if such beneficiary is convicted by a court of competent jurisdiction of any willful act which caused or resulted in the death of the Covered Participant. Under such circumstances, the Life Insurance Benefit shall be provided to the next surviving class of beneficiaries as stated in the second paragraph of Section 14.09. In the event of a charge of wrongdoing against the beneficiary in connection with the death of the Covered Participant, the Fund retains the right to investigate the circumstances of such death and to withhold payment of any benefit until termination of any investigation or prosecution involving the beneficiary.

14.03 **DEPENDENT LIFE INSURANCE BENEFIT**

The amount of the Life Insurance Benefit provided for the Spouse or Child of a Covered Participant is specified in Section 20.04.

Coverage of the Spouse or Child begins simultaneously with coverage of the Participant.

A Spouse's or a Child's coverage ends at the times outlined in Article III, Section 3.30.

The Life Insurance Benefit shall not be provided to a beneficiary if such beneficiary is convicted by a court of competent jurisdiction of any willful act which caused or resulted in the death of the Covered Dependent. Under such circumstances, the Life Insurance Benefit shall be provided to the next surviving class of beneficiaries as stated in the second paragraph of Section 14.09. In the event of a charge of wrongdoing against the beneficiary in connection with the death of the Covered Dependent, the Fund retains the right to investigate the circumstances of such death and to withhold payment of any benefit until termination of any investigation or prosecution involving the beneficiary.

14.04 TOTAL AND PERMANENT DISABILITY INSTALLMENT/WAIVER OF PREMIUM DISABILITY BENEFITS

The Fund provides a Total and Permanent Disability Benefit for Covered Participants who became totally and permanently disabled as defined in Section 1.81 of this document, prior to reaching age sixty (60).

If a Covered Participant becomes totally and permanently disabled, as defined in Section 1.81 of this document, then:

- (a) The Total and Permanent Disability Installment Benefit will be provided to the Covered Participant if he has not yet reached age fifty (50); or
- (b) The Waiver of Premium Disability Benefit will be provided to the Covered Participant if he is between age fifty (50) and age fifty-nine (59), inclusive.

14.05 TOTAL AND PERMANENT DISABILITY INSTALLMENT BENEFIT

The Total and Permanent Disability Installment Benefit award specified in Section 20.04 shall consist of payment in lieu of any other benefits specified in Section 20.04. Payment shall be made in 60 equal monthly installments together with an interest allowance at two and one-half percent (2½%) per annum.

The first (1st) monthly installment shall be due and payable on whichever of the following is the latest date:

- (a) Six (6) months after commencement of such Total and Permanent Disability, or
- (b) In the month that a Social Security Award is dated (not the effective date), if the date is on or before the fifteenth (15th) of the month, or
- (c) In the month following the month that a Social Security Award is dated (not the effective date), if the Award date is after the fifteenth (15th) of the month.

Subsequent monthly installments shall be paid on the corresponding day of each month thereafter. Any installments remaining unpaid at the death of the eligible recipient of this award shall be commuted into one (1) sum on the basis of two and one-half percent (2½%) per annum in accordance with Section 14.09.

It shall be the responsibility of the Covered Participant to furnish the Fund with an application for this benefit within three (3) years after the Participant's date of disability.

It shall further be the responsibility of the Covered Participant to furnish the Fund with proof sufficient to determine initial and continuing eligibility for this award.

A Social Security Award of Disability with a disability date on or prior to the Covered Participant's last date of Coverage will be considered as evidence and constitute proof sufficient of such Disability when application for this benefit is considered by the Fund.

14.06 WAIVER OF PREMIUM DISABILITY BENEFIT

The Waiver of Premium Disability Benefit provides an eligible Covered Participant with a Total and Permanent Disability Benefit in the form of Life Insurance, without payment of further contributions on his behalf or by him. At the death of a Covered Participant who received a Waiver of Premium award, the proceeds of his Total and Permanent Disability Benefit shall be paid according to Section 14.09.

It shall be the responsibility of the Covered Participant to furnish the Fund with an application for this benefit within three (3) years after the Participant's date of disability.

It shall further be the responsibility of the Covered Participant to furnish the Fund with proof sufficient to determine initial and continuing eligibility for this award.

A Social Security Award of Disability with a disability date on or prior to the Covered Participant's last date of Coverage will be considered as evidence and constitute proof sufficient of such Disability when application for this benefit is considered by the Fund.

14.07 ELIGIBILITY AND ADMINISTRATION

The Fund shall have the right at any time to require proof of the continuance of a Total and Permanent Disability. If the recipient fails to furnish satisfactory proof or if it appears at any time that the Total and Permanent Disability has terminated, no further payments shall be made and any Fund obligation with respect to the benefit on the life of such person shall cease, except that the remaining benefit on the life of such person, if eligible under this Plan, may be restored, subject to payment of contributions on behalf of the Covered Participant and to the limitation of the amount listed in Section 20.04.

If a Former Covered Participant receives all or part of the Total and Permanent Disability Installment Benefit, and subsequently becomes an Active Employee, that Employee shall again become eligible for the full Life Insurance Benefit after the completion of five (5) years as a Covered Participant.

For purposes of determining eligibility for the Total and Permanent Disability Installment Benefit or the Waiver of Premium Disability Benefit, the age of the Participant at the date of Disability shall control.

14.08 ACCIDENTAL DEATH AND DISMEMBERMENT BENEFIT

The Fund provides a Covered Participant with an Accidental Death and Dismemberment Benefit for loss, as provided below and not hereinafter excluded, sustained while covered under this Plan and occurring within one hundred-twenty (120) days after the date of the Accidental Bodily Injury resulting in the loss, provided that this benefit is not payable if the death or other loss resulting from an Accidental Bodily Injury occurs after the Coverage of the Covered Participant has been terminated. For any one (1) of the losses listed below, the Fund will pay, subject to the provisions hereinafter contained, the portion set opposite such loss, of the Life Insurance Benefit in force on the date of the accident. Only one (1) of the amounts so specified, the largest, will be paid for all injuries resulting from any one (1) accident.

For Loss of:

Life	The Principal Sum
Both Hands or Both Feet or Sight of Both Eyes	The Principal Sum
One Hand and One Foot	The Principal Sum
One Hand and Sight of One Eye	
One Foot and Sight of One Eye	
One Hand or One Foot	½ Principal Sum
Sight of One Eye	

"Loss" shall mean, with regard to hands and feet, dismemberment by severance through or above wrist or ankle joints; with regard to eyes, "loss" means entire and irrecoverable loss of sight.

(a) Exclusions—

The benefit shall not cover any of the following losses:

- (1) Loss resulting from the contracting of disease;
- (2) Loss caused, or contributed to, by bodily or mental infirmity, disease or medical or surgical treatment thereof, or infection (except through an accidental cut or wound), even though the proximate and precipitating cause of the loss is Accidental Bodily Injury;
- (3) Loss caused or contributed to by war or any act of war, whether war is declared or not, or by any act of international armed conflict, or conflict involving the Uniformed Services of any international authority, or while in the Uniformed Services of any country:
- (4) Loss caused or contributed to by any intentional or reckless act and/or omission by the Covered Participant, if he knew or should have known (based upon objective analysis of the actual circumstances) that his death or bodily injury to himself was likely to result from his act and/or omission, including but not limited to the following:
 - (A) his participation in a violation of any law or laws applicable to his conduct;
 - (B) his operation of a motor vehicle while intoxicated in excess of the maximum legal limits established by a law or laws applicable to his conduct;
 - (C) his ingestion, injection or other consumption into his body of any unlawful substance; or of any other unlawful substance; or of any Prescription Drug which was not prescribed for him by a Physician or, if prescribed for him by a Physician, was consumed by him in an amount or amounts exceeding the maximum prescribed dosage;
 - (D) his participation in an altercation in which he was an aggressor; and
 - (E) his participation (regardless of his mental state) in any suicidal act or omission or in any act or omission which he intended to cause or contribute to his death.

In any determination whether or not death or a bodily injury was likely to result from a particular act and/or omission by a Covered Participant, within the meaning of this subsection, it is and shall be irrelevant whether or not he subjectively expected that his death or that bodily injury to himself would result from his act or omission.

- (b) Written proof of any loss under this Section must be furnished to the Fund within three (3) years after the date of such loss.
- (c) The Fund, at its expense, shall have the right and opportunity to have the Covered Participant examined when, and so often as, it may reasonably require during the

- pendency of a claim under this Article, and also the right and opportunity to request an autopsy in the case of death, where it is not forbidden by law.
- (d) Any benefit for loss of life provided under the Accidental Death and Dismemberment Benefit is payable in accordance with Section 14.09. Any benefit for loss other than life is payable to the Covered Participant.

14.09 BENEFICIARY AND MODE OF SETTLEMENT

It is the responsibility of each Covered Participant to supply the Fund with properly executed enrollment forms as required by the Fund, designating the beneficiary of any benefit described in this Article. If a Covered Participant desires to change the designated beneficiary who had been previously designated on the enrollment form, then, it is his responsibility to provide the Fund with a properly executed Designation of Beneficiary form.

If a Covered Participant's marital status is terminated due to a final decree of divorce, ANY beneficiary designation running in favor of the Covered Participant's divorced spouse made by the Covered Participant pursuant to this Article, prior to the final decree of divorce, will be null and void. If, after a final decree of divorce, a Covered Participant fails to supply the Fund with a properly executed enrollment form as required by the Fund, designating the beneficiary of any benefit described in this Article that were designated, prior to the divorce, in favor of the Covered Participant's former [currently divorced or disqualified/terminated] Spouse, such benefits will be payable pursuant to the preference provisions as described below. (Any benefits designated, prior to the final decree of divorce, and running in favor of persons OTHER THAN the former [now divorced] Spouse will not be affected by this paragraph.)

If a Covered Participant fails to execute proper enrollment forms, then, at his death, benefits will be payable to the first surviving class, as follows:

- 1. Covered Participant's Spouse
- 2. Covered Participant's Children, in equal shares
- 3. Covered Participant's Parents, in equal shares
- 4. Covered Participant's Siblings, in equal shares
- 5. Covered Participant's Estate

If a Covered Participant's Spouse or Dependent Child pre-decease the Covered Participant, the Spouse's or Dependent Child's Life Insurance Benefit will be payable only to the Covered Participant.

In the event a Covered Spouse of a Covered Participant dies simultaneously with the Participant, the Spouse's Life Insurance Benefit will be payable to the first surviving class, as follows:

- 1. Spouse's Surviving Children, in equal shares
- 2. Spouse's Parents, in equal shares
- 3. Spouse's Siblings, in equal shares
- 4. Spouse's Estate

The Fund reserves the right to make reasonable regulations as to the number of beneficiaries and method of payment of the Life Insurance Benefit. If the Fund receives an order from a divorce or family court which, in the judgment of the Trustees, purports to require that a benefit be payable otherwise than as provided in this section, the Trustees may withhold paying any benefit until all claimants to the benefit exhaust the administrative appeals process required by the Plan.

14.10 **GRACE PERIOD**

The Fund shall provide a grace period to a Covered Individual of thirty-one (31) days following the termination of a Covered Participant's active coverage to receive the Life Insurance Benefit. The Accidental Death Benefit shall not be payable during the grace period.

ARTICLE XV. DENTAL BENEFITS

15.01 PAYMENT FOR CERTAIN DENTAL TREATMENT AND SERVICES

A Covered Individual may receive from the Plan Dental Benefits to help defray the cost of covered dental procedures. The remaining Sections of Article XV describe policies and procedures applicable to all plans offering the referenced Dental Benefit. However, levels of payment, if any, and/or program limitations specific to the Covered Individual's Plan are determined by referencing Section 20.02.

15.02 COVERED DENTAL PROCEDURES AND MAXIMUM AMOUNT PAYABLE

Payments of Dental Benefits are subject to the following:

- (a) The Plan shall pay the level of benefits specified in Section 20.02 for covered procedures as described by the American Dental Association's current dental terminology (CDT) for a Covered Individual. All payments are subject to Reasonable and Customary charge limits as determined by the Fund;
- (b) The maximum amount payable for any Covered Individual in a calendar year (January 1st to December 31st) is specified in Section 20.02;
- (c) The maximum amount payable for orthodontic treatment for a Dependent Child up to the twenty-sixth (26th) birthday is specified in Section 20.02; and
- (d) If there are alternate treatment procedures available which render effective treatment, the Fund reserves the right to provide benefits for only the least costly of the procedures.

15.03 LIMITATIONS ON PAYMENT OF DENTAL BENEFITS

Dental Benefit payments shall not be made for charges incurred for:

- (a) Any amount over the Reasonable and Customary allowance established by the Fund;
- (b) Treatment by someone other than a Dentist or doctor, except for cleaning and scaling of teeth and application of fluoride treatment and/or sealants by a licensed dental hygienist when such services are rendered under the direct supervision and guidance of the Dentist;
- (c) Services and/or supplies for cosmetic purposes;
- (d) Orthodontic services and/or supplies for a Covered Participant, Spouse or Dependent Child on or after their twenty-sixth (26th) birthday, including orthodontia in conjunction with TMD and/or other medical/dental conditions;
- (e) Services and/or supplies which do not satisfy Section 4.02 or which are not necessary according to those standards professionally endorsed by the general dental community;
- (f) Services and supplies for which the Covered Individual is not legally required to pay;

- (g) Precision attachments, specialized techniques and personalization or characterization of dental prostheses;
- (h) Procedures, restorations and appliances to increase vertical dimension (the distance between the nose and chin);
- (i) Educational programs, such as plaque control, oral hygiene instruction or nutritional counseling;
- (j) Sealants for Participants, Spouses or a Dependent Child past their fourteenth (14th) birthday;
- (k) Sealants more than once in any eighteen (18) month period;
- (I) Replacement of lost, missing or stolen dental/orthodontic appliances;
- (m) Failure to keep a scheduled visit with a Dentist or hygienist;
- (n) Completion of any dental claim forms;
- (o) Local anesthesia, analgesia or nitrous oxide;
- (p) Prescriptions written by the Dentist. However, prescriptions are covered under the Prescription Drug Benefit;
- (q) Dental treatment for a Dependent Child on and after their twenty-sixth (26th) birthday;
- (r) Temporary restorations or sedative fillings on the same day as restorative dentistry;
- (s) In connection with restorative dentistry, bases;
- (t) In connection with inlays, crowns, bridgework, dentures or prosthetic devices:
 - (1) Expenses for replacement made less than three (3) years after a preceding placement or replacement which was covered by this Plan;
 - (2) Expenses for extension of bridges or prosthetic devices previously paid for by this Plan, except for expenses incurred for new extended areas, except as provided in 15.03(oo); and
 - (3) Expenses for any prosthetic appliance unless the appliance is actually inserted.
- (u) Expenses for rebasing of dentures made less than three (3) years after the previous rebasing covered by the Plan;
- (v) Expenses for adjustments, tissue conditioning, relining and/or rebasing, less than six (6) months after insertion of denture(s);
- (w) Expenses for laboratory relining made less than three (3) years after the previous laboratory relining covered by the Plan;

- (x) Expenses for labial veneers/laminate unless due to accident, fracture or birth defect, or within three (3) years of previous labial veneers covered by the Plan;
- (y) Full mouth X-rays more than once in any two (2) year period;
- (z) Bite-wing X-ray(s) more than once in any six (6) month period;
- (aa) Fluoride application more than once in any six (6) month period;
- (bb) An oral examination more than once in any six (6) month period;
- (cc) Prophylaxis more than once in any six (6) month period;
- (dd) Periodontal prophylaxis more than once in any six (6) month period;
- (ee) Fluoride application performed on a Participant or Spouse;
- (ff) Periodontal scaling and/or root planing more than once in any one (1) year period;
- (gg) Crowns without sufficient breakdown or sufficient decay;
- (hh) Crowns and/or bridgework without sufficient bone support;
- (ii) Crowns and/or bridgework supported by implants;
- (jj) Expenses for multiple periodontal procedures performed on the same day;
- (kk) Expenses for periodontal procedures performed on Dependent Children (will be individually reviewed for possible payment);
- (II) Expenses for space maintainer for Participant or Spouse;
- (mm) Expenses for home medicaments;
- (nn) Any procedure not completed;
- (oo) Permanent crowns and/or bridgework on deciduous (baby) teeth;
- (pp) Root canal therapy, apicoectomy, root resection and hemisection more than once in a lifetime per tooth (root); and
- (qq) General anesthesia; unless administered in conjunction with oral surgery (impacted or surgical extractions), periodontal surgery, fracture, dislocations, apicoectomies, or three
 (3) or more simple extractions rendered on the same date of service.

15.04 ORTHODONTIA BENEFIT—DEPENDENT CHILDREN ONLY

A Covered Dependent Child shall receive from the Plan an orthodontia benefit as follows:

(a) Covered Expenses—

The Plan will provide benefits, as specified in Section 20.02, for orthodontic services rendered by a Dentist or Orthodontist, providing the Participant is covered, to correct the following dental problems:

- (1) The existence of extreme bucco-lingual version of the teeth, either unilateral or bilateral:
- (2) A protrusion of the maxillary teeth of four (4) or more millimeters;
- (3) An open bite of four (4) or more millimeters;
- (4) A protrusive or retrusive relation of the maxillary or mandibular arch of at least one (1) cusp; or
- (5) An arch-length discrepancy of four (4) or more millimeters.

(b) Amount Paid—

The Plan will pay the level of benefits specified in Section 20.02 for covered charges up to the maximum orthodontic benefit per person, per lifetime as specified in Section 20.02.

- (c) Non-Covered Expenses—
 - (1) Orthodontic appliances and/or related services rendered on Dependent Children on and after their twenty-sixth (26th) birthday;
 - (2) Services and/or supplies for orthodontia on any Participant or Spouse;
 - (3) Any covered charges in excess of the lifetime maximum benefit;
 - (4) Any protrusion, retrusion or open-bite which does not meet the requirements outlined in (a) above; and
 - (5) Expenses for incomplete procedures or appliances that are not inserted and/or related services.

15.05 EXTENSION OF DENTAL BENEFITS

If, on the date his Coverage under the Plan terminates, a Covered Individual has undergone any of the following dental procedures which require extended treatment, Dental Benefits will be extended to provide coverage for these procedures, providing the work is completed within one (1) year:

(a) Dentures, full or partial, if the impression was taken while a Covered Individual; or

- (b) Fixed bridgework, gold restorations and crowns, if tooth or teeth were prepared while a Covered Individual; or
- (c) Root canal therapy, if the tooth or teeth were opened for treatment while a Covered Individual.

15.06 COVERAGE FOR SPECIFIC DENTAL PROCEDURES

The American Dental Association procedure codes D7340 through D7997 will be considered for payment under the Covered Individual's Basic and/or Major Medical Benefits (refer to Articles XII, XIII and Section 20.01).

15.07 COVERAGE OF PEDIATRIC ORAL CARE

Pediatric oral care shall not be subject to the annual dollar limits contained in Subsections (b), (c), and (d) of Section 20.02, but all other annual dollar limits shall apply to pediatric oral care. Further, claims for benefits shall remain subject to all other limitations and co-payment requirements contained in the Plan including the limitations in Section 15.03 and the co-payment requirements in Section 20.02.

ARTICLE XVI. VISION BENEFITS

16.01 **COVERED VISION EXPENSES**

- (a) The Plan may provide payments to help defray the cost of eye examinations and materials for Covered Individuals. The Sections of Article XVI describe policies and procedures applicable to all Plans offering Vision Benefits. However, levels of payment, if any, and/or program limitations specific to the Covered Individual's Plan are determined by referencing Section 20.03.
- (b) Only procedures intended to improve the otherwise healthy eye by glasses shall be covered under Vision Benefits. However, eye procedures requiring hospitalization, surgery, X-ray or laboratory work may be covered under Basic Benefits and/or Major Medical Expense Benefits.
- (c) Only vision procedures performed by an optician, optometrist or ophthalmologist shall be covered.

16.02 COVERED VISION PROCEDURES AND MAXIMUM AMOUNT PAYABLE

The Plan shall pay covered expenses incurred up to the amount listed for the covered procedure or item under Section 20.03.

16.03 LIMITATION ON PAYMENT FOR VISION BENEFITS

- (a) No Vision Benefit payment shall be made in any one twelve (12) month period for more than:
 - (1) One (1) complete examination;
 - (2) One (1) pair of lenses;
 - (3) One (1) set of frames, or repair of frames;
 - (4) One (1) pair of contact lenses; or
 - (5) Either one (1) pair of lenses and frames or one (1) pair of contact lenses.
- (b) No Vision Benefit payment shall be made for:
 - Vision care services or supplies received from a medical department maintained by the Participant's Employer, a mutual benefit association, a labor union, trustee or similar group;
 - (2) Vision care services or supplies furnished by, or at the direction of, the United States government or any agency thereof;
 - (3) Medical or surgical treatment of the eye;

- (4) Sunglasses, plain or prescription, or safety glasses;
- (5) Orthoptics or vision training; and
- (6) Replacement due to loss or theft.
- (c) Any amount over the Reasonable and Customary allowance established by the Fund.
- (d) Services and/or supplies for cosmetic purposes.

16.04 COVERAGE OF PEDIATRIC VISION CARE

Pediatric vision care shall not be subject to the annual dollar limits contained in Subsection (a) of Section 20.03 for an examination for glasses, but all other annual dollar limits shall apply to pediatric vision care including examinations for contact lenses. Further, claims for benefits shall remain subject to all other limitations and co-payment requirements contained in the Plan including the limitations set forth in Section 16.03 and the co-payment requirements in Section 20.03.

ARTICLE XVII. OUT-OF-POCKET EXPENSE LIMIT

- 17.01 OUT-OF-POCKET EXPENSE IS THAT PORTION OF ELIGIBLE EXPENSES INCURRED THAT IS THE COVERED INDIVIDUAL'S RESPONSIBILITY AFTER THE FUND HAS PAID ITS REQUIRED BENEFITS. THE FUND PROVIDES COVERED INDIVIDUALS WITH OUT-OF-POCKET EXPENSE LIMITS. AMOUNTS OF THE OUT-OF-POCKET EXPENSE LIMIT (TO THE EXTENT THOSE LIMITS ARE APPLICABLE IN SPECIFIC CIRCUMSTANCES) ARE SPECIFIED IN SECTION 20.05.
 - (a) Expenses that may be applied to the Out-of-Pocket Expense Limit include:
 - The balance of eligible Hospital expenses after the Fund has paid its required Hospital Expense Benefits;
 - (2) The balance of eligible surgical and obstetrical expenses after the Fund has paid its required Surgical and Obstetrical Expense Benefits;
 - (3) The balance of eligible outpatient diagnostic x-ray and laboratory expenses after the Fund has paid its required Outpatient Diagnostic X-ray and Laboratory Expense Benefits:
 - (4) The balance of eligible Accidental Bodily Injury expenses after the Fund has paid its required Outpatient Accidental Bodily Injury Expense Benefits;
 - (5) The balance of any psychiatric, alcoholism and substance abuse treatment expenses after the Fund has paid its required Behavorial Health-Inpatient and/or Behavioral Health-Outpatient Benefits;
 - (6) The balance of eligible organ transplant donor expenses after the Fund has paid its required Organ Transplant Donor Benefits;
 - (7) The balance of eligible outpatient cancer treatment expenses after the Fund has paid its required Outpatient Cancer Treatment Benefits;
 - (8) The balance of eligible ambulance service expenses after the Fund has paid its required Ambulance Service Benefits;
 - (9) The balance of eligible women's health expenses that are within the scope of Section 12.15 after the Fund has paid its required Women's Health Benefits;
 - (10) The balance of eligible Mayo Clinic expenses after the Fund has paid its required Mayo Clinic Treatment Benefits;
 - (11) The balance of eligible Major Medical Expenses after the Fund has paid its required Major Medical Expense Benefits;
 - (12) The balance of any chiropractic expenses after the Fund has paid its required Chiropractic Expense Benefits;
 - (13) Any required TeamCare office visit co-payments; and

- (14) All deductibles.
- (b) Expenses that may not be applied to the Out-of-Pocket Expense Limit include:
 - (1) Any charge or expense which exceeds the Allowed Amount established by the Fund;
 - (2) Any charge or expense which exceeds any applicable maximum dollar amount payable by the Fund as stated in the Plan;
 - (3) The balance of any Prescription Drug expenses after the Fund has paid its required Prescription Drug Benefits;
 - (4) The balance of any hearing aid expenses after the Fund has paid its required Hearing Aid Benefits;
 - (5) The balance of any dental expenses after the Fund has paid its required Dental Benefits:
 - (6) The balance of any vision expenses after the Fund has paid its required Vision Benefits; and
 - (7) All other expenses which are not payable by the Fund because of Coverage exclusions and/or limitations, other than eligible expenses that are specified in Section 17.01(a).
 - (8) Eligible Major Medical Expenses incurred by Covered Individuals receiving benefits pursuant to Section 13.03.
- (c) After the annual Out-of-Pocket Expense Limit applicable to a Covered Individual, as specified in Section 20.05, has been reached, the Fund is obligated to pay the full Allowed Amount for all expenses described in Section 17.01(a) that are incurred by the Covered Individual during the remainder of that calendar year, provided that this obligation of the Fund does not apply to any expenses described in Section 17.01(b).
- 17.02 PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE IS THAT PORTION OF ELIGIBLE PRESCRIPTIONS DRUG EXPENSES INCURRED THAT IS THE COVERED INDIVIDUAL'S RESPONSIBILITY AFTER THE FUND HAS PAID ITS REQUIRED BENEFITS. THE FUND PROVIDES COVERED INDIVIDUALS WITH PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE LIMITS. AMOUNTS OF THE PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE LIMIT (TO THE EXTENT THOSE LIMITS ARE APPLICABLE IN SPECIFIC CIRCUMSTANCES) ARE SPECIFIED IN SECTION 20.08.
 - (a) Expenses that may be applied to the Prescription Drug Out-of-Pocket Expense Limit include the balance of eligible Prescription Drug expenses after the Fund has paid its required Prescription Drug Benefits.
 - (b) Expenses that may not be applied to the Prescription Drug Out-of-Pocket Expense Limit include all other expenses which are not payable by the Fund because of Coverage

- exclusions and/or limitations, other than eligible expenses that are specified in Section 17.02(a).
- (c) After the annual Prescription Drug Out-of-Pocket Expense Limit applicable to a Covered Individual, as specified in Section 20.08, has been reached, the Fund is obligated to pay the full Reasonable and Customary allowance for all expenses described in Section 17.02(a) that are incurred by the Covered Individual during the remainder of that calendar year, provided that this obligation of the Fund does not apply to any expenses described in Section 17.02(b).

17.03 MAXIMUM OUT-OF-POCKET LIMIT ON COST SHARING

The annual maximum out-of-pocket payments by: (1) a Covered Individual and (2) any group consisting of one Covered Participant and all related Covered Dependents shall not exceed the applicable annual limits on cost sharing established by 45 C.F.R. §156.130 as adjusted annually thereafter by the Department of Health and Human Services.

Any and all expenditures required by or on behalf of a Covered Individual with respect to essential health benefits (as established by the essential health benefits benchmark plan adopted by Illinois pursuant to 45 CFR §156.100) shall be applied against the maximum annual out-of-pocket limit including deductibles, coinsurance, copayments, or similar charges, but excluding contributions, self-payments, balance billing amounts for non-network providers, and spending for non-covered services.

ARTICLE XVIII. PLAN BENEFIT LIMIT

18.01 THE TERM PLAN BENEFIT LIMIT IS DEFINED AS THE MAXIMUM PAYABLE BY THE PLAN IN A GIVEN CALENDAR YEAR UNDER ANY OR ALL APPLICABLE PLAN BENEFITS NOT TAKING INTO ACCOUNT ANY RETIREE CONTRIBUTION BENEFIT. THE PLAN MAY INCLUDE SUCH PLAN BENEFIT LIMIT AS DESCRIBED. THE AMOUNT OF THE PLAN BENEFIT LIMIT, IF ANY, IS DETERMINED BY REFERENCING SECTION 20.06.

ARTICLE XIX. EFFECTIVE DATE OF PLAN SPECIFIC PROVISIONS

19.01 THE TERMS AND PROVISIONS OF THIS PUBLISHED EDITION OF THIS PLAN DOCUMENT (COVERING ALL PLANS) ARE APPLICABLE TO THE COVERAGE OF EVERY COVERED INDIVIDUAL WHO HAS BEEN, IS OR HEREAFTER BECOMES ENTITLED TO COVERAGE BY THE PLAN AS OF ANY DATE ON OR AFTER JANUARY 1, 2022, PROVIDED THAT THE TERMS, PROVISIONS, LIMITATIONS AND EXCLUSIONS OF COVERAGE AS OF ANY DATE PRIOR TO JANUARY 1, 2022, ARE GOVERNED BY THE EARLIER EDITION OF THIS PLAN DOCUMENT THAT WAS IN EFFECT ON THAT DATE (INCLUDING ALL PLAN AMENDMENTS OF THAT EDITION ADOPTED AND IN EFFECT ON THAT DATE), AND PROVIDED FURTHER THAT THE TERMS AND PROVISIONS OF THIS EDITION ALSO INCLUDE ALL PLAN AMENDMENTS ADOPTED AND IN EFFECT AFTER PUBLICATION OF THIS EDITION (EVEN IF THEY ARE YET TO BE INCORPORATED IN THIS EDITION).

ARTICLE XX. SCHEDULE OF BENEFITS

20.01 BASIC BENEFITS AND MAJOR MEDICAL EXPENSE BENEFITS

Subject to TeamCare limitations in Section 4.20, the Plan provides accident and health benefits, in the form of Basic Benefits (as set forth in detail in Article XII) and Major Medical Expense Benefits (as set forth in detail in Article XIII) so as to provide comprehensive benefits for Covered Individuals for illness, injury or pregnancy. Actual benefits provided under this plan may be different than shown based upon the specific plan of benefits selected by the Employer. A Schedule of the Basic and Major Medical Benefits follows:

BENEFIT TYPE		PLAN	SCHEDULE OF BENEFITS		
(a) Short-Term Disability Benefit (Participant Only)	The following Plans include Short-Term Disability Coverage for Cove Participants and Covered Dependents as long as Covered Participant eligible for Short-Term Disability Benefit:				
	(1)	A5	\$425/week (or \$60.71 per day) for weeks 1 through 26 and \$320/week (or \$45.71 per day) for the remaining 26 weeks for maximum of 52 weeks.		
	(2)	2N, AB, CK, N6, NN	\$300/week (or \$42.86 per day) for the first 10 weeks and \$350/week (or \$50 per day) for the remaining 16 weeks for maximum of 26 weeks.		
	(3)	AN, WK	\$450/week (or \$64.29 per day) for maximum of 26 weeks.		
	(4)	M5	\$400/week (or \$57.14 per day) for maximum of 26 weeks.		
	(5)	СМ	\$200/week (or \$28.57 per day) for maximum of 26 weeks.		
	(6)	5D	\$100/week (or \$14.29 per day) for maximum of 26 weeks.		
	There	efore, no continuation of the continuation of	not include Short-Term Disability Coverage. on of family coverage while collecting Short-Term s Employer Contributions continue or Self-Payments		
	(7)	58	\$300/week (or \$42.86 per day) for maximum of 26 weeks.		
	(8)	N8, NM	\$150/week (or \$21.43 per day) for maximum of 26 weeks.		

(a) Short-Term Disability Benefit (Participant Only) (continued)	(9)	B5, BN, N1	\$100/week (or \$14.29 per day) for maximum of 20 weeks.
		ollowing Plans <mark>do n</mark> de Short-Term Dis	ot include Short-Term Disability payments but do ability Coverage:
			, ,
	(10)	AK	Medical coverage only.
	The fo	ollowing Plans do n	ot include Short-Term Disability Benefits:
	(11)	1N, 2M, 4M, 5N, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Short-Term Disability Benefit not applicable to these plans.

(b) Hospital Expense Benefit	After Plan Deductible, the percentage of covered charges for the number of days, as indicated below, at average semi-private room rate (coronary care unit, intensive care unit, burn unit or isolation room if medically required). If the Hospital does not have an average semi-private room rate, the amount payable will be the average rate charged by Hospitals in the area for semi-private rooms, as determined by the Fund.			
	(1)	6N, 8N, N3, N6, N8, R3	100% of covered charges for an unlimited number of days at average semi-private room rate.	
	(2)	4M, NM, NN	90% of covered charges for an unlimited number of days. 100% after Out-of-Pocket Expense Limit is met.	
	(3)	1N, 2M, 2N, AB, AN, BA, BN, N1	80% of covered charges for an unlimited number of days. 100% after the Medical Out-of-Pocket Expense Limit is met.	
	(4)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	75% of covered charges for an unlimited number of days. 100% after the Medical Out-of-Pocket Expense Limit is met.	
	(5)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.	

(c) Surgical and Obstetrical Expense Benefit	After I below.	Plan Deductible,	the	percentage	of	the	Allowed	Amount,	as	indicated
	(1)	6N, N6		100%						

(c) Surgical and Obstetrical Expense Benefit (continued)	(2)	4M, 8N, N3, N8, NM, NN, R3	90% (100% after the Out-of-Pocket Expense Limit is met).
(Softmass)	(3)	1N, 2M, 2N, AB, AN, BA, BN, N1	80% (100% after the Out-of-Pocket Expense Limit is met).
	(4)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	75% (100% after the Out-of-Pocket Expense Limit is met).
	(5)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

(d) Outpatient Diagnostic X-ray and Laboratory Expense Benefit	(1)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 6N, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N3, N5, N6, N8, NM, NN, R3, RB	See Section 20.01(p).
	(2)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

(e) Outpatient Accidental Bodily Injury Expense Benefit			e percentage of the Allowed Amount indicated below eatment subject to TeamCare co-pay requirements.
·	(1)	6N, 8N, N3, N6, N8, R3	100%
	(2)	4M, NM, NN	90% (100% after the Out-of-Pocket Expense Limit is met).
	(3)	1N, 2M, 2N, AB, AN, BA, BN, N1	80% (100% after the Out-of-Pocket Expense Limit is met).
	(4)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	75% (100% after the Out-of-Pocket Expense Limit is met).
	(5)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

(f) Prescription Drug	Team	Care RX mail order	program:
Benefit	(1)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 6N, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N3, N5, N6, N8, NM, NN, R3, RB	The Plan pays 80% of covered charges (20% copayment), provided that the maximum co-payment is \$200 for each filled prescription purchased through the TeamCare RX program.
	Retai	pharmacy (TeamC	are and non-TeamCare pharmacies):
	(2)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 6N, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N3, N5, N6, N8, NM, NN, R3, RB	Except for non-exempt maintenance medications (described below in [3]) the Plan pays 75% of covered charges (25% co-payment) and, for non-exempt maintenance medications, the Plan pays 50% of covered charges (50% co-payment) after a two-fill transition period in which the Plan pays 75% of covered charges (25% co-payment), provided that for each filled prescription purchased from a TeamCare RX retail pharmacy (other than a non-exempt maintenance medication purchased after the above-referenced two-fill transition period) the maximum co-payment is \$200.
	(3)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 6N, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N3, N5, N6, N8, NM, NN, R3, RB	A maintenance medication is any prescription drug taken by a Covered Individual over a period exceeding sixty (60) days, other than a drug exempt by the Plan from this classification (exempt drugs include injectable drugs, specialty medications and antidepressants as determined by the Plan).
	(4)		If a generic drug equivalent is available to fill a prescription, the Covered Individual must choose the generic drug or pay (in addition to the copayment) the difference in cost between the generic drug and the brand name drug (if the brand name drug is chosen rather than the available generic drug, the above-stated \$200 maximum [for each filled prescription purchased through the TeamCare RX program] is inapplicable and does not limit the amount payable by the Covered Individual). The Out-of-Pocket Expense Limit does not apply.
	(5)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

(g) Behavioral Health Benefit - Inpatient	(1)	6N, 8N, N3, N6, N8, R3	After Plan Deductible, 100% of covered charges for an unlimited number of days at average semi-private room rate.
	(2)	4M, NM, NN	After Plan Deductible, 90% of covered charges for for an unlimited number of days. 100% after Out-of-Pocket Expense Limit is met.
	(3)	1N, 2M, 2N, AB, AN, BA, BN, N1	After Plan Deductible, 80% of covered charges for for an unlimited number of days. 100% after Out-of-Pocket Expense Limit is met.
	(4)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	After Plan Deductible, 75% of covered charges for for an unlimited number of days. 100% after Out-of-Pocket Expense Limit is met.
	(5)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
(h) Behavioral Health Benefit - Outpatient	(1)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 6N, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N3, N5, N6, N8, NM, NN, R3, RB	Paid pursuant to Sections 20.01(p) or 20.01(q) as appropriate.
	(2)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
(i) Organ Transplant Donor Benefit	(1)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 6N, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N3, N5, N6, N8, NM, NN, R3, RB	After Plan Deductible, the Basic and Major Medical Expense Benefits as outlined in Sections 1.26, 12.10 and 20.01(p).
	(2)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

(j) Hearing Aid Benefit	(1)	6N, N3, N6, R3	After Plan Deductible, 100% of the Allowed Amount, up to a maximum of \$1,000 per hearing aid, per ear, incurred by each Covered Individual once in every thirty-six (36) month period.
	(2)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
	(3)	1N, 2M, 2N, 4M, 5D, 5N, 5S, 8N, A5, AB, AN, B5, BA, BN, CM, M5, N1, N5, N8, NM, NN, RB	Hearing Aid Benefit not applicable to these plans.

(k) Outpatient Cancer Treatment	(1)	6N, N6	After Plan Deductible, 100% of the Allowed Amount during the active treatment period and subject to TeamCare co-pay requirements.
	(2)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
	(3)	All other Plans	See Section 20.01(p).

(I) Ambulance Service	(1)	6N, N6	After Plan Deductible, 100% of the Allowed
Benefit			Amount.
	(2)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
	(3)	All other Plans	See Section 20.01(p).

(m) Chiropractic Expense Benefit	After Plan Deductible, the percentage (indicated below) of the Allowed Amount up to the maximum amount (indicated below) per person, per calendar year for Covered Individuals age twelve (12) and older.		
	(1)	6N, N6	80% up to \$1,000 per person, per calendar year.
	(2)	4M, 8N, N3, N8, NM, NN, R3	70% up to maximum of \$800 per person, per calendar year.

(m) Chiropractic Expense Benefit (continued)	(3)	1N, 2M, 2N, 5D, 5N, 5S, A5, AB, AN, B5, BA, BN, CM, M5, N1, RB	50% up to maximum of \$500 per person, per calendar year.
	(4)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
(n) Mayo Clinic Treatment	(1)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
	(2)	All other Plans	See Section 20.01(p).
	(4)	TALK BLK OLK BLK	
(o) Women's Health Benefit	(1)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
	(2)	All other Plans	See Section 20.01(p).
/> NA - ! NA !	1		halan) of Elimina Marian Mardinal Engage
(p) Major Medical Expense Benefit	The percentage (indicated below) of Eligible Major Medical Expenses as defined in Section 1.26 of this document.		
	(1)	4M, NM, NN	90% (100% after Out-of-Pocket Expense Limit is met).
	(2)	1N, 2M, 2N, 6N, 8N, AB, AN, BA, BN, N1, N3, N6, N8, R3	80% (100% after Out-of-Pocket Expense Limit is met).
	(3)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	75% (100% after Out-of-Pocket Expense Limit is met).
	(4)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.
	ı		
(q) TeamCare Office Visit Co-Payment	Indivi	•	below required for Physician office visits by Covered n is participating in a TeamCare preferred provider

(q) TeamCare Office Visit Co-Payment (continued)	(1)	1N, 2M, 2N, 4M, 6N, 8N, AB, AN, BA, BN, N1, N3, N6, N8, NM, NN, R3	\$20 Co-Payment per in-network office visit.
	(2)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	\$30 Co-Payment per in-network office visit.
	(3)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

20.02 **DENTAL BENEFITS**

The Plan provides Dental Benefits, as set forth in Article XV, for Covered Individuals so as to defray the cost of certain dental procedures. Pediatric oral care shall not be subject to the annual dollar limits contained in Subsections (b), (c), and (d) of Section 20.02, but all other annual dollar limits shall apply to pediatric oral care. Further, claims for benefits shall remain subject to all other limitations and co-payment requirements contained in the Plan including the limitations in Section 15.03 and the co-payment requirements in Section 20.02. A schedule of Dental Benefits by Plan follows:

BENEFIT TYPE		PLAN	SCHEDULE OF BENEFITS
(a) Orthodontic	(1)	AB, AK, AN, CK, N6, WK	100% of the Reasonable and Customary charges for the procedures incurred by a Dependent Child up to the twenty-sixth (26th) birthday up to a \$2,500 lifetime maximum.
	(2)	2M, 2N, BN, N1, N8, NM, NN	50% of the Reasonable and Customary charges for the procedures incurred by a Dependent Child up to the twenty-sixth (26th) birthday up to a \$2,500 lifetime maximum.
	(3)	1N, 4M, 5D, 5N, 5S, 6N, 8N, A5, B5, BA, BK, CM, DK, M5, N3, N5, R3, RB	Benefit is not available under these plans.

(b) Crowns and	(1)	AB, AK, AN, CK,	80% of the Reasonable and Customary charges
Bridgework	(.)	N6, WK	subject to a maximum benefit per person per calendar year of \$1,500 consisting of any combination of payments for covered services as defined in Article XV except for pediatric oral care.
			defined in Article AV except for pediatric oral care.

(b) Crowns and Bridgework (continued)	(2)	BN, CM, M5, N1,	70% of the Reasonable and Customary charges subject to a maximum benefit per person per calendar year of \$1,500 consisting of any combination of payments for covered services as defined in Article XV except for pediatric oral care.
	(3)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.

(c) Preventive Services	for cli codes below	nical oral evaluation, subject to a maxir	d below) of the Reasonable and Customary charges ons and preventive services as described by ADA num benefit per person per calendar year (indicated combination of payments for covered services as
	(1)	2M, 2N, AB, AK, AN, BN, CK, N1, N6, N8, NM, NN, WK	100% of the Reasonable and Customary charges subject to a maximum benefit per person per calendar year of \$2,500 except for pediatric oral care.
	(2)	A5, B5, CM, M5, N5, RB	100% of the Reasonable and Customary charges subject to a maximum benefit per person per calendar year of \$1,500 except for pediatric oral care.
	(3)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.

(d) All other ADA Codes	The percentage (indicated below) of the Reasonable and Customary charge subject to a maximum benefit per person per calendar year (indicated below consisting of any combination of payments for covered services as defined Article XV.			
	(1)	AB, AN, AK, CK, N6, WK	100% of the Reasonable and Customary charges subject to a maximum benefit per person per calendar year of \$2,500 except for pediatric oral care.	
	(2)	2M, 2N, BN, N1, N8, NM, NN	85% of the Reasonable and Customary charges subject to a maximum benefit per person per calendar year of \$2,500 except for pediatric oral care.	

(d) All other ADA Codes (continued)	(3)	A5, B5, CM, M5, N5, RB	85% of the Reasonable and Customary charges subject to a maximum benefit per person per calendar year of \$1,500 except for pediatric oral care.
	(4)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.

20.03 VISION BENEFITS

The Plan provides Vision Benefits, as set forth in Article XVI, so as to defray the cost of eye examinations and materials for Covered Individuals. Pediatric vision care shall not be subject to the annual dollar limits contained in Subsection (a) of Section 20.03 for an examination for glasses, but all other annual dollar limits shall apply to pediatric vision care including examinations for contact lenses. Further, claims for benefits shall remain subject to all other limitations and co-payment requirements contained in the Plan including the limitations set forth in Section 16.03 and the co-payment requirements in Section 20.03. A schedule of Vision Benefits by Plan follows:

BENEFIT TYPE		PLAN	SCHEDULE OF BENEFITS
(a) Examination	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	\$50.00 except for pediatric vision care.
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans except for pediatric vision care.

(b) Frames	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	\$75.00
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.

(c) Lenses (Per Pair) Single Vision	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	\$50.00
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.
(d) Lenses (Per Pair) Bifocal	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	\$50.00
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.
(e) Lenses (Per Pair) Tri-Focal	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	\$50.00
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.
(f) Lenses (Per Pair) Lenticular	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	\$60.00
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.

(g) Lenses (Per Pair) Contacts	(1)	2M, 2N, A5, AB, AK, AN, B5, BN, CK, CM, M5, N1, N5, N6, N8, NM, NN, RB, WK	
	(2)	1N, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, R3	Benefit is not available under these plans.

20.04 LIFE INSURANCE BENEFIT AND ACCIDENTAL DEATH AND DISMEMBERMENT BENEFIT

The Plan provides:

- Life Insurance Benefits for Covered Participants and Covered Dependents;
- · Accidental Death and Dismemberment Benefit for Covered Participants; and

Details of the administration of these benefits are set forth in Article XIV. A schedule of these benefits by Plan follows:

BENEFIT TYPE		PLAN	SCHEDULE OF BENEFITS
(a) Participant Life	(1)	2N, AB, AK, AN, CK, N6, WK	\$40,000
	(2)	N8, NM, NN	\$25,000
	(3)	A5, B5, BN, CM, M5, N1	\$20,000
	(4)	1N, 2M, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Benefit is not available under these plans.

(b) Participant	(1)	2N, AB, AK, AN,	\$40,000
Accidental Death		CK, N6, WK	
	(2)	N8, NM, NN	\$25,000
	(3)	A5, B5, BN, CM,	\$20,000
		MS, N1	

(b) Participant Accidental Death (continued)	(4)	1N, 2M, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Benefit is not available under these plans.
(c) Participant Accidental Dismemberment	(1)	2N, AB, AK, AN, CK, N6, WK	\$40,000
	(2)	N8, NM, NN	\$25,000
	(3)	A5, B5, BN, CM, M5, N1	\$20,000
	(4)	1N, 2M, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Benefit is not available under these plans.
(d) Participant Total and Permanent Disability Installment Benefit	(1)	2N, AB, AK, AN, CK, N6, N8, NM, NN, WK	\$16,000
	(2)	A5, B5, BN, CM, M5, N1	\$11,000
	(3)	1N, 2M, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Benefit is not available under these plans.
(e) Spouse Life Insurance	(1)	2N, AB, AK, AN, CK, N6, WK	\$4,000 payable on Covered Dependants.
	(2)	N8, NM, NN	\$3,000 payable on Covered Dependents.
	(3)	A5, B5, BN, CM, M5, N1	\$2,000 payable on Covered Dependents.
	(4)	1N, 2M, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Benefit is not available under these plans.

(f) Child Life Insurance	(1)	2N, N6, AB, AK, AN, CK, WK	\$2,000 payable on Covered Dependents.
	(2)	N8, NM, NN	\$1,500 payable on Covered Dependents.
	(3)	A5, B5, BN, CM, M5, N1	\$750 payable on Covered Dependents.
	(4)	1N, 2M, 4M, 5D, 5N, 5S, 6N, 8N, BA, BK, DK, N3, N5, R3, RB	Benefit is not available under these plans.

20.05 MEDICAL OUT-OF-POCKET-EXPENSE LIMIT

The Medical Out-of-Pocket Expense Limit, is indicated below. Charges relating to non-covered services, dental, chiropractic and vision services, and prescription drugs do not apply towards the Medical Out-of-Pocket Expense Limit (see Section 17.01 for a full statement of these exclusions).

	PLAN	SCHEDULE OF BENEFITS
(a)	4M, 6N, 8N, N3, N6, N8, NM, NN, R3	\$1,000 per Covered Individual per calendar year, limited to a maximum of \$2,000 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(b)	N1, 1N	\$1,500 per Covered Individual per calendar year, limited to a maximum of \$3,000 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(c)	2M, 2N, AB, AN, BA, BN	\$2,500 per Covered Individual per calendar year, limited to a maximum of \$5,000 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(d)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	\$4,000 per Covered Individual per calendar year, limited to a maximum of \$8,000 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(e)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

20.06 PLAN BENEFIT LIMIT

PLAN	SCHEDULE OF BENEFITS
All Plans	No limit per Covered Individual per calendar year.

20.07 PLAN DEDUCTIBLE

	PLAN	SCHEDULE OF BENEFITS
(a)	1N, N1	\$100 per Covered Individual per calendar year, limited to a maximum \$300 per calendar year in covered charges incurred by a group consisting of one Covered Participant and all related Covered Dependents.
(b)	6N, 8N, N3, N8, N6, R3	\$200 per Covered Individual per calendar year, limited to a maximum \$400 per calendar year in covered charges incurred by a group consisting of one Covered Participant and all related Covered Dependents.
(c)	2M, 2N, AB, AN, BA, BN	\$200 per Covered Individual per calendar year, limited to a maximum \$500 per calendar year in covered charges incurred by a group consisting of one Covered Participant and all related Covered Dependents.
(d)	4M, NM, NN	\$250 per Covered Individual per calendar year, limited to a maximum \$500 per calendar year in covered charges incurred by a group consisting of one Covered Participant and all related Covered Dependents.
(e)	5D, 5N, 5S, A5, B5, CM, M5, N5, RB	\$500 per Covered Individual per calendar year, limited to a maximum \$1,000 per calendar year in covered charges incurred by a group consisting of one Covered Participant and all related Covered Dependents.
(f)	AK, BK, CK, DK, WK	Benefit provided by Kaiser Permanente HMO.

20.08 TEAMCARE PRESCRIPTION DRUG OUT-OF-POCKET EXPENSE LIMITS

Section 12.07 provides for a Prescription Drug Benefit and Section 20.01(f) provides for the corresponding Schedule of Benefits, including a "maximum co-payment ... [of] \$200 for each filled prescription purchased through the TeamCare RX program." The Prescription Drug Benefit applies to all drugs that are Prescription Drugs as defined in Section 1.53. The Fund provides separate TeamCare Prescription Drug Out-of-Pocket Expense Limits for injectable drugs and total prescription drugs (including injectable drugs). These separate limits apply only to Covered Individuals who in a calendar year purchase drugs through the TeamCare RX program. Each such Covered Individual's share of the cost of all covered Prescription Drugs purchased through the TeamCare RX program (including injectable drugs) is applied towards these separate annual limits. After the annual limit for injectable drugs has been reached, the Fund is obligated to pay the full cost of all injectable drugs purchased by the Covered Individual through the TeamCare RX program during the remainder of that calendar year. After the annual limit for total prescription drugs has been reached, the Fund is obligated to pay the full cost of all prescription drugs purchased by the Covered Individual through the TeamCare RX program during the remainder of that calendar year.

	PLAN	SCHEDULE OF BENEFITS
(a)	4M, 6N, 8N, N3, N6, N8, NM, NN	TeamCare RX Injectable Drug Out-of-Pocket Expense Limit of \$1,000 per Covered Individual per calendar year. Total TeamCare Prescription Drug Out-of-Pocket Expense Limit of \$7,550 per Covered Individual per calendar year, limited to a maximum of \$15,100 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(b)	1N, N1	TeamCare RX Injectable Drug Out-of-Pocket Expense Limit of \$1,000 per Covered Individual per calendar year. Total TeamCare Prescription Drug Out-of-Pocket Expense Limit of \$7,050 per Covered Individual per calendar year, limited to a maximum of \$14,100 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(c)	2N, AB, AN, BA, BN	TeamCare RX Injectable Drug Out-of-Pocket Expense Limit of \$1,000 per Covered Individual per calendar year. Total TeamCare Prescription Drug Out-of-Pocket Expense Limit of \$6,050 per Covered Individual per calendar year, limited to a maximum of \$12,100 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(d)	5N, 5S, A5, B5, CM, M5, N5	TeamCare RX Injectable Drug Out-of-Pocket Expense Limit of \$1,000 per Covered Individual per calendar year. Total TeamCare Prescription Drug Out-of-Pocket Expense Limit of \$4,550 per Covered Individual per calendar year, limited to a maximum of \$9,100 per calendar year for each group consisting of one Covered Participant and all related Covered Dependents.
(e)	AK, BK, CK, DK, WK	TeamCare RX Injectable Drug Out-of-Pocket Expense Limit not applicable to these plans. Benefit subject to Kaiser Permanente HMO rules.